SECURITIES AND EXCHANGE COMMISSION

FORM 8-K

Current report filing

Filing Date: **2022-10-07** | Period of Report: **2022-09-30** SEC Accession No. 0001213900-22-062578

(HTML Version on secdatabase.com)

FILER

NRI Real Estate Investment & Technology, Inc.

CIK:1877561| IRS No.: 871031361 | State of Incorp.:MD | Fiscal Year End: 1231

Type: 8-K | Act: 34 | File No.: 000-56395 | Film No.: 221298940

SIC: 6798 Real estate investment trusts

Mailing Address 1340 S. DIXIE HWY, SUITE 812 CORAL GABLES FL 33134

Business Address 1340 S. DIXIE HWY, SUITE 812 CORAL GABLES FL 33134 305-529-9928

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): September 30, 2022

NRI Real Estate Investment and Technology, Inc. (Exact Name of Registrant as Specified in Its Charter)

Maryland	000-56395	87-1031361		
(State or other jurisdiction of incorporation)	(Commission File Number)	(I.R.S. Employer Identification)		
1340 5	South Dixie Highway, Suite 612, Coral Gables, Flo	rida 33146		
	(Address of principal executive offices) (Zip Code)			
	(305) 259-9928 (Registrant's Telephone Number, Including Area C	ode)		
(Fc	ormer name or former address, if changed since last	report)		
Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):				
□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)				
□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a -12)				
□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d -2(b))				
□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e -4(c))				
Securities registered pursuant to Section	12(b) of the Act:			
Title of each class	Trading Symbol(s)	Name of each exchange on which registered		
None	N/A	N/A		
	gistrant is an emerging growth company as defined of the Securities Exchange Act of 1934 (§240.12b-			
Emerging growth company				
	ate by check mark if the registrant has elected notial accounting standards provided pursuant to Sec			

Item 1.01 Entry into a Material Definitive Agreement.

Amended and Restated Limited Partnership Agreement

NRI Real Estate Investment and Technology, Inc., a Maryland corporation ("we," "us,", "our" or the "Company"), entered into an Amended and Restated Limited Partnership Agreement (the "Amended and Restated Partnership Agreement") of NRI Real Token LP (the "Operating Partnership"), as of September 30, 2022, with NRI Real Token Thesis LLC, a Delaware limited liability company ("NRI Thesis") and the limited partners of NRI Real Token LP, for the purpose of amending the Limited Partnership Agreement of NRI Real Token LP, dated November 19, 2021, as amended by the First Amendment to Limited Partnership Agreement on April 11, 2022 which designated a series of preferred units in the Operating Partnership (collectively, the "Original Partnership Agreement") to (i) assign all the rights and obligations of NRI Thesis, as a general partner of the Operating Partnership, to the Company and convert the general partnership interest held by NRI Thesis into a limited partnership interest, and (ii) redesignate the "Special GP Interest" (as such term is defined in the Original Partnership Agreement) as the "Special OP Units" as set forth in the Amended and Restated Partnership Agreement. The effect of this amendment and restatement of the Original Partnership Agreement is to make the Company the sole general partner of the Operating Partnership with the full, complete and exclusive discretion of manage and control the business of the Operating Partnership.

The Amended and Restated Partnership Agreement otherwise retains all material terms of the Original Partnership Agreement.

Amended and Restated Advisory Agreement

The Company entered into an Amended and Restated Advisory Agreement (the "Amended and Restated Advisory Agreement"), as of September 30, 2022, with NRI Real Token Advisors LLC, a Delaware limited liability company ("NRI Advisors"), and the Operating Partnership for the purpose of amending the Advisory Agreement, dated November 19, 2021 (the "Original Advisory Agreement") in order to remove NRI Thesis as a party to the Original Advisory Agreement due to the conversion of its general partnership interest in the Operating Partnership to a limited partnership interest and the vesting of the full, complete and exclusive discretion of manage and control the business of the Operating Partnership in the Company. The signature of NRI Thesis on the Amended and Restated Advisory Agreement acknowledges its agreement to these changes to the Original Advisory Agreement.

The Amended and Restated Advisory Agreement otherwise retains all material terms of the Original Advisory Agreement.

The foregoing descriptions of the Amended and Restated Partnership Agreement and the Amended and Restated Advisory Agreement is only a summary and is qualified in its entirety by reference to the full text of the Amended and Restated Partnership Agreement and Amended and Restated Advisory Agreement, which is filed as Exhibit 10.1 and 10.2, respectively, to this Current Report on Form 8-K and is incorporated by reference in this Item 1.01.

Item 5.03 Amendment to Articles of Incorporation or Bylaws; Change in Fiscal Year.

On September 30, 2022, the Company amended and restated the Amended and Restated Bylaws of the Company (the "Second A&R Bylaws"), effective as of September 30, 2022. The Company amended Section 7 of the Second A&R Bylaws to correctly reference the Amended and Restated Advisory Agreement.

The A&R Bylaws otherwise retains all material terms of the Company's prior Amended and Restated Bylaws in effect.

The foregoing description of the Second A&R Bylaws is only a summary and is qualified in its entirety by reference to the full text of the Second A&R Bylaws, which is filed as Exhibit 3.1 to this Current Report on Form 8-K and is incorporated by reference in this Item 5.03.

1

(d) Exhibits.

Exhibit	Description
Number	Description
3.1	Second Amended & Restated Bylaws of the Company, effective September 30, 2022
10.1*	Amended and Restated Limited Partnership Agreement of NRI Real Token LP, as of September 30, 2022, by and among
	the Company, NRI Real Token Thesis LLC and the limited partners
10.2	Amended and Restated Advisory Agreement, as of September 30, 2022, by and among the Company, NRI Real Token
	Advisors LLC and NRI Real Token LP
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

Pursuant to Item 601(a)(5) of Regulation S-K, schedules and similar attachments to this exhibit have been omitted because they do not contain information material to an investment or voting decision and such information is not otherwise disclosed in such exhibit. The Company will supplementally provide a copy of any omitted schedule or similar attachment to the U.S. Securities and Exchange Commission or its staff upon request.

2

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

NRI REAL ESTATE INVESTMENT AND TECHNOLOGY, INC.

Date: October 6, 2022 By: /s/ Brent Reynolds

Name: Brent Reynolds

Title: Chief Executive Officer, President and

Director

3

NRI REAL ESTATE INVESTMENT AND TECHNOLOGY, INC.

SECOND AMENDED AND RESTATED BYLAWS

ARTICLE I

OFFICES

Section 1. <u>PRINCIPAL OFFICE</u>. The principal office of the Corporation in the State of Maryland shall be located at such place as the Board of Directors may designate.

Section 2. <u>ADDITIONAL OFFICES</u>. The Corporation may have additional offices, including a principal executive office, at such places as the Board of Directors may from time to time determine or the business of the Corporation may require.

ARTICLE II

MEETINGS OF STOCKHOLDERS

Section 1. <u>PLACE</u>. All meetings of stockholders shall be held at the principal executive office of the Corporation or at such other place as shall be set in accordance with these Bylaws and stated in the notice of the meeting.

Section 2. <u>ANNUAL MEETING</u>. An annual meeting of stockholders for the election of directors and the transaction of any business within the powers of the Corporation shall be held on the date and at the time and place set by the Board of Directors.

Section 3. SPECIAL MEETINGS.

(a) General. Each of the chair of the board, chief executive officer, president and Board of Directors may call a special meeting of stockholders. Except as provided in subsection (b)(4) of this Section 3, a special meeting of stockholders shall be held on the date and at the time and place set by the chair of the board, chief executive officer, president or Board of Directors, whoever has called the meeting. Subject to subsection (b) of this Section 3, a special meeting of stockholders shall also be called by the secretary of the Corporation to act on any matter that may properly be considered at a meeting of stockholders upon the written request of stockholders entitled to cast not less than a majority of all the votes entitled to be cast on such matter at such meeting (the "Special Meeting Percentage").

(b) Stockholder-Requested Special Meetings. (1) Any stockholder of record seeking to have stockholders request a special meeting shall, by sending written notice to the secretary (the "Record Date Request Notice") by registered mail, return receipt requested, request the Board of Directors to fix a record date to determine the stockholders entitled to request a special meeting (the "Request Record Date"). The Record Date Request Notice shall set forth the purpose of the meeting and the matters proposed to be acted on at it, shall be signed by one or more stockholders of record as of the date of signature (or their agents duly authorized in a writing accompanying the Record Date Request Notice), shall bear the date of signature of each such stockholder (or such agent) and shall set forth all information relating to each such stockholder and each matter proposed to be acted on at the meeting that would be required to be disclosed in connection with the solicitation of proxies for the election of directors in an election contest (even if an election contest is not involved), or would otherwise be required in connection with such a solicitation, in each case pursuant to Regulation 14A (or any successor provision) under the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder (the "Exchange Act"). Upon receiving the Record Date Request Notice, the Board of Directors may fix a Request Record Date. The Request Record Date shall not precede and shall not be more than ten days after the close of business on the date on which the resolution fixing the Request Record Date is adopted by the Board of Directors. If the Board of Directors, within ten days after the date on which a valid Record Date Request Notice is received, fails to adopt a resolution fixing the Request Record Date, the Request Record Date shall be the close of business on the tenth day after the first date on which a Record Date Request Notice is received by the secretary.

(2) In order for any stockholder to request a special meeting to act on any matter that may properly be considered at a meeting of stockholders, one or more written requests for a special meeting (collectively, the "Special Meeting Request") signed by stockholders of record (or their agents duly authorized in a writing accompanying the request) as of the Request Record Date entitled to cast not less than the Special Meeting Percentage shall be delivered to the secretary. In addition, the Special Meeting Request shall (a) set forth the purpose of the meeting and the matters proposed to be acted on at it (which shall be limited to those lawful matters set forth in the Record Date Request Notice received by the secretary), (b) bear the date of signature of each such stockholder (or such agent) signing the Special Meeting Request, (c) set forth (i) the name and address, as they appear in the Corporation's books, of each stockholder signing such request (or on whose behalf the Special Meeting Request is signed), (ii) the class, series and number of all shares of stock of the Corporation which are owned (beneficially or of record) by each such stockholder and (iii) the nominee holder for, and number of, shares of stock of the Corporation owned beneficially but not of record by such stockholder, (d) be sent to the secretary by registered mail, return receipt requested, and (e) be received by the secretary within 60 days after the Request Record Date. Any requesting stockholder (or agent duly authorized in a writing accompanying the revocation of the Special Meeting Request) may revoke his, her or its request for a special meeting at any time by written revocation delivered to the secretary.

(3) The secretary shall inform the requesting stockholders of the reasonably estimated cost of preparing and mailing or delivering the notice of the meeting (including the Corporation's proxy materials). The secretary shall not be required to call a special meeting upon stockholder request and such meeting shall not be held unless, in addition to the documents required by paragraph (2) of this Section 3(b), the secretary receives payment of such reasonably estimated cost prior to the preparation and mailing or delivery of such notice of the meeting.

2

(4) In the case of any special meeting called by the secretary upon the request of stockholders (a "Stockholder-Requested Meeting"), such meeting shall be held at such place, date and time as may be designated by the Board of Directors; provided, however, that the date of any Stockholder-Requested Meeting shall be not more than 90 days after the record date for such meeting (the "Meeting Record Date"); and provided further that if the Board of Directors fails to designate, within ten days after the date that a valid Special Meeting Request is actually received by the secretary (the "Delivery Date"), a date and time for a Stockholder-Requested Meeting, then such meeting shall be held at 2:00 p.m., local time, on the 90th day after the Meeting Record Date or, if such 90th day is not a Business Day (as defined below), on the first preceding Business Day; and provided further that in the event that the Board of Directors fails to designate a place for a Stockholder-Requested Meeting within ten days after the Delivery Date, then such meeting shall be held at the principal executive office of the Corporation. In fixing a date for a Stockholder-Requested Meeting, the Board of Directors may consider such factors as it deems relevant, including, without limitation, the nature of the matters to be considered, the facts and circumstances surrounding any request for the meeting and any plan of the Board of Directors to call an annual meeting or a special meeting. In the case of any Stockholder-Requested Meeting, if the Board of Directors fails to fix a Meeting Record Date that is a date within 30 days after the Delivery Date, then the close of business on the 30th day after the Delivery Date shall be the Meeting Record Date. The Board of Directors may revoke the notice for any Stockholder-Requested Meeting in the event that the requesting stockholders fail to comply with the provisions of paragraph (3) of this Section 3(b).

(5) If written revocations of the Special Meeting Request have been delivered to the secretary and the result is that stockholders of record (or their agents duly authorized in writing), as of the Request Record Date, entitled to cast less than the Special Meeting Percentage have delivered, and not revoked, requests for a special meeting on the matter to the secretary: (i) if the notice of meeting has not already been delivered, the secretary shall refrain from delivering the notice of the meeting and send to all requesting stockholders who have not revoked such requests written notice of any revocation of a request for a special meeting on the matter, or (ii) if the notice of meeting has been delivered and if the secretary first sends to all requesting stockholders who have not revoked requests for a special meeting on the matter written notice of any revocation of a request for the special meeting and written notice of the Corporation's intention to revoke the notice of the meeting or for the chair of the meeting to adjourn the meeting without action on the matter, (A) the secretary may revoke the notice of the meeting at any time before ten days before the commencement of the meeting or (B) the chair of the meeting may call the meeting to order and adjourn the meeting from time to time without acting on the matter. Any request for a special meeting received after a revocation by the secretary of a notice of a meeting shall be considered a request for a new special meeting.

(6) The chair of the board, chief executive officer, president or Board of Directors may appoint regionally or nationally recognized independent inspectors of elections to act as the agent of the Corporation for the purpose of promptly performing a ministerial review of the validity of any purported Special Meeting Request received by the secretary. For the purpose of permitting the inspectors to perform such review, no such purported Special Meeting Request shall be deemed to have been received by the

secretary until the earlier of (i) five Business Days after actual receipt by the secretary of such purported request and (ii) such date as the independent inspectors certify to the Corporation that the valid requests received by the secretary represent, as of the Request Record Date, stockholders of record entitled to cast not less than the Special Meeting Percentage. Nothing contained in this paragraph (6) shall in any way be construed to suggest or imply that the Corporation or any stockholder shall not be entitled to contest the validity of any request, whether during or after such five Business Day period, or to take any other action (including, without limitation, the commencement, prosecution or defense of any litigation with respect thereto, and the seeking of injunctive relief in such litigation).

(7) For purposes of these Bylaws, "Business Day" shall mean any day other than a Saturday, a Sunday or a day on which banking institutions in the State of Florida are authorized or obligated by law or executive order to close.

3

Section 4. NOTICE. Not less than ten nor more than 90 days before each meeting of stockholders, the secretary shall give to each stockholder entitled to vote at such meeting and to each stockholder not entitled to vote who is entitled to notice of the meeting notice in writing or by electronic transmission stating the time and place of the meeting and, in the case of a special meeting or as otherwise may be required by any statute, the purpose for which the meeting is called, by mail, by presenting it to such stockholder personally, by leaving it at the stockholder's residence or usual place of business, by electronic transmission or by any other means permitted by Maryland law. If mailed, such notice shall be deemed to be given when deposited in the United States mail addressed to the stockholder at the stockholder's address as it appears on the records of the Corporation, with postage thereon prepaid. If transmitted electronically, such notice shall be deemed to be given when transmitted to the stockholder by an electronic transmission to any address or number of the stockholder at which the stockholder receives electronic transmissions. The Corporation may give a single notice to all stockholders who share an address, which single notice shall be effective as to any stockholder at such address, unless such stockholder objects to receiving such single notice or revokes a prior consent to receiving such single notice. Failure to give notice of any meeting to one or more stockholders, or any irregularity in such notice, shall not affect the validity of any meeting fixed in accordance with this Article II or the validity of any proceedings at any such meeting.

Subject to Section 11(a) of this Article II, any business of the Corporation may be transacted at an annual meeting of stockholders without being specifically designated in the notice, except such business as is required by any statute to be stated in such notice. No business shall be transacted at a special meeting of stockholders except as specifically designated in the notice. The Corporation may postpone or cancel a meeting of stockholders by making a public announcement (as defined in Section 11(c)(3) of this Article II) of such postponement or cancellation prior to the meeting. Notice of the date, time and place to which the meeting is postponed shall be given not less than ten days prior to such date and otherwise in the manner set forth in this section.

Section 5. ORGANIZATION AND CONDUCT. Every meeting of stockholders shall be conducted by an individual appointed by the Board of Directors to be chair of the meeting or, in the absence of such appointment or appointed individual, by the chair of the board or, in the case of a vacancy in the office or absence of the chair of the board, by one of the following individuals present at the meeting in the following order: the chief executive officer, the president, the vice presidents in their order of rank and, within each rank, in their order of seniority, the secretary, or, in the absence of such officers, a chair chosen by the stockholders by the vote of a majority of the votes cast by stockholders present in person or by proxy. The secretary or, in the case of a vacancy in the office or absence of the secretary, an assistant secretary or an individual appointed by the Board of Directors or the chair of the meeting shall act as secretary. In the event that the secretary presides at a meeting of stockholders, an assistant secretary, or, in the absence of all assistant secretaries, an individual appointed by the Board of Directors or the chair of the meeting, shall record the minutes of the meeting. Even if present at the meeting, the person holding the office named herein may delegate to another person the power to act as chair or secretary of the meeting. The order of business and all other matters of procedure at any meeting of stockholders shall be determined by the chair of the meeting. The chair of the meeting may prescribe such rules, regulations and procedures and take such action as, in the discretion of the chair and without any action by the stockholders, are appropriate for the proper conduct of the meeting, including, without limitation, (a) restricting admission to the time set for the commencement of the meeting; (b) limiting attendance or participation at the meeting to stockholders of record of the Corporation, their duly authorized proxies and such other individuals as the chair of the meeting may determine; (c) recognizing speakers at the meeting and determining when and for how long speakers and any individual speaker may address the meeting; (d) determining when and for how long the polls should be opened and when the polls should be closed and when announcement of the results should be made; (e) maintaining order and security at the meeting; (f) removing any stockholder or any other individual who refuses to comply with meeting procedures, rules or guidelines as set forth by the chair of the meeting; (g) concluding a meeting or recessing or adjourning the meeting, whether or not a quorum is present, to a later date and time and at a place either (i) announced at the meeting or (ii) provided at a future time through means announced at the meeting; and (h) complying with any state

and local laws and regulations concerning safety and security. Unless otherwise determined by the chair of the meeting, meetings of stockholders shall not be required to be held in accordance with any rules of parliamentary procedure.

Section 6. QUORUM. At any meeting of stockholders, the presence in person or by proxy of stockholders entitled to cast a majority of all the votes entitled to be cast at such meeting on any matter shall constitute a quorum; but this section shall not affect any requirement under any statute or the charter of the Corporation (the "Charter") for the vote necessary for the approval of any matter. If such quorum is not established at any meeting of the stockholders, the chair of the meeting may adjourn the meeting from time to time to a date not more than 120 days after the original record date without notice other than announcement at the meeting. The date, time and place of the meeting, as reconvened, shall be either (i) announced at the meeting or (ii) provided at a future time through means announced at the meeting.

The stockholders present either in person or by proxy, at a meeting which has been duly called and at which a quorum has been established, may continue to transact business until adjournment, notwithstanding the withdrawal from the meeting of enough stockholders to leave fewer than would be required to establish a quorum.

4

Section 7. <u>VOTING</u>. A plurality of all the votes cast at a meeting of stockholders duly called and at which a quorum is present shall be sufficient to elect a director; provided, however, so long as the Amended and Restated Advisory Agreement (as amended from time to time, the "Advisory Agreement"), by and among the Corporation, NRI Real Token LP, a Delaware limited partnership, and NRI Real Token Advisors LLC, a Delaware limited liability company, remains in effect, it shall be a qualification of at least two directors of the Corporation that he or she shall have been nominated by NRI Real Token Thesis LLC, a Delaware limited liability company ("NRI"). Each share entitles the holder thereof to vote for as many individuals as there are directors to be elected and for whose election the holder is entitled to vote. A majority of the votes cast at a meeting of stockholders duly called and at which a quorum is present shall be sufficient to approve any other matter which may properly come before the meeting, unless more than a majority of the votes cast is required by statute or by the Charter. Unless otherwise provided by statute or by the Charter, each outstanding share of stock, regardless of class, entitles the holder thereof to cast one vote on each matter submitted to a vote at a meeting of stockholders. Voting on any question or in any election may be *viva voce* unless the chair of the meeting shall order that voting be by ballot or otherwise.

Section 8. <u>PROXIES</u>. A holder of record of shares of stock of the Corporation may cast votes in person or by proxy executed by the stockholder or by the stockholder's duly authorized agent in any manner permitted by applicable law. Such proxy or evidence of authorization of such proxy shall be filed with the record of the proceedings of the meeting. No proxy shall be valid more than eleven months after its date unless otherwise provided in the proxy.

Section 9. <u>VOTING OF STOCK BY CERTAIN HOLDERS</u>. Stock of the Corporation registered in the name of a corporation, limited liability company, partnership, joint venture, trust or other entity, if entitled to be voted, may be voted by the president or a vice president, managing member, manager, general partner or trustee thereof, as the case may be, or a proxy appointed by any of the foregoing individuals, unless some other person who has been appointed to vote such stock pursuant to a bylaw or a resolution of the governing body of such corporation or other entity or agreement of the partners of a partnership presents a certified copy of such bylaw, resolution or agreement, in which case such person may vote such stock. Any trustee or fiduciary, in such capacity, may vote stock registered in such trustee's or fiduciary's name, either in person or by proxy.

Shares of stock of the Corporation directly or indirectly owned by it shall not be voted at any meeting and shall not be counted in determining the total number of outstanding shares entitled to be voted at any given time, unless they are held by it in a fiduciary capacity, in which case they may be voted and shall be counted in determining the total number of outstanding shares at any given time.

The Board of Directors may adopt by resolution a procedure by which a stockholder may certify in writing to the Corporation that any shares of stock registered in the name of the stockholder are held for the account of a specified person other than the stockholder. The resolution shall set forth the class of stockholders who may make the certification, the purpose for which the certification may be made, the form of certification and the information to be contained in it; if the certification is with respect to a record date, the time after the record date within which the certification must be received by the Corporation; and any other provisions with respect to the procedure which the Board of Directors considers necessary or appropriate. On receipt by the secretary of the Corporation of such certification, the person specified in the certification shall be regarded as, for the purposes set forth in the certification, the holder of record of the specified stock in place of the stockholder who makes the certification.

Section 10. <u>INSPECTORS</u>. The Board of Directors or the chair of the meeting may appoint, before or at the meeting, one or more inspectors for the meeting and any successor to the inspector. Except as otherwise provided by the chair of the meeting, the inspectors, if any, shall (a) determine the number of shares of stock represented at the meeting, in person or by proxy, and the validity and effect of proxies, (b) receive and tabulate all votes, ballots or consents, (c) report such tabulation to the chair of the meeting, (d) hear and determine all challenges and questions arising in connection with the right to vote, and (e) do such acts as are proper to fairly conduct the election or vote. Each such report shall be in writing and signed by the inspector or by a majority of them if there is more than one inspector acting at such meeting. If there is more than one inspector, the report of a majority shall be the report of the inspectors. The report of the inspectors on the number of shares represented at the meeting and the results of the voting shall be *prima facie* evidence thereof.

Section 11. <u>ADVANCE NOTICE OF STOCKHOLDER NOMINEES FOR DIRECTOR AND OTHER</u> STOCKHOLDER PROPOSALS.

(a) Annual Meetings of Stockholders. (1) Nominations of individuals for election to the Board of Directors and the proposal of other business to be considered by the stockholders may be made at an annual meeting of stockholders (i) pursuant to the Corporation's notice of meeting, (ii) by or at the direction of the Board of Directors or (iii) by any stockholder of the Corporation who was a stockholder of record at the record date set by the Board of Directors for the purpose of determining stockholders entitled to vote at the annual meeting, at the time of giving of notice by the stockholder as provided for in this Section 11(a) and at the time of the annual meeting (and any postponement or adjournment thereof), who is entitled to vote at the meeting in the election of each individual so nominated or on any such other business and who has complied with this Section 11(a).

(2) For any nomination or other business to be properly brought before an annual meeting by a stockholder pursuant to clause (iii) of paragraph (a)(1) of this Section 11, the stockholder must have given timely notice thereof in writing to the secretary of the Corporation and any such other business must otherwise be a proper matter for action by the stockholders. To be timely, a stockholder's notice shall set forth all information required under this Section 11 and shall be delivered to the secretary at the principal executive office of the Corporation not earlier than the 150th day nor later than 5:00 p.m., Eastern Time, on the 120th day prior to the first anniversary of the date of the proxy statement (as defined in Section 11(c)(3) of this Article II) for the preceding year's annual meeting; provided, however, that in connection with the Corporation's first annual meeting or in the event that the date of the annual meeting is advanced or delayed by more than 30 days from the first anniversary of the date of the preceding year's annual meeting, in order for notice by the stockholder to be timely, such notice must be so delivered not earlier than the 150th day prior to the date of such annual meeting and not later than 5:00 p.m., Eastern Time, on the later of the 120th day prior to the date of such annual meeting, as originally convened, or the tenth day following the day on which public announcement of the date of such meeting is first made. The public announcement of a postponement or adjournment of an annual meeting shall not commence a new time period for the giving of a stockholder's notice as described above.

6

(3) Such stockholder's notice shall set forth:

(i) as to each individual whom the stockholder proposes to nominate for election or reelection as a director (each, a "Proposed Nominee"), all information relating to the Proposed Nominee that would be required to be disclosed in connection with the solicitation of proxies for the election of the Proposed Nominee as a director in an election contest (even if an election contest is not involved), or would otherwise be required in connection with such solicitation, in each case pursuant to Regulation 14A (or any successor provision) under the Exchange Act;

(ii) as to any other business that the stockholder proposes to bring before the meeting, a description of such business, the stockholder's reasons for proposing such business at the meeting and any material interest in such business of such

stockholder or any Stockholder Associated Person (as defined below), individually or in the aggregate, including any anticipated benefit to the stockholder or the Stockholder Associated Person therefrom;

(iii) as to the stockholder giving the notice, any Proposed Nominee and any Stockholder Associated Person,

(A) the class, series and number of all shares of stock or other securities of the Corporation or any affiliate thereof (collectively, the "Company Securities"), if any, which are owned (beneficially or of record) by such stockholder, Proposed Nominee or Stockholder Associated Person, the date on which each such Company Security was acquired and the investment intent of such acquisition, and any short interest (including any opportunity to profit or share in any benefit from any decrease in the price of such stock or other security) in any Company Securities of any such person,

(B) the nominee holder for, and number of, any Company Securities owned beneficially but not of record by such stockholder, Proposed Nominee or Stockholder Associated Person,

(C) whether and the extent to which such stockholder, Proposed Nominee or Stockholder Associated Person, directly or indirectly (through brokers, nominees or otherwise), is subject to or during the last six months has engaged in any hedging, derivative or other transaction or series of transactions or entered into any other agreement, arrangement or understanding (including any short interest, any borrowing or lending of securities or any proxy or voting agreement), the effect or intent of which is to (I) manage risk or benefit of changes in the price of Company Securities for such stockholder, Proposed Nominee or Stockholder Associated Person or (II) increase or decrease the voting power of such stockholder, Proposed Nominee or Stockholder Associated Person in the Corporation or any affiliate thereof disproportionately to such person's economic interest in the Company Securities, and

(D) any substantial interest, direct or indirect (including, without limitation, any existing or prospective commercial, business or contractual relationship with the Corporation), by security holdings or otherwise, of such stockholder, Proposed Nominee or Stockholder Associated Person, in the Corporation or any affiliate thereof, other than an interest arising from the ownership of Company Securities where such stockholder, Proposed Nominee or Stockholder Associated Person receives no extra or special benefit not shared on a *pro rata* basis by all other holders of the same class or series;

7

(iv) as to the stockholder giving the notice, any Stockholder Associated Person with an interest or ownership referred to in clauses (ii) or (iii) of this paragraph (3) of this Section 11(a) and any Proposed Nominee,

(A) the name and address of such stockholder, as they appear on the Corporation's stock ledger, and the current name and address, if different, of each such Stockholder Associated Person and any Proposed Nominee, and

(B) the investment strategy or objective, if any, of such stockholder and each such Stockholder Associated Person who is not an individual and a copy of the prospectus, offering memorandum or similar document, if any, provided to investors or potential investors in such stockholder and each such Stockholder Associated Person;

(v) the name and address of any person who contacted or was contacted by the stockholder giving the notice or any Stockholder Associated Person about the Proposed Nominee or other business proposal; and

(vi) to the extent known by the stockholder giving the notice, the name and address of any other stockholder supporting the nominee for election or reelection as a director or the proposal of other business.

(4) Such stockholder's notice shall, with respect to any Proposed Nominee, be accompanied by a written undertaking executed by the Proposed Nominee (i) that such Proposed Nominee (a) is not, and will not become, a party to any agreement, arrangement or understanding with any person or entity other than the Corporation in connection with service or action as a director that has not been disclosed to the Corporation and (b) will serve as a director of the Corporation if elected; and (ii) attaching a completed Proposed Nominee questionnaire (which questionnaire shall be provided by the Corporation, upon request by the stockholder providing the notice, and shall include all information relating to the Proposed Nominee that would be required to be disclosed in connection with the solicitation of proxies for the election of the Proposed Nominee as a director in an election contest (even if an election contest is not involved), or would otherwise be required in connection with such solicitation, in each case pursuant to Regulation 14A (or any successor

provision) under the Exchange Act, or would be required pursuant to the rules of any national securities exchange on which any securities of the Corporation are listed or over-the-counter market on which any securities of the Corporation are traded).

(5) Notwithstanding anything in this subsection (a) of this Section 11 to the contrary, in the event that the number of directors to be elected to the Board of Directors is increased, and there is no public announcement of such action at least 130 days prior to the first anniversary of the date of the proxy statement (as defined in Section 11(c)(3) of this Article II) for the preceding year's annual meeting, a stockholder's notice required by this Section 11(a) shall also be considered timely, but only with respect to nominees for any new positions created by such increase, if it shall be delivered to the secretary at the principal executive office of the Corporation not later than 5:00 p.m., Eastern Time, on the tenth day following the day on which such public announcement is first made by the Corporation.

8

(6) For purposes of this Section 11, "Stockholder Associated Person" of any stockholder shall mean (i) any person acting in concert with such stockholder, (ii) any beneficial owner of shares of stock of the Corporation owned of record or beneficially by such stockholder (other than a stockholder that is a depositary) and (iii) any person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, such stockholder or such Stockholder Associated Person.

(b) Special Meetings of Stockholders. Only such business shall be conducted at a special meeting of stockholders as shall have been brought before the meeting pursuant to the Corporation's notice of meeting and, except as contemplated by and in accordance with the next two sentences of this Section 11(b), no stockholder may nominate an individual for election to the Board of Directors or make a proposal of other business to be considered at a special meeting. Nominations of individuals for election to the Board of Directors may be made at a special meeting of stockholders at which directors are to be elected only (i) by or at the direction of the Board of Directors or (ii) provided that the special meeting has been called in accordance with Section 3(a) of this Article II for the purpose of electing directors, by any stockholder of the Corporation who is a stockholder of record at the record date set by the Board of Directors for the purpose of determining stockholders entitled to vote at the special meeting, at the time of giving of notice provided for in this Section 11 and at the time of the special meeting (and any postponement or adjournment thereof), who is entitled to vote at the meeting in the election of each individual so nominated and who has complied with the notice procedures set forth in this Section 11. In the event the Corporation calls a special meeting of stockholders for the purpose of electing one or more individuals to the Board of Directors, any stockholder may nominate an individual or individuals (as the case may be) for election as a director as specified in the Corporation's notice of meeting, if the stockholder's notice, containing the information required by paragraphs (a)(3) and (4) of this Section 11, is delivered to the secretary at the principal executive office of the Corporation not earlier than the 120th day prior to such special meeting and not later than 5:00 p.m., Eastern Time, on the later of the 90th day prior to such special meeting or the tenth day following the day on which public announcement is first made of the date of the special meeting and of the nominees proposed by the Board of Directors to be elected at such meeting. The public announcement of a postponement or adjournment of a special meeting shall not commence a new time period for the giving of a stockholder's notice as described above.

(c) <u>General</u>. (1) If information submitted pursuant to this Section 11 by any stockholder proposing a nominee for election as a director or any proposal for other business at a meeting of stockholders shall be inaccurate in any material respect, such information may be deemed not to have been provided in accordance with this Section 11. Any such stockholder shall notify the Corporation of any inaccuracy or change (within two Business Days of becoming aware of such inaccuracy or change) in any such information. Upon written request by the secretary or the Board of Directors, any such stockholder shall provide, within five Business Days of delivery of such request (or such other period as may be specified in such request), (A) written verification, satisfactory, in the discretion of the Board of Directors or any authorized officer of the Corporation, to demonstrate the accuracy of any information submitted by the stockholder pursuant to this Section 11, and (B) a written update of any information (including, if requested by the Corporation, written confirmation by such stockholder that it continues to intend to bring such nomination or other business proposal before the meeting) submitted by the stockholder pursuant to this Section 11 as of an earlier date. If a stockholder fails to provide such written verification or written update within such period, the information as to which written verification or a written update was requested may be deemed not to have been provided in accordance with this Section 11.

(2) Only such individuals who are nominated in accordance with this Section 11 shall be eligible for election by stockholders as directors, and only such business shall be conducted at a meeting of stockholders as shall have been brought before the meeting in accordance with this Section 11. The chair of the meeting shall have the power to determine whether a nomination or any other business proposed to be brought before the meeting was made or proposed, as the case may be, in accordance with this Section 11.

(3) For purposes of this Section 11, "the date of the proxy statement" shall have the same meaning as "the date of the company's proxy statement released to shareholders" as used in Rule 14a-8(e) promulgated under the Exchange Act, as interpreted by the Securities and Exchange Commission from time to time. "Public announcement" shall mean disclosure (A) in a press release reported by the Dow Jones News Service, Associated Press, Business Wire, PR Newswire or other widely circulated news or wire service or (B) in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to the Exchange Act.

(4) Notwithstanding the foregoing provisions of this Section 11, a stockholder shall also comply with all applicable requirements of state law and of the Exchange Act with respect to the matters set forth in this Section 11. Nothing in this Section 11 shall be deemed to affect any right of a stockholder to request inclusion of a proposal in, or the right of the Corporation to omit a proposal from, any proxy statement filed by the Corporation with the Securities and Exchange Commission pursuant to Rule 14a-8 (or any successor provision) under the Exchange Act. Nothing in this Section 11 shall require disclosure of revocable proxies received by the stockholder or Stockholder Associated Person pursuant to a solicitation of proxies after the filing of an effective Schedule 14A by such stockholder or Stockholder Associated Person under Section 14(a) of the Exchange Act.

(5) Notwithstanding anything in these Bylaws to the contrary, except as otherwise determined by the chair of the meeting, if the stockholder giving notice as provided for in this Section 11 does not appear in person or by proxy at such annual or special meeting to present each nominee for election as a director or the proposed business, as applicable, such matter shall not be considered at the meeting.

Section 12. <u>TELEPHONE AND REMOTE COMMUNICATION MEETINGS</u>. The Board of Directors or chair of the meeting may permit one or more stockholders to participate in a meeting by means of a conference telephone or other communications equipment in any manner permitted by Maryland law. In addition, the Board of Directors may determine that a meeting not be held at any place, but instead may be held solely by means of remote communications in any matter permitted by Maryland law. Participation in a meeting by these means constitutes presence in person at the meeting.

10

Section 13. <u>CONTROL SHARE ACQUISITION ACT</u>. Notwithstanding any other provision of the Charter or these Bylaws, Title 3, Subtitle 7 of the Maryland General Corporation Law, or any successor statute (the "MGCL"), shall not apply to any acquisition by any person of shares of stock of the Corporation. This section may be repealed, in whole or in part, at any time, whether before or after an acquisition of control shares and, upon such repeal, may, to the extent provided by any successor bylaw, apply to any prior or subsequent control share acquisition.

Section 14. <u>STOCKHOLDERS' CONSENT IN LIEU OF MEETING</u>. Any action required or permitted to be taken at any meeting of stockholders may be taken without a meeting (a) if a unanimous consent setting forth the action is given in writing or by electronic transmission by each stockholder entitled to vote on the matter and filed with the minutes of proceedings of the stockholders or (b) if the action is advised, and submitted to the stockholders for approval, by the Board of Directors and a consent in writing or by electronic transmission of stockholders entitled to cast not less than the minimum number of votes that would be necessary to authorize or take the action at a meeting of stockholders at which all stockholders entitled to vote on the matter are present and voted is delivered to the Corporation in accordance with the MGCL. The Corporation shall give notice of any action taken by less than unanimous consent to each stockholder not later than ten days after the effective time of such action.

ARTICLE III

DIRECTORS

Section 1. <u>GENERAL POWERS</u>. The business and affairs of the Corporation shall be managed under the direction of the Board of Directors.

Section 2. <u>NUMBER, TENURE, QUALIFICATIONS AND RESIGNATION</u>. A majority of the entire Board of Directors may establish, increase or decrease the number of directors, provided that the number thereof shall never be less than the minimum number required by the MGCL, nor more than 15, and further provided that the tenure of office of a director shall not be affected by any decrease in the number of directors. Any director of the Corporation may resign at any time by delivering his or her resignation to the Board of Directors, the chair of the board or the secretary. Any resignation shall take effect immediately upon its receipt or at such later time specified in the resignation. The acceptance of a resignation shall not be necessary to make it effective unless otherwise stated in the resignation. So long as the Advisory Agreement remains in effect, it shall be a qualification of at least two directors of the Corporation that he or she shall have been nominated by NRI.

Section 3. <u>ANNUAL AND REGULAR MEETINGS</u>. An annual meeting of the Board of Directors may be held at such time and place as shall be specified in a notice given as hereinafter provided for special meetings of the Board of Directors. The Board of Directors may provide, by resolution, the time and place of regular meetings of the Board of Directors without other notice than such resolution.

Section 4. <u>SPECIAL MEETINGS</u>. Special meetings of the Board of Directors may be called by or at the request of the chair of the board, the chief executive officer, the president or a majority of the directors then in office. The person or persons authorized to call special meetings of the Board of Directors may fix the time and place of any special meeting of the Board of Directors called by them. The Board of Directors may provide, by resolution, the time and place of special meetings of the Board of Directors without other notice than such resolution.

11

Section 5. Notice of any special meeting of the Board of Directors shall be delivered personally or by telephone, electronic mail, facsimile transmission, courier or United States mail to each director at his or her business or residence address. Notice by personal delivery, telephone, electronic mail or facsimile transmission shall be given at least 24 hours prior to the meeting. Notice by United States mail shall be given at least three days prior to the meeting. Notice by courier shall be given at least two days prior to the meeting. Telephone notice shall be deemed to be given when the director or his or her agent is personally given such notice in a telephone call to which the director or his or her agent is a party. Electronic mail notice shall be deemed to be given upon transmission of the message to the electronic mail address given to the Corporation by the director. Facsimile transmission notice shall be deemed to be given upon completion of the transmission of the message to the number given to the Corporation by the director and receipt of a completed answer-back indicating receipt. Notice by United States mail shall be deemed to be given when deposited in the United States mail properly addressed, with postage thereon prepaid. Notice by courier shall be deemed to be given when deposited with or delivered to a courier properly addressed. Neither the business to be transacted at, nor the purpose of, any annual, regular or special meeting of the Board of Directors need be stated in the notice, unless specifically required by statute or these Bylaws.

Section 6. <u>QUORUM</u>. A majority of the directors shall constitute a quorum for the transaction of business at any meeting of the Board of Directors, provided that, if less than a majority of such directors is present at such meeting, a majority of the directors present may adjourn the meeting from time to time without further notice, and provided further that if, pursuant to applicable law, the Charter or these Bylaws, the vote of a majority or other percentage of a specified group of directors is required for action, a quorum must also include a majority or such other percentage of such group.

The directors present at a meeting which has been duly called and at which a quorum has been established may continue to transact business until adjournment, notwithstanding the withdrawal from the meeting of enough directors to leave fewer than required to establish a quorum.

Section 7. <u>VOTING</u>. The action of a majority of the directors present at a meeting at which a quorum is present shall be the action of the Board of Directors, unless the concurrence of a greater proportion is required for such action by applicable law, the Charter or these Bylaws. If enough directors have withdrawn from a meeting to leave fewer than required to establish a quorum, but the meeting is not adjourned, the action of the majority of that number of directors necessary to constitute a quorum at such meeting shall be the action of the Board of Directors, unless the concurrence of a greater proportion is required for such action by applicable law, the Charter or these Bylaws.

Section 8. <u>ORGANIZATION</u>. At each meeting of the Board of Directors, the chair of the board or, in the absence of the chair, the chief executive officer shall act as chair of the meeting. Even if present at the meeting, the director named herein may designate another director to act as chair of the meeting. In the absence of both the chair of the board and the chief executive officer, or, in the absence of all such individuals, the president or, in the absence of the president, a director chosen by a majority of the directors present, shall act as chair of the meeting. The secretary or, in his or her absence, an assistant secretary of the Corporation, or, in the absence of the secretary and all assistant secretaries, an individual appointed by the chair of the meeting, shall act as secretary of the meeting.

Section 9. <u>TELEPHONE MEETINGS</u>. Directors may participate in a meeting by means of a conference telephone or other communications equipment if all persons participating in the meeting can hear each other at the same time. Participation in a meeting by these means shall constitute presence in person at the meeting.

Section 10. <u>CONSENT BY DIRECTORS WITHOUT A MEETING</u>. Any action required or permitted to be taken at any meeting of the Board of Directors may be taken without a meeting, if a consent in writing or by electronic transmission to such action is given by each director and is filed with the minutes of proceedings of the Board of Directors.

Section 11. <u>VACANCIES</u>. If for any reason any or all of the directors cease to be directors, such event shall not terminate the Corporation or affect these Bylaws or the powers of the remaining directors hereunder. Until such time as the Corporation becomes subject to Section 3-804(c) of the MGCL and subject to Section 2 of this Article III, any vacancy on the Board of Directors for any cause other than an increase in the number of directors may be filled by a majority of the remaining directors, even if such majority is less than a quorum; any vacancy in the number of directors created by an increase in the number of directors may be filled by a majority of the entire Board of Directors; and any individual so elected as director shall serve until the next annual meeting of stockholders and until his or her successor is duly elected and qualifies. At such time as the Corporation becomes subject to Section 3-804(c) of the MGCL and except as may be provided by the Board of Directors in setting the terms of any class or series of preferred stock, any vacancy on the Board of Directors may be filled only by a majority of the remaining directors, even if the remaining directors do not constitute a quorum, and any director elected to fill a vacancy shall serve for the remainder of the full term of the class in which the vacancy occurred and until a successor is duly elected and qualifies. So long as NRI has a right to nominate one or more directors of the Corporation, any director elected to fill a vacancy on the Board of Directors must satisfy the qualifications set forth in Section 2 of this Article III.

Section 12. <u>COMPENSATION</u>. Directors shall not receive any stated salary for their services as directors but, by resolution of the Board of Directors, may receive compensation per year and/or per meeting and/or per visit to real property or other facilities owned or leased by the Corporation and for any service or activity they performed or engaged in as directors. Directors may be reimbursed for expenses of attendance, if any, at each annual, regular or special meeting of the Board of Directors or of any committee thereof and for their expenses, if any, in connection with each property visit and any other service or activity they perform or engage in as directors; but nothing herein contained shall be construed to preclude any directors from serving the Corporation in any other capacity and receiving compensation therefor.

13

Section 13. <u>RELIANCE</u>. Each director and officer of the Corporation shall, in the performance of his or her duties with respect to the Corporation, be entitled to rely on any information, opinion, report or statement, including any financial statement or other financial data, prepared or presented by an officer or employee of the Corporation whom the director or officer reasonably believes to be reliable and competent in the matters presented, by a lawyer, certified public accountant or other person, as to a matter which the director or officer reasonably believes to be within the person's professional or expert competence, or, with respect to a director, by a committee of the Board of Directors on which the director does not serve, as to a matter within its designated authority, if the director reasonably believes the committee to merit confidence.

Section 14. <u>RATIFICATION</u>. The Board of Directors or the stockholders may ratify any act, omission, failure to act or determination made not to act (an "Act") by the Corporation or its officers to the extent that the Board of Directors or the stockholders could have originally authorized the Act and, if so ratified, such Act shall have the same force and effect as if originally duly authorized, and such ratification shall be binding upon the Corporation and its stockholders. Any Act questioned in any proceeding on the ground of lack of authority, defective or irregular execution, adverse interest of a director, officer or stockholder, non-disclosure, miscomputation, the application of improper principles or practices of accounting or otherwise, may be ratified, before or after judgment, by the Board of Directors or by the stockholders, and such ratification shall constitute a bar to any claim or execution of any judgment in respect of such questioned Act.

Section 15. <u>CERTAIN RIGHTS OF DIRECTORS</u>. Any director, in his or her personal capacity or in a capacity as an affiliate, employee, or agent of any other person, or otherwise, may have business interests and engage in business activities similar to, in addition to or in competition with those of or relating to the Corporation.

Section 16. <u>EMERGENCY PROVISIONS</u>. Notwithstanding any other provision in the Charter or these Bylaws, this Section 16 shall apply during the existence of any catastrophe, or other similar emergency condition, as a result of which a quorum of the Board of Directors under Article III of these Bylaws cannot readily be obtained (an "Emergency"). During any Emergency, unless otherwise provided by the Board of Directors, (i) a meeting of the Board of Directors or a committee thereof may be called by any director or officer by any means feasible under the circumstances; (ii) notice of any meeting of the Board of Directors during such an Emergency may be given less than 24 hours prior to the meeting to as many directors and by such means as may be feasible at the time, including publication, television or radio; and (iii) the number of directors necessary to constitute a quorum shall be one-third of the entire Board of Directors.

ARTICLE IV

COMMITTEES

Section 1. <u>NUMBER, TENURE AND QUALIFICATIONS</u>. The Board of Directors may appoint from among its members an Executive Committee, an Audit Committee, a Compensation Committee, a Nominating and Corporate Governance Committee and one or more other committees, composed of one or more directors, to serve at the pleasure of the Board of Directors. In the absence of any member of any such committee, the members thereof present at any meeting, whether or not they constitute a quorum, may appoint another director to act in the place of such absent member.

14

Section 2. <u>POWERS</u>. The Board of Directors may delegate to any committee appointed under Section 1 of this Article any of the Board of Directors, except as prohibited by law. Except as may be otherwise provided by the Board of Directors, any committee may delegate some or all of its power and authority to one or more subcommittees, composed of one or more directors, as the committee deems appropriate in its sole discretion.

Section 3. <u>MEETINGS</u>. Notice of committee meetings shall be given in the same manner as notice for special meetings of the Board of Directors. A majority of the members of the committee shall constitute a quorum for the transaction of business at any meeting of the committee. The act of a majority of the committee members present at a meeting shall be the act of such committee. The Board of Directors, or in the absence of such designation, the applicable committee, may designate a chair of any committee, and such chair or, in the absence of a chair, any two members of any committee (if there are at least two members of the committee) may fix the time and place of its meeting unless the Board shall otherwise provide.

Section 4. <u>TELEPHONE MEETINGS</u>. Members of a committee of the Board of Directors may participate in a meeting by means of a conference telephone or other communications equipment if all persons participating in the meeting can hear each other at the same time. Participation in a meeting by these means shall constitute presence in person at the meeting.

Section 5. <u>CONSENT BY COMMITTEES WITHOUT A MEETING</u>. Any action required or permitted to be taken at any meeting of a committee of the Board of Directors may be taken without a meeting, if a consent in writing or by electronic transmission to such action is given by each member of the committee and is filed with the minutes of proceedings of such committee.

Section 6. <u>VACANCIES</u>. Subject to the provisions hereof, the Board of Directors shall have the power at any time to change the membership of any committee, to appoint the chair of any committee, to fill any vacancy, to designate an alternate member to replace any absent or disqualified member or to dissolve any such committee.

ARTICLE V

OFFICERS

Section 1. <u>GENERAL PROVISIONS</u>. The officers of the Corporation shall include a president, a secretary and a treasurer and may include a chair of the board, a vice chair of the board, a chief executive officer, one or more vice presidents, a chief operating officer, a chief financial officer, a chief investment officer, one or more assistant secretaries and one or more assistant treasurers. In addition, the Board of Directors may from time to time elect such other officers with such powers and duties as it shall deem necessary or appropriate. The officers of the Corporation shall be elected annually by the Board of Directors, except that the chief executive officer or president may from time to time appoint one or more vice presidents, assistant secretaries and assistant treasurers or other officers. Each officer shall serve until his or her successor is elected and qualifies or until his or her death, or his or her resignation or removal in the manner hereinafter provided. Any two or more offices except president and vice president may be held by the same person. Election of an officer or agent shall not of itself create contract rights between the Corporation and such officer or agent.

15

Section 2. <u>REMOVAL AND RESIGNATION</u>. Any officer or agent of the Corporation may be removed, with or without cause, by the Board of Directors if in its judgment the best interests of the Corporation would be served thereby, but such removal shall be without prejudice to the contract rights, if any, of the person so removed. Any officer of the Corporation may resign at any time by delivering his or her resignation to the Board of Directors, the chair of the board, the chief executive officer, the president or the secretary. Any resignation shall take effect immediately upon its receipt or at such later time specified in the resignation. The acceptance of a resignation shall not be necessary to make it effective unless otherwise stated in the resignation. Such resignation shall be without prejudice to the contract rights, if any, of the Corporation.

Section 3. <u>VACANCIES</u>. A vacancy in any office may be filled by the Board of Directors for the balance of the term.

Section 4. <u>CHAIR OF THE BOARD</u>. The Board of Directors may designate from among its members a chair of the board, who shall not, solely by reason of these Bylaws, be an officer of the Corporation. The Board of Directors may designate the chair of the board as an executive or non-executive chair. The chair of the board shall preside over the meetings of the Board of Directors. The chair of the board shall perform such other duties as may be assigned to him or her by these Bylaws or the Board of Directors.

Section 5. <u>CHIEF EXECUTIVE OFFICER</u>. The Board of Directors may designate a chief executive officer. In the absence of such designation, the chair of the board shall be the chief executive officer of the Corporation. The chief executive officer shall have general responsibility for implementation of the policies of the Corporation, as determined by the Board of Directors, and for the management of the business and affairs of the Corporation. He or she may execute any deed, mortgage, bond, contract or other instrument, except in cases where the execution thereof shall be expressly delegated by the Board of Directors or by these Bylaws to some other officer or agent of the Corporation or shall be required by law to be otherwise executed; and in general shall perform all duties incident to the office of chief executive officer and such other duties as may be prescribed by the Board of Directors from time to time.

Section 6. <u>CHIEF OPERATING OFFICER</u>. The Board of Directors may designate a chief operating officer. The chief operating officer shall have the responsibilities and duties as determined by the Board of Directors or the chief executive officer.

Section 7. <u>CHIEF FINANCIAL OFFICER</u>. The Board of Directors may designate a chief financial officer. The chief financial officer shall have the responsibilities and duties as determined by the Board of Directors or the chief executive officer.

Section 8. <u>CHIEF INVESTMENT OFFICER</u>. The Board of Directors may designate a chief investment officer. The chief investment officer shall have the responsibilities and duties as determined by the Board of Directors or the chief executive officer.

16

Section 9. <u>PRESIDENT</u>. In the absence of a chief executive officer, the president shall in general supervise and control all of the business and affairs of the Corporation. In the absence of a designation of a chief operating officer by the Board of Directors, the president shall be the chief operating officer. He or she may execute any deed, mortgage, bond, contract or other instrument, except in cases where the execution thereof shall be expressly delegated by the Board of Directors or by these Bylaws to some other officer or agent of the Corporation or shall be required by law to be otherwise executed; and in general shall perform all duties incident to the office of president and such other duties as may be prescribed by the Board of Directors from time to time.

Section 10. <u>VICE PRESIDENTS</u>. In the absence of the president or in the event of a vacancy in such office, the vice president (or in the event there be more than one vice president, the vice presidents in the order designated at the time of their election or, in the absence of any designation, then in the order of their election) shall perform the duties of the president and when so acting shall have all the powers of and be subject to all the restrictions upon the president; and shall perform such other duties as from time to time may be assigned to such vice president by the chief executive officer, the president or the Board of Directors. The Board of Directors may designate one or more vice presidents as executive vice president, senior vice president, or vice president for particular areas of responsibility.

Section 11. <u>SECRETARY</u>. The secretary shall (a) keep the minutes of the proceedings of the stockholders, the Board of Directors and committees of the Board of Directors in one or more books provided for that purpose; (b) see that all notices are duly given in accordance with the provisions of these Bylaws or as required by law; (c) be custodian of the corporate records and of the seal of the Corporation; (d) keep a register of the address of each stockholder which shall be furnished to the secretary by such stockholder; (e) have general charge of the stock transfer books of the Corporation; and (f) in general perform such other duties as from time to time may be assigned to him or her by the chief executive officer, the president or the Board of Directors.

Section 12. <u>TREASURER</u>. The treasurer shall have the custody of the funds and securities of the Corporation, shall keep full and accurate accounts of receipts and disbursements in books belonging to the Corporation, shall deposit all moneys and other valuable effects in the name and to the credit of the Corporation in such depositories as may be designated by the Board of Directors and in general perform such other duties as from time to time may be assigned to him or her by the chief executive officer, the president or the Board of Directors. In the absence of a designation of a chief financial officer by the Board of Directors, the treasurer shall be the chief financial officer of the Corporation.

The treasurer shall disburse the funds of the Corporation as may be ordered by the Board of Directors, taking proper vouchers for such disbursements, and shall render to the president and Board of Directors, at the regular meetings of the Board of Directors or whenever it may so require, an account of all his or her transactions as treasurer and of the financial condition of the Corporation.

17

Section 13. <u>ASSISTANT SECRETARIES AND ASSISTANT TREASURERS</u>. The assistant secretaries and assistant treasurers, in general, shall perform such duties as shall be assigned to them by the secretary or treasurer, respectively, or by the chief executive officer, the president or the Board of Directors.

Section 14. <u>COMPENSATION</u>. The compensation of the officers shall be fixed from time to time by or under the authority of the Board of Directors and no officer shall be prevented from receiving such compensation by reason of the fact that he or she is also a director.

ARTICLE VI

CONTRACTS, CHECKS AND DEPOSITS

Section 1. <u>CONTRACTS</u>. The Board of Directors may authorize any officer or agent to enter into any contract or to execute and deliver any instrument in the name of and on behalf of the Corporation and such authority may be general or confined to specific instances. Any agreement, deed, mortgage, lease or other document shall be valid and binding upon the Corporation when duly authorized or ratified by action of the Board of Directors and executed by an authorized person.

Section 2. <u>CHECKS AND DRAFTS</u>. All checks, drafts or other orders for the payment of money, notes or other evidences of indebtedness issued in the name of the Corporation shall be signed by such officer or agent of the Corporation in such manner as shall from time to time be determined by the Board of Directors.

Section 3. <u>DEPOSITS</u>. All funds of the Corporation not otherwise employed shall be deposited or invested from time to time to the credit of the Corporation as the Board of Directors, the chief executive officer, the president, the chief financial officer, or any other officer designated by the Board of Directors may determine.

ARTICLE VII

STOCK

Section 1. <u>CERTIFICATES</u>. Except as may be otherwise provided by the Board of Directors or any officer of the Corporation, stockholders of the Corporation are not entitled to certificates representing the shares of stock held by them. In the event that the Corporation issues shares of stock represented by certificates, such certificates shall be in such form as prescribed by the Board of Directors or a duly authorized officer, shall contain the statements and information required by the MGCL and shall be signed by the officers of the Corporation in any manner permitted by the MGCL. In the event that the Corporation issues shares of stock without certificates, to the extent then required by the MGCL the Corporation shall provide to the record holders of such shares a written statement of the information required by the MGCL to be included on stock certificates. There shall be no difference in the rights and obligations of stockholders based on whether or not their shares are represented by certificates.

18

Section 2. <u>TRANSFERS</u>. All transfers of shares of stock shall be made on the books of the Corporation in such manner as the Board of Directors or any officer of the Corporation may prescribe and, if such shares are certificated, upon surrender of certificates duly endorsed. The issuance of a new certificate upon the transfer of certificated shares is subject to the determination of the Board of Directors or an officer of the Corporation that such shares shall no longer be represented by certificates. Upon the transfer of any uncertificated shares, the Corporation shall provide to the record holders of such shares, to the extent then required by the MGCL, a written statement of the information required by the MGCL to be included on stock certificates.

The Corporation shall be entitled to treat the holder of record of any share of stock as the holder in fact thereof and, accordingly, shall not be bound to recognize any equitable or other claim to or interest in such share or on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise expressly provided by the laws of the State of Maryland.

Notwithstanding the foregoing, transfers of shares of any class or series of stock will be subject in all respects to the Charter and all of the terms and conditions contained therein.

Section 3. <u>REPLACEMENT CERTIFICATE</u>. Any officer of the Corporation may direct a new certificate or certificates to be issued in place of any certificate or certificates theretofore issued by the Corporation alleged to have been lost, destroyed, stolen or mutilated, upon the making of an affidavit of that fact by the person claiming the certificate to be lost, destroyed, stolen or mutilated; provided, however, if such shares have ceased to be certificated, no new certificate shall be issued unless requested in writing by such stockholder and the Board of Directors or an officer of the Corporation has determined that such certificates may be issued. Unless otherwise determined by an officer of the Corporation, the owner of such lost, destroyed, stolen or mutilated certificate or certificates, or his or her legal representative, shall be required, as a condition precedent to the issuance of a new certificate or certificates, to give the Corporation a bond in such sums as it may direct as indemnity against any claim that may be made against the Corporation.

Section 4. <u>FIXING OF RECORD DATE</u>. The Board of Directors may set, in advance, a record date for the purpose of determining stockholders entitled to notice of or to vote at any meeting of stockholders or determining stockholders entitled to receive payment of any dividend or the allotment of any other rights, or in order to make a determination of stockholders for any other proper purpose. Such record date, in any case, shall not be prior to the close of business on the day the record date is fixed and shall be not more than 90 days and, in the case of a meeting of stockholders, not less than ten days, before the date on which the meeting or particular action requiring such determination of stockholders of record is to be held or taken.

When a record date for the determination of stockholders entitled to notice of or to vote at any meeting of stockholders has been set as provided in this section, such record date shall continue to apply to the meeting if postponed or adjourned, except if the meeting is postponed or adjourned to a date more than 120 days after the record date originally fixed for the meeting, in which case a new record date for such meeting shall be determined as set forth herein.

Section 5. <u>STOCK LEDGER</u>. The Corporation shall maintain at its principal office or at the office of its counsel, accountants or transfer agent, an original or duplicate stock ledger containing the name and address of each stockholder and the number of shares of each class held by such stockholder.

Section 6. <u>FRACTIONAL STOCK</u>; <u>ISSUANCE OF UNITS</u>. The Board of Directors may authorize the Corporation to issue fractional shares of stock or authorize the issuance of scrip, all on such terms and under such conditions as it may determine. Notwithstanding any other provision of the Charter or these Bylaws, the Board of Directors may authorize the issuance of units consisting of different securities of the Corporation.

Section 7. <u>SECURITY TOKENS</u>. Ownership of shares of the Corporation's stock may be represented by "tokens," or entries in a distributed electronic network or database maintained by or on behalf of the Company in accordance with Sections 2-114 and 2-209 of the MGCL. The tokens, if issued, will be uncertificated, and the ownership and transfer of the tokens will be recorded on a distributed ledger.

ARTICLE VIII

ACCOUNTING YEAR

The Board of Directors shall have the power, from time to time, to fix the fiscal year of the Corporation by a duly adopted resolution.

ARTICLE IX

DISTRIBUTIONS

Section 1. <u>AUTHORIZATION</u>. Dividends and other distributions upon the stock of the Corporation may be authorized by the Board of Directors, subject to the provisions of law and the Charter. Dividends and other distributions may be paid in cash, property or stock of the Corporation, subject to the provisions of law and the Charter.

Section 2. <u>CONTINGENCIES</u>. Before payment of any dividend or other distribution, there may be set aside out of any assets of the Corporation available for dividends or other distributions such sum or sums as the Board of Directors may from time to time, in its sole discretion, think proper as a reserve fund for contingencies, for equalizing dividends, for repairing or maintaining any property of the Corporation or for such other purpose as the Board of Directors shall determine, and the Board of Directors may modify or abolish any such reserve.

ARTICLE X

INVESTMENT POLICY

Subject to the provisions of the Charter, the Board of Directors may from time to time adopt, amend, revise or terminate any policy or policies with respect to investments by the Corporation as it shall deem appropriate in its sole discretion.

20

ARTICLE XI

SEAL

Section 1. <u>SEAL</u>. The Board of Directors may authorize the adoption of a seal by the Corporation. The seal shall contain the name of the Corporation and the year of its incorporation and the words "Incorporated Maryland." The Board of Directors may authorize one or more duplicate seals and provide for the custody thereof.

Section 2. <u>AFFIXING SEAL</u>. Whenever the Corporation is permitted or required to affix its seal to a document, it shall be sufficient to meet the requirements of any law, rule or regulation relating to a seal to place the word "(SEAL)" adjacent to the signature of the person authorized to execute the document on behalf of the Corporation.

ARTICLE XII

WAIVER OF NOTICE

Whenever any notice of a meeting is required to be given pursuant to the Charter or these Bylaws or pursuant to applicable law, a waiver thereof in writing or by electronic transmission, given by the person or persons entitled to such notice, whether before or after the time stated therein, shall be deemed equivalent to the giving of such notice. Neither the business to be transacted at nor the purpose of any meeting need be set forth in the waiver of notice of such meeting, unless specifically required by statute. The attendance of any person at any meeting shall constitute a waiver of notice of such meeting, except where such person attends a meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting has not been lawfully called or convened.

ARTICLE XIII

EXCLUSIVE FORUM FOR CERTAIN LITIGATION

Unless the Corporation consents in writing to the selection of an alternative forum, the Circuit Court for Baltimore City, Maryland, or, if that Court does not have jurisdiction, the United States District Court for the District of Maryland, Northern Division, shall be the sole and exclusive forum for (a) any Internal Corporate Claim, as such term is defined in Section 1-101(q) of the MGCL, or any successor provision thereof, (b) any derivative action or proceeding brought on behalf of the Corporation, other than actions arising under federal securities laws, (c) any action asserting a claim of breach of any duty owed by any director or officer or other employee of the Corporation to the Corporation or to the stockholders of the Corporation, (d) any action asserting a claim against the Corporation or any director or officer or other employee of the Corporation arising pursuant to any provision of the MGCL or the Charter or these Bylaws, or (e) any other action asserting a claim against the Corporation or any director or officer or other employee of the Corporation that is governed by the internal affairs doctrine. None of the foregoing actions, claims or proceedings may be brought in any court sitting outside the State of Maryland unless the Corporation consents in writing to such court.

ARTICLE XIV

AMENDMENT OF BYLAWS

The Board of Directors shall have the exclusive power to adopt, alter or repeal any provision of these Bylaws and to make new Bylaws; provided, however, that, so long as one director of the Corporation has been nominated for election by NRI and is then serving on the Board of Directors (an "NRI Nominated Director"), the approval of at least one NRI Nominated Director shall be required to amend the proviso in the first sentence of Section 7 of Article II, the last sentence of Section 2 of Article III, the last sentence of Section 11 of Article III or this proviso.

Approved: September 30, 2022

AMENDED AND RESTATED LIMITED PARTNERSHIP AGREEMENT

OF

NRI REAL TOKEN LP

A DELAWARE LIMITED PARTNERSHIP

TABLE OF CONTENTS

		Page
ARTICLE 1 DEFINED TERMS		2
ARTICLE	E 2 PARTNERSHIP FORMATION AND IDENTIFICATION	12
2.1	Formation	12
2.2	Name, Office and Registered Agent	12
2.3	Partners	12
2.4	Term and Dissolution	12
2.5	Filing of Certificate and Perfection of Limited Partnership	13
2.6	Certificates Describing Partnership Units	13
ARTICLE	ARTICLE 3 BUSINESS OF THE PARTNERSHIP	
ARTICLE 4 CAPITAL CONTRIBUTIONS AND ACCOUNTS		15
4.1	Capital Contributions	15
4.2	Additional Capital Contributions and Issuances of Additional Partnership Interests	15
4.3	Additional Funding	17
4.4	Capital Accounts	17
4.5	Percentage Interests	18
4.6	No Interest On Contributions	18
4.7	Return Of Capital Contributions	18
4.8	No Third Party Beneficiary	18
ARTICLE 5 PROFITS AND LOSSES; DISTRIBUTIONS		19
5.1	Allocation of Profit and Loss.	19

5.2	Distributions	23
5.3	REIT Distribution Requirements	26
5.4	No Right to Distributions in Kind	26
5.5	Limitations on Return of Capital Contributions	26
5.6	Intentionally Omitted	26
5.7	Distributions Upon Liquidation	26
5.8	Substantial Economic Effect	26
ARTICLE 6 RIGHTS, OBLIGATIONS AND POWERS OF THE GENERAL PARTNER		27
6.1	Management of the Partnership.	27
6.2	Delegation of Responsibility	31
6.3	Indemnification and Exculpation of Indemnitees	32
6.4	Liability of the General Partner	33
6.5	Reimbursement of General Partner	35
6.6	Outside Activities	35
6.7	Employment or Retention of Affiliates	35
6.8	General Partner Participation	36
6.9	Title to Partnership Assets	36
6.10	Redemption and Exchanges of REIT Shares.	36
6.11	No Duplication of Fees or Expenses	36
	i	
ARTICLE	7 CHANGES IN GENERAL PARTNER	37
7.1	Transfer of the General Partner's Partnership Interest	37
7.2	Admission of a Substitute or Additional General Partner	39
7.3	Effect of Bankruptcy, Withdrawal, Death or Dissolution of a General Partner.	39
7.4	Removal of a General Partner	40
ARTICLE	8 RIGHTS AND OBLIGATIONS OF THE LIMITED PARTNERS	41
8.1	Management of the Partnership	41
0.0	D C4.	4.4

53

ARTICLE 11 AMENDMENT OF AGREEMENT

ARTICLE	12 GENERAL PROVISIONS	54
12.1	Notices	54
12.2	Survival of Rights	54
12.3	Additional Documents	54
12.4	Severability	54
12.5	Entire Agreement	54
12.6	Pronouns and Plurals	54
12.7	Headings	54
12.8	Counterparts	54
12.9	Governing Law	54

ii

EXHIBITS

EXHIBIT A — Partners, Contributions, Types of Interests, Type and Number of Units and Percentage Interests EXHIBIT B — Notice of Exercise of OP Unit Redemption Right

ANNEXES

ANNEX A — Designation of the Series A Preferred Units of NRI Real Token LP

ii

AMENDED AND RESTATED LIMITED PARTNERSHIP AGREEMENT OF NRI REAL TOKEN LP

This Amended and Restated Limited Partnership Agreement (this "Agreement") of NRI Real Token LP (the "Partnership") is entered into as of this 30th day of September, 2022, between NRI Real Estate Investment & Technology, Inc. (f/k/a NRI Real Token, Inc.), a Maryland corporation (in its capacity as general partner of the Partnership, together with its successors and assigns that are admitted to the Partnership as a general partner of the Partnership in accordance with the terms hereof, the "General Partner"), NRI Real Token Thesis LLC, a Delaware limited liability company (the "Special OP Unitholder") and the Limited Partners set forth on Exhibit A attached hereto. Capitalized terms used herein but not otherwise defined shall have the meanings given them in Article 1.

AGREEMENT

WHEREAS, Partnership, was formed on May 28, 2021 as a limited partnership under the laws of the State of Delaware, pursuant to a Certificate of Limited Partnership filed with the Office of the Secretary of State of the State of Delaware on May 28, 2021;

WHEREAS, the Limited Partnership Agreement of the Partnership was entered into by the parties thereto on November 19, 2021, as amended by the First Amendment to Limited Partnership Agreement of the Partnership on April 11, 2022 (the "Original Agreement");

WHEREAS, the General Partner has qualified as a real estate investment trust under the Internal Revenue Code of 1986, as amended, and intends to continue to so qualify;

WHEREAS, the General Partner was defined as the "REIT General Partner" under the Original Agreement, and the Special OP Unitholder was defined as the "Sponsor General Partner" under the Original Agreement;

WHEREAS, the General Partner and the Special OP Unitholder desire to (i) convert the General Partnership Interest of the "Sponsor General Partner" into a Limited Partnership Interest and assign all rights and obligations of the "Sponsor General Partner" under the Original Agreement to the General Partner as set forth in this Agreement and the General Partner is willing to assume all such rights and obligations and (ii) redesignate the "Special GP Interest" (as such term is defined in the Original Agreement) as the Special OP Units as set forth herein;

WHEREAS, (a) <u>Article 11</u> of the Original Agreement permits the General Partners to mutually consent to amend the Original Agreement without the consent of any Limited Partner, except in certain limited circumstances, and (b) <u>Section 11(c)</u> of the Original Agreement requires the consent of the "Sponsor General Partner" to amend <u>Sections 5.1(b)</u>, <u>5.2</u> and <u>6.12</u> of the Original Agreement;

WHEREAS, pursuant to the authority granted to the General Partner and Special OP Unitholder pursuant to <u>Article 11</u> of the Original Agreement, they mutually desire to amend and restate the Original Agreement in its entirety to include all amendments to date and all changes necessary to accomplish the goals of the fifth WHEREAS clause above; and

WHEREAS, the parties hereto wish to establish herein their respective rights and obligations in connection with all of the foregoing and certain other matters.

1

NOW, THEREFORE, in consideration of the foregoing, of mutual covenants between the parties hereto, and of other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

ARTICLE 1

DEFINED TERMS

The following defined terms used in this Agreement shall have the meanings specified below:

"ACT" means the Delaware Revised Uniform Limited Partnership Act, as it may be amended from time to time.

"ADDITIONAL FUNDS" has the meaning set forth in Section 4.3 hereof.

"ADDITIONAL SECURITIES" means any additional REIT Shares (other than REIT Shares issued in connection with a redemption pursuant to <u>Section 8.4</u> hereof) or rights, options, warrants or convertible or exchangeable securities containing the right to subscribe for or purchase REIT Shares, as set forth in <u>Section 4.2(a)(ii)</u>.

"ADMINISTRATIVE EXPENSES" means (i) all administrative and operating costs and expenses incurred by the Partnership, (ii) those administrative costs and expenses of the General Partner, including any salaries or other payments to directors, officers or employees of the General Partner, and any accounting and legal expenses of the General Partner, which expenses, the Partners have agreed, are expenses of the Partnership and not the General Partner, (iii) costs and expenses relating to the formation and continuity of existence and operation of the General Partner and any Subsidiaries thereof (which Subsidiaries shall, for purposes hereof, be included within the definition of General Partner), including taxes, fees and assessments associated therewith, (iv) costs and expenses relating to any Offering and registration of securities by the Partnership or and all statements, reports, fees and expenses incidental thereto, including, without limitation, underwriting discounts and selling commissions applicable to any such Offering, and any costs and expenses associated with any claims made by any holders of such securities or any underwriters or placement agents thereof, (v) costs and expenses associated with any repurchase of any securities by the General Partner, (vi) costs and expenses associated with the preparation and filing of any periodic or other reports and communications by the General Partner or the Partnership under federal, state or local laws or regulations, (vii) costs and expenses associated with compliance by the General Partner or the Partnership with laws, rules and regulations promulgated by any regulatory body, including the Commission and any securities exchange, (viii) costs and expenses associated with any 401(k) plan, incentive plan, bonus plan or other plan providing for compensation for the employees of the General Partner, (ix) costs and expenses incurred by the General Partner relating to any issuing or redemption of Partnership Interests and (x) all other operating or administrative costs of the General Partner incurred in the ordinary course of its business on behalf of or in connection with the Partnership.

"ADVISOR" or "ADVISORS" has the meaning set forth in the Advisory Agreement.

"ADVISORY AGREEMENT" means that certain Amended and Restated Advisory Agreement, dated as of even date herewith, among the General Partner, the Partnership and the Advisor pursuant to which the Advisor will direct or perform the functions outlined therein.

"AFFECTED GAIN" has the meaning set forth in Section 5.1(i)(ii) hereof.

2

"AFFILIATE" means, with respect to any Person, (i) any Person directly or indirectly owning, controlling or holding, with the power to vote, ten percent or more of the outstanding voting securities of such other Person; (ii) any Person ten percent or more of whose outstanding voting securities are directly or indirectly owned, controlled or held, with the power to vote, by such other Person; (iii) any Person directly or indirectly controlled by or under common control with such other Person; (iv) any executive officer, director, trustee or general partner of such other Person and (v) any legal entity for which such Person acts as an executive officer, director, trustee or general partner.

"AGREED VALUE" means the fair market value of a Partner's non-cash Capital Contribution as of the date of contribution as agreed to by such Partner and the General Partner. The names and addresses of the Partners, number of Partnership Units issued to each Partner, and the Agreed Value of any non-cash Capital Contributions as of the date of contribution are set forth on Exhibit A.

"AGREEMENT" has the meaning set forth in the preamble.

"APPLICABLE PERCENTAGE" has the meaning set forth in <u>Section 8.4(b)</u> hereof. "ASSET" means any asset (other than the Property), Mortgage, other debt or other investment (other than investments in bank accounts, money market funds or other current assets) owned by the General Partner, indirectly through the Partnership or one or more of its subsidiaries.

"BUSINESS DAY" means any day on which the New York Stock Exchange is open for trading.

"CAPITAL ACCOUNT" has the meaning set forth in Section 4.4 hereof.

"CAPITAL CONTRIBUTION" means the total amount of cash, cash equivalents, and the Agreed Value of any Property or other asset (other than cash) contributed or agreed to be contributed, as the context requires, to the Partnership by each Partner pursuant to the terms of this Agreement. Any reference to the Capital Contribution of a Partner shall include the Capital Contribution made by a predecessor holder of the Partnership Interest of such Partner.

"CARRYING VALUE" means, with respect to any asset of the Partnership, the asset's adjusted net basis for federal income tax purposes or, in the case of any asset contributed to the Partnership, the fair market value of such asset at the time of contribution, reduced by any amounts attributable to the inclusion of liabilities in basis pursuant to Section 752 of the Code, except that the Carrying Values of all assets may, at the discretion of the General Partner be adjusted to equal their respective fair market values (as determined by the General Partner), in accordance with the rules set forth in Regulations Section 1.704-1(b)(2)(iv)(f), as provided for in Section 4.4. In the case of any asset of the Partnership that has a Carrying Value that differs from its adjusted tax basis, the Carrying Value shall be adjusted by the amount of depreciation, depletion and amortization calculated for purposes of the allocations of net income and net loss pursuant to Article 5 hereof rather than the amount of depreciation, depletion and amortization determined for federal income tax purposes.

3

"CASH AMOUNT" means an amount of cash per Partnership Unit equal to the Value of the REIT Shares Amount on the date of receipt by the General Partner of a Notice of Redemption.

"CERTIFICATE" means any instrument or document that is required under the laws of the State of Delaware, or any other jurisdiction in which the Partnership conducts business, to be signed and sworn to by the Partners of the Partnership (either by themselves or pursuant to the power-of-attorney granted to the General Partner in Section 8.2 hereof) and filed for recording in the appropriate public offices within the State of Delaware or such other jurisdiction to perfect or maintain the Partnership as a limited partnership, to effect the admission, withdrawal, or substitution of any Partner of the Partnership, or to protect the limited liability of the Limited Partners as limited partners under the laws of the State of Delaware or such other jurisdiction.

"CHARTER" means the Articles of Incorporation of the General Partner filed with the Maryland State Department of Assessments and Taxation, as amended or restated from time to time.

"CLAWBACK" has the meaning set forth in Section 5.2(f) hereof.

"CLAWBACK UNITS" has the meaning set forth in Section 5.2(f) hereof.

"CODE" means the Internal Revenue Code of 1986, as amended from time to time, or any successor statute thereto. Reference to any particular provision of the Code shall mean that provision as in effect from time to time, as the same may be amended, and any successor provision thereto, as interpreted by any applicable regulations as in effect from time to time.

"COMMISSION" means the U.S. Securities and Exchange Commission.

"CONTRIBUTED PROPERTY" means each property, partnership interest, contract right or other asset, in such form as may be permitted by the Act, contributed or deemed contributed to the Partnership by any Partner, including any interest in any successor partnership occurring as a result of a termination of the Partnership pursuant to Section 708 of Code.

"CONVERSION FACTOR" means 1.0, provided that in the event that the General Partner (i) declares or pays a dividend on its outstanding REIT Shares in REIT Shares or makes a distribution to all holders of its outstanding REIT Shares in REIT Shares, (ii) subdivides its outstanding REIT Shares, or (iii) combines its outstanding REIT Shares into a smaller number of REIT Shares, the Conversion Factor shall be adjusted by multiplying the Conversion Factor by a fraction, the numerator of which shall be the number of REIT Shares issued and outstanding on the record date for such dividend, distribution, subdivision or combination (assuming for such purposes that such dividend, distribution, subdivision or combination has occurred as of such time), and the denominator of which shall be the actual number of REIT Shares (determined without the above assumption) issued and outstanding on such date and, provided further, that in the event that an entity other than an Affiliate of the General Partner shall become General Partner pursuant to any merger, consolidation or combination of the General Partner with or into another entity (the "Successor Entity"), the Conversion Factor shall be adjusted by multiplying the Conversion Factor by the number of shares of the Successor Entity into which one REIT Share is converted pursuant to such merger, consolidation or combination, determined as of the date of such merger, consolidation or combination. Any adjustment to the Conversion Factor shall become effective immediately after the effective date of such event retroactive to the record date, if any, for such event; provided, however, that if the General Partner receives a Notice of Redemption after the record date, but prior to the effective date of such dividend, distribution, subdivision or combination, the Conversion Factor shall be determined as if the General Partner had received the Notice of Redemption immediately prior to the record date for such dividend, distribution, subdivision or combination.

4

"CRYSTALLIZATION EVENT" means, in reference to a distribution of units to be made pursuant to section 5.2(b), the periodic valuation or revaluation of the Properties of the Partnership to reflect each asset's then fair market value, by a manner and with assistance of third-party professional consultants as determined by the General Partner in its reasonable discretion, for purposes of determining the value of additional OP Units to be distributed to the Special OP Unitholder.

"DEFAULTING LIMITED PARTNER" has the meaning set forth in Section 5.2(c) hereof.

"DESIGNATED INDIVIDUAL" has the meaning set forth in Section 10.5(a)(i) hereof.

"DIRECTOR" has the meaning set forth in the Charter.

"EFFECTIVE DATE" means the date as of which this Agreement is entered into.

"EVENT OF BANKRUPTCY" as to any Person means the filing of a petition for relief as to such Person as debtor or bankrupt under the Bankruptcy Code of 1978 or similar provision of law of any jurisdiction (except if such petition is contested by such Person and has been dismissed within 90 days); insolvency or bankruptcy of such Person as finally determined by a court proceeding; filing by such Person of a petition or application to accomplish the same or for the appointment of a receiver or a trustee for such Person or a substantial part of his assets; commencement of any proceedings relating to such Person as a debtor under any other reorganization, arrangement, insolvency, adjustment of debt or liquidation law of any jurisdiction, whether now in existence or hereinafter in effect, either by such Person or by another, provided that if such proceeding is commenced by another, such Person indicates his approval of such proceeding, consents thereto or acquiesces therein, or such proceeding is contested by such Person and has not been finally dismissed within 90 days.

"EXCEPTED HOLDER LIMIT" has the meaning set forth in the Charter.

"EXCHANGED REIT SHARES" has the meaning set forth in Section 6.10(b) hereof.

"GENERAL PARTNER" has the meaning set forth in the Preamble, and any Person who becomes a substitute or additional General Partner as provided herein, and any of their successors as General Partner.

"GENERAL PARTNER LOAN" has the meaning set forth in Section 5.2(c) hereof.

"GENERAL PARTNERSHIP INTEREST" means a Partnership Interest held by the General Partner.

4

"HURDLE RATE" has the meaning set forth in Section 5.2(f) hereof.

"INDEMNITEE" means the General Partner, the Advisor or any of its Affiliates or any employee, director or Affiliate of the General Partner or the Partnership.

"INDEPENDENT DIRECTORS" has the meaning set forth in the Charter.

"JOINT VENTURE" means those joint venture, co-investment, co-ownership or partnership arrangements in which the General Partner or any of its Subsidiaries is a co- venturer or general partner established to acquire or hold Assets.

"LIABILITY SHORTFALL" has the meaning set forth in Section 5.1(i)(iv) hereof.

"LIMITED PARTNER" means any Person named as a Limited Partner on Exhibit A attached hereto and any Person who becomes a Substitute Limited Partner, in such Person's capacity as a Limited Partner in the Partnership.

"LIMITED PARTNERSHIP INTEREST" means the ownership interest of a Limited Partner in the Partnership at any particular time, including the right of such Limited Partner to any and all benefits to which such Limited Partner may be entitled as provided in this Agreement and in the Act, together with the obligations of such Limited Partner to comply with all the provisions of this Agreement and of such Act, including, for the avoidance of doubt, the Special OP Units.

"MORTGAGES" means, in connection with mortgage financing provided, invested in, participated in or purchased, all of the notes, deeds of trust, security interests or other evidences of indebtedness or obligations, which are secured or collateralized by Real Property owned by the borrowers under such notes, deeds of trust, security interests or other evidences of indebtedness or obligations.

"NET SALES PROCEEDS" means, in the case of a transaction described in clause (i)(A) of the definition of Sale, the proceeds of any such transaction less the amount of selling expenses incurred by or on behalf of the General Partner or the Partnership, including all real estate commissions, closing costs and legal fees and expenses. In the case of a transaction described in clause (i)(B) of such definition, Net Sales Proceeds means the proceeds of any such transaction less the amount of selling expenses incurred by or on behalf of the General Partner or the Partnership, including any legal fees and expenses and other selling expenses incurred in connection with such transaction. In the case of a transaction described in clause (i)(C) of such definition, Net Sales Proceeds means the proceeds of any such transaction actually distributed to the Partnership from the Joint Venture less the amount of any selling expenses, including legal fees and expenses incurred by or on behalf of the General Partner (other than those paid by the Joint Venture). In the case of a transaction or

series of transactions described in clause (i)(D) of the definition of Sale, Net Sales Proceeds means the proceeds of any such transaction (including the aggregate of all payments under a Mortgage or in satisfaction thereof other than regularly scheduled interest payments) less the amount of selling expenses incurred by or on behalf of the General Partner or the Partnership, including all commissions, closing costs and legal fees and expenses. In the case of a transaction described in clause (i)(E) of such definition, Net Sales Proceeds means the proceeds of any such transaction less the amount of selling expenses incurred by or on behalf of the General Partner or the Partnership, including any legal fees and expenses and other selling expenses incurred in connection with such transaction. In the case of a transaction described in clause (ii) of the definition of Sale, Net Sales Proceeds means the proceeds of such transaction or series of transactions less all amounts generated thereby which are reinvested in one or more Properties within 180 days thereafter and less the amount of any real estate commissions, closing costs, and legal fees and expenses and other selling expenses incurred by or allocated to the General Partner or the Partnership in connection with such transaction or series of transactions. Net Sales Proceeds shall also include any amounts that the General Partner determines, in its discretion, to be economically equivalent to proceeds of a Sale. Net Sales Proceeds shall not include any reserves established by the General Partner in its sole discretion.

6

"NOTICE OF REDEMPTION" means the Notice of Exercise of OP Unit Redemption Right substantially in the form attached as Exhibit B hereto.

"OFFER" has the meaning set forth in Section 7.1(c)(ii) hereof.

"OFFERING" means the offer and sale of Partnership Interests to the public.

"OP UNIT" means a Partnership Unit which is designated as an OP Unit of the Partnership.

"OP UNIT REDEMPTION RIGHT" has the meaning set forth in Section 8.4(a) hereof.

"OP UNITHOLDER" means a Partner that holds OP Units.

"OWNERSHIP LIMIT" has the meaning set forth in the Charter.

"PARTNER" means any General Partner or Limited Partner.

"PARTNER NONRECOURSE DEBT MINIMUM GAIN" has the meaning set forth in Regulations Section 1.704-2(i). A Partner's share of Partner Nonrecourse Debt Minimum Gain shall be determined in accordance with Regulations Section 1.704-2(i)(5).

"PARTNERSHIP" has the meaning set forth in the recitals.

"PARTNERSHIP INTEREST" means an ownership interest in the Partnership held by either a Limited Partner or the General Partner and includes any and all benefits to which the holder of such a Partnership Interest may be entitled as provided in this Agreement, together with all obligations of such Person to comply with the terms and provisions of this Agreement.

"PARTNERSHIP LOAN" has the meaning set forth in Section 5.2(c) hereof.

"PARTNERSHIP MINIMUM GAIN" has the meaning set forth in Regulations Section 1.704-2(d). In accordance with Regulations Section 1.704-2(d), the amount of Partnership Minimum Gain is determined by first computing, for each Partnership nonrecourse liability, any gain the Partnership would realize if it disposed of the property subject to that liability for no consideration other than full satisfaction of the liability, and then aggregating the separately computed gains. A Partner's share of Partnership Minimum Gain shall be determined in accordance with Regulations Section 1.704-2(g)(1).

"PARTNERSHIP RECORD DATE" means the record date established by the General Partner for the distribution of cash pursuant to <u>Section 5.2</u> hereof, which record date shall be the same as the record date established by the General Partner for a distribution to its shareholders of some or all of its portion of such distribution.

"PARTNERSHIP REPRESENTATIVE" has the meaning set forth in Section 10.5(a)(i) hereof.

"PARTNERSHIP UNIT" means a fractional, undivided share of the Partnership Interests of all Partners issued hereunder, excluding the Partnership Interest represented by the Special OP Units, and includes OP Units, Series A Preferred Units and any other class or series of Partnership Units that may be established after the date hereof. The number of Partnership Units outstanding and the Percentage Interests represented by such Partnership Units, if any, shall be as set forth on Exhibit A, as such Exhibit may be amended from time to time.

"PERCENTAGE INTEREST" means the percentage ownership interest in the Partnership of each Partner, as determined by dividing the number of Partnership Units owned by a Partner by the total number of Partnership Units then outstanding (other than the Preferred Units). The Percentage Interest of each Partner shall be as set forth on Exhibit A, as such Exhibit may be amended from time to time.

"PERSON" means an individual, corporation, partnership, limited liability company, estate, trust (including a trust qualified under Sections 401(a) or 501(c)(17) of the Code), a portion of a trust permanently set aside for or to be used exclusively for the purposes described in Section 642(c) of the Code, association, private foundation within the meaning of Section 509(a) of the Code, joint stock company or other entity and also includes a group as that term is used for purposes of Section 13(d)(3) of the Securities Exchange Act of 1934, as amended from time to time, and a group to which an Excepted Holder Limit applies.

"PRECONTRIBUTION GAIN" has the meaning set forth in Section 5.1(i)(iii) hereof.

"PREFERRED UNITS" means all Partnership Units designated as preferred units by the General Partner from time to time in accordance with Section 4.2(a), including the Series A Preferred Units.

"PRIORITY RETURN" means an 8% cumulative, pre-tax annual return, compounded monthly.

"PROPERTY" means, as the context requires, all or a portion of the Real Property acquired by the Partnership, directly or indirectly, and commonly known as THesis Miami, located at 1350 S Dixie Hwy, Coral Gables, Florida 33146,

"PROPERTIES" means, as the context requires, all or a portion of the Property and each Real Property that may be acquired by the Partnership, directly or indirectly through joint venture or co-ownership arrangements or other partnership or investment activities.

8

"REAL PROPERTY" means land, rights in land (including leasehold interests), and any buildings, structures, improvements, furnishings, fixtures and equipment located on or used in connection with land and rights or interests in land.

"RECEIVED REIT SHARES" has the meaning set forth in Section 6.10(b) hereof.

"REDEMPTION" has the meaning set forth in Section 8.4(a).

"REGULATIONS" means the Federal income tax regulations promulgated under the Code, as amended and as hereafter amended from time to time. Reference to any particular provision of the Regulations shall mean that provision of the Regulations on the date hereof and any successor provision of the Regulations.

"REGULATORY ALLOCATIONS" has the meaning set forth in <u>Section 5.1(i)</u> hereof. "REIT" means a corporation, trust, association or other legal entity (other than a real estate syndication) that qualifies as a real estate investment trust under Sections 856 through 860 of the Code, and any successor or other provisions of the Code relating to real estate investment trusts (including provisions as to the attribution of ownership of beneficial interests therein) and the regulations promulgated thereunder.

"REIT SHARE" means one share of common stock, par value \$0.01 per share, of the General Partner (or Successor Entity, as the case may be), which may be held in book-entry, paper certificate or digital (represented by a Security Token) form.

"REIT SHARES AMOUNT" means, with respect to Tendered Units, a number of REIT Shares equal to the product of the number of OP Units offered for exchange by a Tendering Party, multiplied by the Conversion Factor, as adjusted to and including the Specified Redemption Date; provided that in the event the General Partner issues to all holders of REIT Shares rights, options, warrants or convertible or exchangeable securities entitling the shareholders to subscribe for or purchase REIT Shares, or any other securities or property (collectively, the "rights"), and the rights have not expired at the Specified Redemption Date, then the REIT Shares Amount shall also include the rights issuable to a holder of the REIT Shares Amount of REIT Shares on the record date fixed for purposes of determining the holders of REIT Shares entitled to rights.

"RELATED PARTY" means, with respect to any Person, any other Person whose ownership of shares of the General Partner's capital stock would be attributed to the first such Person under Code Section 544 (as modified by Code Section 856(h)(1)(B)).

"RESTRICTED OP UNITS" has the meaning set forth in Section 8.6.

"RESTRICTION NOTICE" has the meaning set forth in Section 8.4(e) hereof.

"SAFE HARBOR" means, the election described in the Safe Harbor Regulation, pursuant to which a partnership and all of its partners may elect to treat the fair market value of a partnership interest that is transferred in connection with the performance of services as being equal to the liquidation value of that interest.

9

"SAFE HARBOR ELECTION" means the election by a partnership and its partners to apply the Safe Harbor, as described in the Safe Harbor Regulation and Internal Revenue Service Notice 2005-43, issued on May 19, 2005.

"SAFE HARBOR REGULATION" means Proposed Treasury Regulations Section 1.83-3(1) issued on May 19, 2005.

"SALE" means (i) any transaction or series of transactions whereby: (A) the Partnership directly or indirectly (except as described in other subsections of this definition) sells, grants, transfers, conveys, or relinquishes its ownership of any Property or portion thereof, including the lease of any Property consisting of a building only, and including any event with respect to any Property which gives rise to a significant amount of insurance proceeds or condemnation awards; (B) the Partnership directly or indirectly (except as described in other subsections of this definition) sells, grants, transfers, conveys, or relinquishes its ownership of all or substantially all of the interest of the Partnership in any Joint Venture in which it is a co-venturer or partner; (C) any Joint Venture directly or indirectly (except as described in other subsections of this definition) in which the Partnership as a co-venturer or partner sells, grants, transfers, conveys, or relinquishes its ownership of any Property or portion thereof, including any event with respect to any Property which gives rise to insurance claims or condemnation awards; (D) the Partnership directly or indirectly (except as described in other subsections of this definition) sells, grants, conveys or relinquishes its interest in any Mortgage or portion thereof (including with respect to any Mortgage, all payments thereunder or in satisfaction thereof other than regularly scheduled interest payments) of amounts owed pursuant to such Mortgage and any event which gives rise to a significant amount of insurance proceeds or similar awards; or (E) the Partnership directly or indirectly (except as described in other subsections of this definition) sells, grants, transfers, conveys, or relinquishes its ownership of any other Asset not previously described in this definition or any portion thereof, but (ii) not including any transaction or series of transactions specified in clause (i)(A) through (E) above in which the proceeds of such transaction or series of transactions are reinvested by the Partnership in one or more Properties within 180 days thereafter.

"SECURITIES ACT" means the Securities Act of 1933, as amended from time to time, or any successor statute thereto. Reference to any provision of the Securities Act shall mean such provision as in effect from time to time, as the same may be amended, and any successor provision thereto, as interpreted by any applicable regulations as in effect from time to time.

"SECURITY TOKEN" means the digital representation of one REIT Share designated as an "NRI Real Estate Security Token."

"SERIES A PREFERRED UNITS" has the meaning set forth in Annex A.

"SERVICE" means the United States Internal Revenue Service.

"SPECIAL OP UNIT TERMINAL DISTRIBUTION" has the meaning set forth in Section 8.5(b)(iii) hereof.

"SPECIAL OP UNIT VALUE" has the meaning set forth in Section 8.5(b)(i) hereof.

"SPECIAL OP UNITHOLDER" has the meaning set forth in the Preamble.

"SPECIFIED REDEMPTION DATE" means the first business day of the month that is at least sixty (60) Business Days after the receipt by the General Partner of the Notice of Redemption.

"SUBSIDIARY" means, with respect to any Person, any corporation or other entity of which a majority of (i) the voting power of the voting equity securities or (ii) the outstanding equity interests is owned, directly or indirectly, by such Person.

"SUBSIDIARY PARTNERSHIP" means any partnership of which the partnership interests therein are owned directly or indirectly by the Partnership.

"SUBSTITUTE LIMITED PARTNER" means any Person admitted to the Partnership as a Limited Partner pursuant to <u>Section 9.3</u> hereof.

"SUCCESSOR ENTITY" has the meaning set forth in the definition of "Conversion Factor" contained herein.

"SURVIVOR" has the meaning set forth in Section 7.1(d) hereof.

"TAX ITEMS" has the meaning set forth in Section 5.1(i)(i) hereof.

"TENDERED UNITS" has the meaning set forth in Section 8.4(a) hereof.

"TENDERING PARTY" has the meaning set forth in <u>Section 8.4(a)</u> hereof.

"TERMINAL EVENT" has the meaning set forth in Section 8.5(a) hereof.

"TRANSACTION" has the meaning set forth in Section 7.1(c) hereof.

"TRANSFER" has the meaning set forth in Section 9.2(a) hereof.

"UNITHOLDERS" means all holders of Partnership Units.

"VALUE" means for the REIT Shares, the fair market value of the REIT Shares which will equal: (i) (a) if REIT Shares are traded, the average closing price per share for the previous thirty Business Days, or (b) an amount determined by reference to the market value of the REIT Shares based upon the average closing price, or average of bid and asked prices (if closing prices are not available) on one or more alternative trading systems throughout the United States or similar exchanges located elsewhere during the prior period of thirty (30) days, and (ii) if REIT Shares of are not traded, such price per REIT Share as a majority of the Directors of the General Partner determines in good faith.

"VALUATION MECHANISMS" has the meaning set forth in Section 8.5(b)(i) hereof.

ARTICLE 2 PARTNERSHIP FORMATION AND IDENTIFICATION

- **2.1 <u>Formation</u>**. The Partnership was formed as a limited partnership pursuant to the Act and all other pertinent laws of the State of Delaware, for the purposes and upon the terms and conditions set forth in this Agreement.
- 2.2 Name, Office and Registered Agent. The name of the Partnership is NRI Real Token LP, a Delaware limited partnership. The specified office and place of business of the Partnership shall be 1340 S. Dixie Highway, Coral Gables, FL 33134. The General Partner may at any time change the location of such office, provided the General Partner gives notice to the Partners of any such change. The name and address of the Partnership's registered agent is National Registered Agents, Inc., 1209 Orange Street, Wilmington, New Castle County, Delaware 19801. The sole duty of the registered agent as such is to forward to the Partnership any notice that is served on him as registered agent.

2.3 Partners.

- (a) The General Partner of the Partnership is NRI Real Estate Investment & Technology, Inc. (f/k/a/ NRI Real Token, Inc.), a Maryland corporation whose principal place of business is the same as that of the Partnership.
- (b) The Limited Partners are those Persons identified as Limited Partners on Exhibit A hereto, as amended from time to time.

2.4 Term and Dissolution.

- (a) The term of the Partnership shall continue in full force and effect until the first to occur of any of the following events:
 - (i) The occurrence of an Event of Bankruptcy as to the General Partner or the dissolution, removal or withdrawal of the General Partner unless the business of the Partnership is continued pursuant to Section 7.3(b) hereof;
 - (ii) The passage of ninety (90) days after the sale or other disposition of all or substantially all of the assets of the Partnership (provided that if the Partnership receives an installment obligation as consideration for such sale or other disposition, the Partnership shall continue, unless sooner dissolved under the provisions of this Agreement, until such time as such note or notes are paid in full); or

12

- (iii) The election by the General Partner that the Partnership should be dissolved.
- (b) Upon dissolution of the Partnership (unless the business of the Partnership is continued pursuant to Section 7.3(b) hereof), the General Partner (or its trustee, receiver, successor or legal representative) shall amend or cancel any Certificate(s) and liquidate the Partnership's assets and apply and distribute the proceeds thereof in accordance with Section 5.7 hereof. Notwithstanding the foregoing, the liquidating General Partner may either (i) defer liquidation of, or withhold from distribution for a reasonable time, any assets of the Partnership (including those necessary to satisfy the Partnership's debts and obligations), or (ii) distribute the assets to the Partners in kind.
- 2.5 Filing of Certificate and Perfection of Limited Partnership. The General Partner shall execute, acknowledge, record and file at the expense of the Partnership, any and all amendments to the Certificate(s) and all requisite fictitious name statements and notices in such places and jurisdictions as may be necessary to cause the Partnership to be treated as a limited partnership under, and otherwise to comply with, the laws of each state or other jurisdiction in which the Partnership conducts business.
- **2.6** <u>Certificates Describing Partnership Units and Special OP Units</u>. At the request of a Limited Partner, the General Partner, at its option, may issue (but in no way is obligated to issue) a certificate summarizing the terms of such Limited Partner's interest in the

Partnership, including the number of Partnership Units (and, if applicable, the Special OP Units) as of the date of such certificate. Any such certificate (i) shall be in form and substance as approved by the General Partner, (ii) shall not be negotiable and (iii) shall bear a legend to the following effect:

"This certificate is not negotiable. The Partnership Units and Special OP Units represented by this certificate are governed by and transferable only in accordance with the provisions of the Limited Partnership Agreement of NRI Real Token LP, as amended from time to time."

13

ARTICLE 3

BUSINESS OF THE PARTNERSHIP

The purpose and nature of the business to be conducted by the Partnership is (i) to conduct any business that may be lawfully conducted by a limited partnership organized pursuant to the Act, including, without limitation, acquiring, developing and managing the Property, which business shall be conducted in such a manner as to permit the General Partner at all times to qualify as a REIT, unless the General Partner otherwise ceases to qualify as a REIT, and in a manner such that the General Partner will not be subject to any taxes under Section 857 or 4981 of the Code, (ii) to enter into any partnership, joint venture, co- ownership or other similar arrangement to engage in any of the foregoing or the ownership of interests in any entity engaged in any of the foregoing and (iii) to do anything necessary or incidental to the foregoing. In connection with the foregoing, and without limiting the General Partner's right in its sole and absolute discretion to qualify or cease qualifying as a REIT, the Partners acknowledge that the General Partner intends to qualify as a REIT for federal income tax purposes and upon such qualification the avoidance of income and excise taxes on the General Partner inures to the benefit of all the Partners and not solely to the General Partner. Notwithstanding the foregoing, the Limited Partners agree that the General Partner may terminate its status as a REIT under the Code at any time to the full extent permitted under the Charter. The General Partner on behalf of the Partnership shall also be empowered to do any and all acts and things necessary or prudent to ensure that the Partnership will not be classified as a "publicly traded partnership" for purposes of Section 7704 of the Code.

14

ARTICLE 4

CAPITAL CONTRIBUTIONS AND ACCOUNTS

- **4.1** <u>Capital Contributions</u>. The General Partner, the Special OP Unitholder and the initial Limited Partners have made (or shall be deemed to have made) capital contributions to the Partnership in exchange for the Partnership Interests set forth opposite their names on <u>Exhibit A</u>, as such exhibit may be amended from time to time. The Partners shall own Partnership Units in the amounts set forth in <u>Exhibit A</u> and shall have a Percentage Interest in the Partnership as set forth in <u>Exhibit A</u>, which Percentage Interest shall be adjusted in <u>Exhibit A</u> from time to time by the General Partner to the extent necessary to reflect accurately exchanges, Redemptions, Capital Contributions, the issuance of additional Partnership Units or similar events having an effect on a Partner's Percentage Interest.
- 4.2 Additional Capital Contributions and Issuances of Additional Partnership Interests. Except as provided in this Section 4.2 or in Section 4.3, the Partners shall have no right or obligation to make any additional Capital Contributions or loans to the Partnership. The General Partner may contribute additional capital to the Partnership, from time to time, and receive additional Partnership Interests in respect thereof, in the manner contemplated in this Section 4.2.
 - (a) Issuances of Additional Partnership Interests.
 - (i) <u>General</u>. As of the effective date of this Agreement, the Partnership shall have two classes of Partnership Units, entitled "OP Units" and "Series A Preferred Units," and a Partnership Interest referred to as the "Special OP Units." The General Partner is hereby authorized to cause the Partnership to issue such additional Partnership

Interests in the form of Partnership Units for any Partnership purpose at any time or from time to time, including but not limited to Partnership Units issued in connection with acquisitions of properties, to the Partners (including the General Partner) or to other Persons for such consideration and on such terms and conditions as shall be established by the General Partner in its sole and absolute discretion, all without the approval of any Limited Partners. Any additional Partnership Interests issued may be issued in one or more classes, or one or more series of any of such classes, with such designations, preferences and relative, participating, optional or other special rights, powers and duties, including rights, powers and duties senior to Limited Partnership Interests, all as shall be determined by the General Partner in its sole and absolute discretion, and without the approval of any Limited Partner, subject to Delaware law, including, without limitation, (i) the allocations of items of Partnership income, gain, loss, deduction and credit to each such class or series of Partnership Interests to share in Partnership distributions; and (iii) the rights of each such class or series of Partnership Interests upon dissolution and liquidation of the Partnership; provided, however, that no additional Partnership Interests shall be issued to the General Partner unless:

(1) (A) the additional Partnership Interests are issued in connection with an issuance of REIT Shares of or other interests in the General Partner, which shares or interests have designations, preferences and other rights, all such that the economic interests are substantially similar to the designations, preferences and other rights of the additional Partnership Interests issued to the General Partner by the Partnership in accordance with this Section 4.2 and (B) the General Partner shall make a Capital Contribution to the Partnership in an amount equal to the proceeds raised in connection with the issuance of such shares of stock of or other interests in the General Partner;

15

- (2) (A) the additional Partnership Interests are issued in connection with an issuance of REIT Shares of or other interests in the General Partner pursuant to a taxable share dividend declared by the General Partner, which shares or interests have designations, preferences and other rights, all such that the economic interests are substantially similar to the designations, preferences and other rights of the additional Partnership Interests issued to the General Partner by the Partnership in accordance with this Section 4.2 and (B) if the General Partner allows the holders of its REIT Shares to elect whether to receive such dividend in REIT Shares, other interests of the General Partner or cash, the Partnership will give the Limited Partners (excluding the General Partner in its capacity as a Limited Partner) the same election to receive (I) Partnership Units or cash, or (II) at the election of the General Partner, REIT Shares or cash, and (C) if the Partnership issues additional Partnership Units pursuant to this Section 4.2(a)(i)(2), then an amount of income equal to the value of the Partnership Units received will be allocated to those Unitholders that elect to receive additional Partnership Units; or
- (3) the additional Partnership Interests are issued to all Partners holding Partnership Units in proportion to their respective Percentage Interests.
- (ii) <u>Upon Issuance of Additional Securities</u>. If the General Partner issues any Additional Securities (other than REIT Shares issued in connection with an exchange pursuant to <u>Section 8.4</u> or <u>8.5(b)</u> and a taxable share dividend as described in <u>Section 4.2(a)(i)(2)</u> hereof) other than to all holders of REIT Shares, (A) the General Partner shall cause the Partnership to issue to the General Partner, as the General Partner may designate, Partnership Interests or rights, options, warrants or convertible or exchangeable securities of the Partnership having designations, preferences and other rights, all such that the economic interests are substantially similar to those of the Additional Securities, and (B) the General Partner contributes the proceeds from the issuance of such Additional Securities and from any exercise of rights contained in such Additional Securities, directly and through the General Partner, to the Partnership. For example, in the event the General Partner issues REIT Shares for a cash purchase price and contributes all of the proceeds of such issuance to the Partnership, the General Partner shall be issued a number of additional Partnership Units equal to the product of (A) the number of such REIT Shares issued by the General Partner, the proceeds of which were so contributed, multiplied by (B) a fraction, the numerator of which is 100%, and the denominator of which is the Conversion Factor in effect on the date of such contribution.

- (b) <u>Certain Deemed Contributions of Proceeds of Issuance of REIT Shares</u>. Except as otherwise permitted hereunder, in connection with any and all issuances of REIT Shares, the General Partner shall make Capital Contributions to the Partnership of the proceeds therefrom, provided that if the proceeds actually received and contributed by the General Partner are less than the gross proceeds of such issuance as a result of any underwriter's discount or other expenses paid or incurred in connection with such issuance, then the General Partner shall be deemed to have made Capital Contributions to the Partnership in the aggregate amount of the gross proceeds of such issuance and the Partnership shall be deemed simultaneously to have paid such offering expenses in accordance with <u>Section 6.5</u> hereof and in connection with the required issuance of additional Partnership Units to the General Partner for such Capital Contributions pursuant to <u>Section 4.2(a)</u> hereof.
- (c) <u>Redemptions or Repurchases of REIT Shares</u>. If the General Partner shall redeem or repurchase shares of any class of its shares of common stock, the purchase price thereof and all costs incurred in connection with such redemption or repurchase shall be reimbursed to the General Partner by the Partnership pursuant to <u>Section 6.5</u> hereof and the General Partner shall cause the Partnership to redeem or repurchase an equivalent number of Partnership Units of the appropriate class or series held by the General Partner (which, in the case of REIT Shares, shall be a number equal to the quotient of the number of such REIT Shares divided by the Conversion Factor) in the manner provided in Section 6.10 hereof.
- 4.3 Additional Funding. If the General Partner determines that it is in the best interests of the Partnership to provide for additional Partnership funds ("Additional Funds") for any Partnership purpose, then the General Partner may (i) cause the Partnership to obtain such funds from outside borrowings, or (ii) elect to have the General Partner or any of its Affiliates provide such Additional Funds to the Partnership through loans or otherwise, provided, however, that the Partnership may not borrow money from such Affiliates on terms other than market based, arm's length commercially reasonable terms as determined by the General Partner in its reasonable discretion.

4.4 Capital Accounts.

(a) A separate capital account (each a "Capital Account") shall be maintained for each Partner in accordance with the rules of Treasury Regulations Section 1.704-1(b)(2)(iv), and this Section 4.4 and Article 5 shall be interpreted and applied in a manner consistent therewith. Whenever the Partnership would be permitted to adjust the Capital Accounts of the Partners pursuant to Treasury Regulations Section 1.704-1(b)(2)(iv)(f) to reflect revaluations of Partnership property, the Partnership may so adjust the Capital Accounts of the Partners. In the event that the Capital Accounts of the Partners are adjusted pursuant to Treasury Regulations Section 1.704-1(b)(2)(iv)(f) to reflect revaluations of Partnership property, (i) the Capital Accounts of the Partners shall be adjusted in accordance with Treasury Regulations Section 1.704-1(b)(2)(iv)(g) for allocations of depreciation, depletion, amortization and gain or loss, as computed for book purposes, with respect to such property, (ii) the Partners' distributive shares of depreciation, depletion, amortization and gain or loss, as computed for tax purposes, with respect to such property shall be determined so as to take account of the variation between the adjusted tax basis and book value of such property in the same manner as under Code Section 704(c), and (iii) the amount of upward and/or downward adjustments to the book value of the Partnership property shall be treated as income, gain, deduction and/or loss for purposes of applying the allocation provisions of Article 5. In the event that Code Section 704(c) applies to Partnership property, the Capital Accounts of the Partners shall be adjusted in accordance with Treasury Regulations Section 1.704-1(b)(2)(iv)(g) for allocations of depreciation, depletion, amortization and gain and loss, as computed for book purposes, with respect to such property.

17

- (b) Notwithstanding any provision herein to the contrary, any fees, expenses or other costs of the Partnership that are required to be paid by the General Partner without reimbursement and that are required to be treated as capital contributions to the Partnership for purposes of the Treasury Regulations promulgated under Section 704(b) of the Code, shall be added to the balance of the General Partner's Capital Account.
- **4.5** <u>Percentage Interests</u>. If the number of outstanding Partnership Units increases or decreases during a taxable year, each Partner's Percentage Interest shall be adjusted by the General Partner effective as of the effective date of each such increase or decrease

to a percentage equal to the number of Partnership Units held by such Partner divided by the aggregate number of Partnership Units outstanding after giving effect to such increase or decrease. If the Partners' Percentage Interests are adjusted pursuant to this Section 4.5, the net income and net losses (and items thereof) for the taxable year in which the adjustment occurs shall be allocated between the part of the year ending on the day when the Partnership's property is revalued by the General Partner and the part of the year beginning on the following day either (i) as if the taxable year had ended on the date of the adjustment or (ii) based on the number of days in each part. The General Partner, in its sole and absolute discretion, shall determine which method shall be used to allocate net income and net losses (or items thereof) for the taxable year in which the adjustment occurs. The allocation of net income and net losses (or items thereof) for the year shall be based on the Percentage Interests before adjustment, and the allocation of net income and net losses (or items thereof) for the later part shall be based on the adjusted Percentage Interests.

- **4.6 No Interest On Contributions.** No Partner shall be entitled to interest on its Capital Contribution.
- **4.7** <u>Return Of Capital Contributions</u>. No Partner shall be entitled to withdraw any part of its Capital Contribution or its Capital Account or to receive any distribution from the Partnership, except as specifically provided in this Agreement. Except as otherwise provided herein, there shall be no obligation to return to any Partner or withdrawn Partner any part of such Partner's Capital Contribution for so long as the Partnership continues in existence.
- 4.8 No Third Party Beneficiary. No creditor or other third party having dealings with the Partnership shall have the right to enforce the right or obligation of any Partner to make Capital Contributions or loans or to pursue any other right or remedy hereunder or at law or in equity, it being understood and agreed that the provisions of this Agreement shall be solely for the benefit of, and may be enforced solely by, the parties hereto and their respective successors and assigns. None of the rights or obligations of the Partnersherein set forth to make Capital Contributions or loans to the Partnership shall be deemed an asset of the Partnership for any purpose by any creditor or other third party, nor may such rights or obligations be sold, transferred or assigned by the Partnership or pledged or encumbered by the Partnership to secure any debt or other obligation of the Partnership or of any of the Partners. In addition, it is the intent of the parties hereto that no distribution to any Limited Partner shall be deemed a return of money or other property in violation of the Act. However, if any court of competent jurisdiction holds that, notwithstanding the provisions of this Agreement, any Limited Partner is obligated to return such money or property, such obligation shall be the obligation of such Limited Partner and not of the General Partner. Without limiting the generality of the foregoing, a deficit Capital Account of a Partner shall not be deemed to be a liability of such Partner nor an asset or property of the Partnership.

18

ARTICLE 5

PROFITS AND LOSSES; DISTRIBUTIONS

5.1 Allocation of Profit and Loss.

- (a) General.
- (i) Net income and net loss (or items thereof) of the Partnership for each fiscal year or other applicable period of the Partnership shall be allocated among the Unitholders in accordance with their respective Percentage Interests;
- (ii) Notwithstanding the foregoing, and subject only to the provisions of paragraphs (b) and (h) and, to the extent set forth in this clause (ii) below, paragraph (c), unless otherwise waived with respect to a particular calendar year or period, net income shall first be allocated to the Special OP Unitholder until it has received aggregate allocations of income for all fiscal years equal to the aggregate amount of distributions such holder is entitled to receive or has received with respect to such Special OP Units for such fiscal year and all prior fiscal years, provided that in the event the Special OP Unitholder's entitlement to income allocations in such fiscal year would be satisfied or waived pursuant to the allocations set forth in paragraph (c) below, then such allocations shall be made or waived pursuant to paragraph (c) below in lieu of the provisions of this clause (ii).
- (b) <u>General Partner Gross Income Allocations</u>. There shall be specially allocated to the General Partner an amount of (i) first, items of Partnership income and (ii) second, items of Partnership gain during each fiscal year or other applicable period, before any other allocations are made hereunder, in an amount equal to the excess, if any, of (A) the cumulative distributions

made to the General Partner under <u>Section 6.5(b)</u> hereof, other than distributions which would properly be treated as "guaranteed payments" or which are attributable to the reimbursement of expenses which would properly be deductible by the Partnership, over (B) the cumulative allocations of Partnership income and gain to the General Partner under this <u>Section 5.1(b)</u>.

(c) Special Allocation with Respect to Sales.

(i) In connection with the sale of all or substantially all of the assets of the Partnership, items of income, gain, credit, loss and deduction of the Partnership for each fiscal year or other applicable period from such sale, other than any such items allocated under Section 5.1(b), shall be allocated among the Partners in a manner that will, as nearly as possible (after giving effect to the allocations under Sections 5.1(a) and 5.1(d)) cause the Capital Account balance of each Partner at the end of such fiscal year or other applicable period to equal (A) the amount of the hypothetical distribution that such Partner would receive if the Partnership were liquidated on the last day of such period and all assets of the Partnership, including cash, were sold for cash equal to their Carrying Value, taking into account any adjustments thereto for such period, all liabilities of the Partnership were satisfied in full in cash according to their terms (limited with respect to each nonrecourse liability to the Carrying Value of the assets securing such liability) and Net Sales Proceeds (after satisfaction of such liabilities) were distributed in full pursuant to Section 5.2(b), minus (B) the sum of such Partner's share of Partnership Minimum Gain and Partner Nonrecourse Debt Minimum Gain and the amount, if any and without duplication, that the Partner would be obligated to contribute to the capital of the Partnership, all computed as of the date of the hypothetical sale of assets.

19

(ii) Following the third anniversary of the effective date of this Agreement, items of income, gain, credit, loss and deduction of the Partnership for each fiscal year or other applicable period from Sales, other than any such items allocated under Section 5.1(b), shall be allocated among the Partners in a manner that will, as nearly as possible (after giving effect to the allocations under Sections 5.1(a) and 5.1(d)) cause the Capital Account balance of each Partner at the end of such fiscal year or other applicable period to equal (i) the amount of the hypothetical distribution that such Partner would receive if the Partnership were liquidated on the last day of such period and all assets of the Partnership, including cash, were sold for cash equal to their Carrying Value, taking into account any adjustments thereto for such period, all liabilities of the Partnership were satisfied in full in cash according to their terms (limited with respect to each nonrecourse liability to the Carrying Value of the assets securing such liability) and Net Sales Proceeds (after satisfaction of such liabilities) were distributed in full pursuant to Section 5.2(b), minus (ii) the sum of such Partner's share of Partnership Minimum Gain and Partner Nonrecourse Debt Minimum Gain and the amount, if any and without duplication, that the Partner would be obligated to contribute to the capital of the Partnership, all computed as of the date of the hypothetical sale of assets.

Notwithstanding anything to the contrary, the Special OP Unitholder may waive its entitlement to all or a portion of such allocations pursuant to this Section 5.1(c) with respect to a calendar year or portion thereof.

(d) Nonrecourse Deductions; Minimum Gain Chargeback. Notwithstanding any provision to the contrary, (i) any expense of the Partnership that is a "nonrecourse deduction" within the meaning of Regulations Section 1.704-2(b)(1) shall be allocated in accordance with the Partners' respective Percentage Interests, (ii) any expense of the Partnership that is a "partner nonrecourse deduction" within the meaning of Regulations Section 1.704- 2(i)(2) shall be allocated to the Partner that bears the "economic risk of loss" with respect to the liability to which such deductions are attributable in accordance with Regulations Section 1.704- 2(i)(1), (iii) if there is a net decrease in Partnership Minimum Gain within the meaning of Regulations Section 1.704-2(f)(1) for any Partnership taxable year, then, subject to the exceptions set forth in Regulations Section 1.704-2(f) and the ordering rules contained in Regulations Section 1.704-2(j), and (iv) if there is a net decrease in Partner Nonrecourse Debt Minimum Gain within the meaning of Regulations Section 1.704-2(j)(4) for any Partnership taxable year, then, subject to the exceptions set forth in Regulations Section 1.704-2(j(g), items of gain and income shall be allocated among the Partners in accordance with Regulations Section 1.704-2(j(g), items of gain and income shall be allocated among the Partners in accordance with Regulations Section 1.704-2(j(d) and the ordering rules contained in Regulations Section 1.704-2(j). A Partner's "interest in partnership profits" for purposes of determining its share of the excess nonrecourse liabilities of the Partnership within the meaning of Regulations Section 1.752-3(a)(3) shall be such Partner's Percentage Interest.

- (e) Qualified Income Offset. If a Partner unexpectedly receives in any taxable year an adjustment, allocation, or distribution described in subparagraphs (4), (5), or (6) of Regulations Section 1.704-1(b)(2)(ii)(d) that causes or increases a deficit balance in such Partner's Capital Account that exceeds the sum of such Partner's shares of Partnership Minimum Gain and Partner Nonrecourse Debt Minimum Gain, as determined in accordance with Regulations Sections 1.704-2(g)(1) and 1.704-2(i)(5), such Partner shall be allocated specially for such taxable year (and, if necessary, later taxable years) items of income and gain in an amount and manner sufficient to eliminate such deficit Capital Account balance as quickly as possible as provided in Regulations Section 1.704-1(b)(2)(ii)(d). This Section 5.1(e) is intended to constitute a "qualified income offset" under Section 1.704-1(b)(2)(ii)(d) of the Regulations and shall be interpreted consistently therewith. After the occurrence of an allocation of income or gain to a Partner in accordance with this Section 5.1(e), to the extent permitted by Regulations Section 1.704-1(b), items of expense or loss shall be allocated to such Partner in an amount necessary to offset the income or gain previously allocated to such Partner under this Section 5.1(e).
- (f) Capital Account Deficits. Loss (or items of loss) shall not be allocated to a Limited Partner to the extent that such allocation would cause or increase a deficit in such Partner's Capital Account at the end of any fiscal year (after reduction to reflect the items described in Regulations Section 1.704-1(b)(2)(ii)(d)(4), (5) and (6)) to exceed the sum of such Partner's shares of Partnership Minimum Gain and Partner Nonrecourse Debt Minimum Gain, as determined in accordance with Regulations Sections 1.704-2(g)(1) and 1.704-2(i)(5). Any net loss in excess of that limitation shall be allocated to the General Partner. After the occurrence of an allocation of net loss to the General Partner in accordance with this Section 5.1(f), to the extent permitted by Regulations Section 1.704-1(b), income shall be allocated to such Partner in an amount necessary to offset the net loss previously allocated to such Partner under this Section 5.1(f).
- (g) Allocations Between Transferor and Transferee. If a Partner transfers any part or all of its Partnership Interest, the distributive shares of the various items of income and loss allocable among the Partners during such fiscal year of the Partnership shall be allocated between the transferor and the transferee Partner either (i) as if the Partnership's fiscal year had ended on the date of the transfer, or (ii) based on the number of days of such fiscal year that each was a Partner without regard to the results of Partnership activities in the respective portions of such fiscal year in which the transferor and the transferee were Partners. The General Partner, in its sole and absolute discretion, shall determine which method shall be used to allocate the distributive shares of the various items of income and loss between the transferor and the transferee Partner.

(h) <u>Curative Allocations</u>. The allocations set forth in <u>Sections 5.1(d)</u>, <u>5.1(e)</u> and <u>5.1(f)</u> of this Agreement (the "**Regulatory Allocations**") are intended to comply with certain requirements of the Regulations. The General Partner is authorized to offset all Regulatory Allocations either with other Regulatory Allocations or with special allocations of other items of Partnership income, gain, loss or deduction pursuant to this <u>Section 5.1(i)</u>. Therefore, notwithstanding any other provision of this <u>Section 5.1</u> (other than the Regulatory Allocations), the General Partner shall make such offsetting special allocations of Partnership income, gain, loss or deduction in whatever manner it deems appropriate so that, after such offsetting allocations are made, each Partner's Capital Account is, to the extent possible, equal to the Capital Account balance such Partner would have had if the Regulatory Allocations were not part of this Agreement and all Partnership items were allocated pursuant to <u>Sections 5.1(a)</u>, <u>5.1(b)</u>, <u>5.1(c)</u>, <u>5.1(g)</u> and <u>5.1(h)</u>.

(i) Tax Allocations.

(i) <u>Items of Income or Loss</u>. Except as is otherwise provided in this <u>Section 5.1</u>, an allocation of net income or net loss to a Partner shall be treated as an allocation to such Partner of the same share of each item of income, gain, loss, deduction and item of tax- exempt income or Section 705(a)(2)(B) expenditure (or item treated as such expenditure pursuant to Section 1.704-1(b)(2)(iv)(i) of the Regulations) ("**Tax Items**") that is taken into account in computing net income or net loss.

- (ii) Section 1245/1250 Recapture. Subject to Section 5.1(i)(iii) below, if any portion of gain from the sale of Partnership assets is treated as gain which is ordinary income by virtue of the application of Sections 1245 or 1250 of the Code ("Affected Gain"), then such Affected Gain shall be allocated among the Partners in the same proportion that the depreciation and amortization deductions giving rise to the Affected Gain were allocated. This Section 5.1(i)(ii) shall not alter the amount of net income (or items thereof) allocated among the Partners, but merely the character of such net income (or items thereof). For purposes hereof, in order to determine the proportionate allocations of depreciation and amortization deductions for each fiscal year or other applicable period, such deductions shall be deemed allocated on the same basis as net income and net loss for such respective period.
- (iii) <u>Precontribution Gain, Revaluations</u>. With respect to any Contributed Property, the Partnership shall use any permissible method contained in the Regulations promulgated under Section 704(c) of the Code selected by the General Partner, in its sole discretion, to take into account any variation between the adjusted basis of such asset and the fair market value of such asset as of the time of the contribution ("**Precontribution Gain**"). Each Partner hereby agrees to report income, gain, loss and deduction on such Partner's U.S. federal income tax return in a manner consistent with the method used by the Partnership. If any asset has a Carrying Value which is different from the Partnership's adjusted basis for such asset for U.S. federal income tax purposes because the Partnership has revalued such asset pursuant to Section 1.704-1(b)(2)(iv)(f) of the Regulations, the allocations of Tax Items shall be made in accordance with the principles of Section 704(c) of the Code and the Regulations and the methods of allocation promulgated thereunder. The intent of this <u>Section 5.1(i)(iii)</u> is that each Partner who contributed to the capital of the Partnership a Contributed Property will bear, through reduced allocations of depreciation, increased allocations of gain or other items, the tax detriments associated with any Precontribution Gain. This <u>Section 5.1(i)(iii)</u> is to be interpreted consistently with such intent.

(iv) Excess Nonrecourse Liability Safe Harbor. Pursuant to Section 1.752-3(a)(3) of the Regulations, solely for purposes of determining each Partner's proportionate share of the "excess nonrecourse liabilities" of the Partnership (as defined in Section 1.752- 3(a)(3) of the Regulations), the Partners' respective interests in Partnership profits shall be determined under any permissible method reasonably determined by the General Partner; provided, however, that each Partner who has contributed an asset to the Partnership shall be allocated, to the extent possible, a share of "excess nonrecourse liabilities" of the Partnership which results in such Partner being allocated nonrecourse liabilities in an amount which is at least equal to the amount of income required to be allocated to such Partner pursuant to Section 704(c) of the Code and the Regulations promulgated thereunder (the "Liability Shortfall"). If there is an insufficient amount of nonrecourse liabilities to be able to allocate to each Partner nonrecourse liabilities equal to the Liability Shortfall, nonrecourse liabilities shall be allocated to each Partner in pro rata in accordance with each such Partner's Liability Shortfall.

5.2 <u>Distributions</u>.

- (a) The Partnership may distribute cash on a quarterly (or, at the election of the General Partner, more frequent) basis, to the extent of available cash, in an amount determined by the General Partner in its sole discretion, to the Partners who are Partners on the Partnership Record Date with respect to such quarter (or other distribution period) in accordance with Section 5.2(b); provided, however, that if a new or existing Partner acquires an additional Partnership Interest in exchange for a Capital Contribution on any date other than a Partnership Record Date, the cash distribution attributable to such additional Partnership Interest relating to the Partnership Record Date next following the issuance of such additional Partnership Interest shall be reduced in the proportion equal to one minus (i) the number of days that such additional Partnership Interest is held by such Partner bears to (ii) the number of days between such Partnership Record Date and the immediately preceding Partnership Record Date.
- (b) Except for distributions pursuant to Section 5.7 in connection with the dissolution and liquidation of the Partnership and subject to the provisions of Sections 5.2(c), 5.2(d), 5.3, 5.5 and 8.5, all distributions of cash shall be made:
 - (i) first, 100% to the OP Unitholders in accordance with their respective Percentage Interests on the Partnership Record Date until the OP Unitholders have received, or are deemed to have received in the circumstances set forth below in Section 5.2(c), cumulative distributions under this Section 5.2(b) equal to aggregate Capital

(ii) second, any additional amount distributable with respect to the OP Units shall be distributed (A) 75% to the OP Unitholders, pro rata in accordance with their respective Percentage Interests on the Partnership Record Date and (B) 25% to the Special OP Unitholder; provided that the OP Units held by the Special OP Unitholder and its Affiliates shall be entitled to 100% of such distributions;

provided, further, however, in the event that the Special OP Unitholder has received a Special OP Unit Terminal Distribution in accordance with Section 8.5(b)(iii) no further amount shall be distributed to the Special OP Unitholder until the OP Unitholders have collectively received aggregate distributions equal to the sum of (X) the amount such Unitholders are entitled to receive pursuant to Section 5.2(b)(i) plus (Y) an amount equal to 75% of (I) the portion of the Special OP Unit Terminal Distribution computed with respect to Section 5.2(b)(ii)(B) divided by (II) 25%.

(c) In connection with a Crystallization Event initiated pursuant to Section 6.1(d), the Special OP Unitholder shall be entitled to receive a distribution equal in value to the aggregate amount of distributions that would have been made with respect to the Special OP Units under Section 5.2(b)(ii) if all assets (subject to their liabilities) of the Partnership were sold for their fair market value, as determined in good faith by the General Partner, immediately prior to such time and in a similar manner to that set forth in Section 8.5(b)(i) below, including by applying the appropriate Valuation Mechanism, any remaining liabilities of the Partnership were satisfied in full in cash according to their terms and assuming for these purposes that all such remaining liabilities had matured, and Net Sales Proceeds (after satisfaction of such liabilities) were distributed in full pursuant to Section 5.2(b), provided that such distribution to the Special OP Unitholder shall be made in a whole number of OP Units equal to the quotient of (i) the aggregate amount of such distribution divided by (ii) the Value of one REIT Share divided by (iii) the Conversion Factor. For the avoidance of doubt, the amount the Special OP Unitholder shall be entitled to receive pursuant to this Section 5.2(c) with respect to any Crystallization Event, shall be based on the incremental value of the assets of the Partnership in excess of the highest value of the assets of the Partnership utilized in any preceding Crystallization Event or Terminal Event. Any OP Unit issued in connection with a Crystallization Event may be subject to claw back in accordance with Section 5.2(f).

24

(d) Notwithstanding any other provision of this Agreement, the General Partner is authorized to take any action that it determines to be necessary or appropriate to cause the Partnership to comply with any withholding requirements established under the Code or any other federal, state or local law including, without limitation, pursuant to Sections 1441, 1442, 1445, 1446, 1471, 1472 and 3406 of the Code. To the extent that the Partnership is required to withhold and pay over to any taxing authority any amount resulting from the allocation or distribution of income to any Partner or assignee (including by reason of Section 1446 of the Code), either (i) if the actual amount to be distributed to the Partner equals or exceeds the amount required to be withheld by the Partnership, the amount withheld shall be treated as a distribution of cash in the amount of such withholding to such Partner, or (ii) if the actual amount to be distributed to the Partner is less than the amount required to be withheld by the Partnership, the actual amount shall be treated as a distribution of cash in the amount of such withholding and the additional amount required to be withheld shall be treated as a loan (a "Partnership Loan") from the Partnership to the Partner on the day the Partnership pays over such amount to a taxing authority. A Partnership Loan shall be repaid through withholding by the Partnership with respect to subsequent distributions to the applicable Partner or assignee. In the event that a Limited Partner (a "Defaulting Limited Partner") fails to pay any amount owed to the Partnership with respect to the Partnership Loan within fifteen (15) days after demand for payment thereof is made by the Partnership on the Limited Partner, the General Partner, in its sole and absolute discretion, may elect to make the payment to the Partnership on behalf of such Defaulting Limited Partner. In such event, on the date of payment, the General Partner shall be deemed to have extended a loan (a "General Partner Loan") to the Defaulting Limited Partner in the amount of the payment made by the General Partner and shall succeed to all rights and remedies of the Partnership against the Defaulting Limited Partner as to that amount. Without limitation, the General Partner shall have the right to receive any distributions that otherwise would be made by the Partnership to the Defaulting Limited

Partner until such time as the General Partner Loan has been paid in full, and any such distributions so received by the General Partner shall be treated as having been received by the Defaulting Limited Partner and immediately paid to the General Partner. Any amounts treated as a Partnership Loan or a General Partner Loan pursuant to this Section 5.2(c) shall bear interest at the lesser of (i) the base rate on corporate loans at large United States money center commercial banks, as published from time to time in The Wall Street Journal, or (ii) the maximum lawful rate of interest on such obligation, such interest to accrue from the date the Partnership or the General Partner, as applicable, is deemed to extend the loan until such loan is repaid in full.

- (e) In no event may a Partner receive a distribution of cash with respect to a Partnership Unit if such Partner is entitled to receive a cash distribution as the holder of record of a REIT Share for which all or part of such Partnership Unit has been or will be exchanged.
- (f) Upon a Terminal Event causing the final liquidation of the Partnership, if the OP Unitholders would not receive (other than with respect to OP Units received by the Special OP Unitholder in accordance with Sections 5.2(c) and 8.5 (the "Clawback Units")) aggregate, cumulative distributions from the Partnership for all years from operating income, sales proceeds and other sources, including liquidating distributions, in an amount at least equal to (i) the sum of the aggregate Capital Contributions to the Partnership by the Unitholders for all years (excluding for this purpose Capital Contributions made for OP Units that have been redeemed for cash or REIT Shares in accordance with Section 8.4 prior to such final liquidation) plus (ii) a Priority Return on such Capital Contributions (the "Hurdle Rate"), the Clawback Units shall be subject to clawback (the "Clawback"). If the Clawback is invoked, the holder(s) of such Clawback Units shall be required to forfeit in aggregate the number of such Clawback Units equal in value to the lesser of (A) the amount required by the OP Unitholders to achieve the Hurdle Rate (other than with respect to the Clawback Units), and (B) all Clawback Units; provided that any Clawback Units shall first be clawed back from the Special OP Unitholder, then from any other Partner holding Clawback Units (other than the General Partner) pro rata and pari passu in accordance with the number of Clawback Units held by such Partners, and finally from the General Partner to the extent any Clawback Units had been previously redeemed for REIT Shares pursuant to Section 8.4; provided further, that any Clawback Units clawed back from the General Partner shall result in the concurrent forfeiture of REIT Shares received as a result of redemption of Clawback Units pursuant to Section 8.4 (as adjusted by the Conversion Factor, if necessary). Any Clawback Units so forfeited shall be treated as forfeited immediately prior to the final liquidation of the Partnership, no liquidating distributions shall be made with respect to the forfeited Clawback Units and the portion of the Capital Account associated with any forfeited Clawback Units shall be reallocated to the other OP Unitholders in the reasonable discretion of the General Partner. For the avoidance of doubt, any cash distributions previously made with respect to Clawback Units shall not be subject to Clawback.

25

- **5.3 REIT Distribution Requirements**. The General Partner shall use its commercially reasonable efforts to cause the Partnership to distribute amounts sufficient to enable the General Partner to make shareholder distributions that will allow the General Partner to (i) meet its distribution requirement for qualification as a REIT as set forth in Section 857 of the Code and (ii) avoid any federal income or excise tax liability imposed by the Code, other than to the extent the General Partner elects to retain and pay income tax on its net capital gain; provided, however, the General Partner shall not be bound to comply with this covenant to the extent any distribution required to be made in order to satisfy the requirements for qualification as a REIT under the Code and Regulations would violate the Act or other applicable law or contravene the terms of any notes, mortgages or other types of debt obligations to which the Partnership may be subject in conjunction with borrowed funds.
- **5.4** No Right to Distributions in Kind. No Partner shall be entitled to demand property other than cash in connection with any distributions by the Partnership.
- **5.5** <u>Limitations on Return of Capital Contributions</u>. Notwithstanding any of the provisions of this <u>Article 5</u>, no Partner shall have the right to receive and the General Partner shall not have the right to make, a distribution that includes a return of all or part of a Partner's Capital Contributions, unless after giving effect to the return of a Capital Contribution, the sum of all Partnership liabilities, other than the liabilities to a Partner for the return of his Capital Contribution, does not exceed the fair market value of the Partnership's assets.

5.6 Intentionally Omitted.

- **5.7** <u>Distributions Upon Liquidation</u>. Upon liquidation of the Partnership, after payment of, or adequate provision for, debts and obligations of the Partnership, including any Partner loans, any remaining assets of the Partnership shall be distributed to all Partners in proportion to their respective positive Capital Account balances, determined after taking into account all allocations required to be made pursuant to <u>Section 5.1</u> hereof and all prior distributions made pursuant to this <u>Article 5</u>, in compliance with Treasury Regulation Section 1.704-1(b)(2)(ii)(b)(2). Notwithstanding any other provision of this Agreement, the amount by which the value, as determined in good faith by the General Partner, of any property other than cash to be distributed in kind to the Partners exceeds or is less than the Carrying Value of such property shall, to the extent not otherwise recognized by the Partnership, be taken into account in computing net income and net loss of the Partnership (or items thereof) for purposes of crediting or charging the Capital Accounts of, and distributing proceeds to, the Partners, pursuant to this Agreement. To the extent deemed advisable by the General Partner, appropriate arrangements (including the use of a liquidating trust) may be made to assure that adequate funds are available to pay any contingent debts or obligations.
- **5.8** <u>Substantial Economic Effect</u>. It is the intent of the Partners that the allocations of net income and net loss (and items thereof) under this Agreement have substantial economic effect (or be consistent with the Partners' interests in the Partnership in the case of the allocation of losses attributable to nonrecourse debt) within the meaning of Section 704(b) of the Code as interpreted by the Regulations promulgated pursuant thereto. <u>Article 5</u> and other relevant provisions of this Agreement shall be interpreted in a manner consistent with such intent.

ARTICLE 6

RIGHTS, OBLIGATIONS AND POWERS OF THE GENERAL PARTNER

6.1 Management of the Partnership.

- (a) Except as otherwise expressly provided in this Agreement, the General Partner shall have full, complete and exclusive discretion to manage and control the business of the Partnership for the purposes herein stated, and shall make all decisions affecting the business and assets of the Partnership. Subject to the restrictions specifically contained in this Agreement, the powers of the General Partner shall include, without limitation, the authority to take the following actions on behalf of the Partnership:
 - (i) to acquire, purchase, own, operate, lease, dispose and exchange of any Property or other Assets, that the General Partner determines are necessary or appropriate or in the best interests of the business of the Partnership;
 - (ii) to construct buildings and make other improvements on the properties owned or leased by the Partnership;
 - (iii) to authorize, issue, sell, redeem or otherwise purchase any Partnership Interests or any securities (including secured and unsecured debt obligations of the Partnership, debt obligations of the Partnership convertible into any class or series of Partnership Interests, or options, rights, warrants or appreciation rights relating to any Partnership Interests) of the Partnership;
 - (iv) to borrow or lend money for the Partnership or the General Partner, issue or receive evidences of indebtedness in connection therewith, refinance, increase the amount of, modify, amend or change the terms of, or extend the time for the payment of, any such indebtedness, and secure such indebtedness by mortgage, deed of trust, pledge or other lien on the Partnership's assets; provided, however, that the entry of the Partnership into any such borrowing arrangements shall not cause the leverage ratio, as determined in accordance with the loan to value measurement metrics described in the extant Mortgage, to exceed 80%;
 - (v) to pay, either directly or by reimbursement, for all operating costs and general administrative expenses of the Partnership to third parties or to the General Partner or its Affiliates as set forth in this Agreement;
 - (vi) to guarantee or become a co-maker of indebtedness of the General Partner or any Subsidiary thereof, refinance, increase the amount of, modify, amend or change the terms of, or extend the time for the payment of, any such guarantee or indebtedness, and secure such guarantee or indebtedness by mortgage, deed of trust, pledge or other lien on the Partnership's assets;

- (vii) to use assets of the Partnership (including, without limitation, cash on hand) for any purpose consistent with this Agreement, including, without limitation, payment, either directly or by reimbursement, of all operating costs and general administrative expenses of the General Partner, the Partnership or any Subsidiary of either, to third parties or to the General Partner as set forth in this Agreement;
- (viii) to lease all or any portion of any of the Partnership's assets, whether or not the terms of such leases extend beyond the termination date of the Partnership and whether or not any portion of the Partnership's assets so leased are to be occupied by the lessee, or, in turn, subleased in whole or in part to others, for such consideration and on such terms as the General Partner may determine;
- (ix) to prosecute, defend, arbitrate, or compromise any and all claims or liabilities in favor of or against the Partnership, on such terms and in such manner as the General Partner may reasonably determine, and similarly to prosecute, settle or defend litigation with respect to the Partners, the Partnership, or the Partnership's assets;
- (x) to file applications, communicate, and otherwise deal with any and all governmental agencies having jurisdiction over, or in any way affecting, the Partnership's assets or any other aspect of the Partnership business;
 - (xi) to make or revoke any election permitted or required of the Partnership by any taxing authority;
- (xii) to maintain such insurance coverage for public liability, fire and casualty, and any and all other insurance for the protection of the Partnership, for the conservation of Partnership assets, or for any other purpose convenient or beneficial to the Partnership, in such amounts and such types, as it shall determine from time to time;
- (xiii) to determine whether or not to apply any insurance proceeds for any property to the restoration of such property or to distribute the same;
- (xiv) to establish one or more divisions of the Partnership, to hire and dismiss employees of the Partnership or any division of the Partnership, and to retain legal counsel, accountants, consultants, real estate brokers, and such other persons, as the General Partner may deem necessary or appropriate in connection with the Partnership business and to pay therefor such remuneration as the General Partner may deem reasonable and proper;
- (xv) to retain other services of any kind or nature in connection with the Partnership business, and to pay therefor such remuneration as the General Partner may deem reasonable and proper;
- (xvi) to negotiate and conclude agreements on behalf of the Partnership with respect to any of the rights, powers and authority conferred upon the General Partner;

- (xvii) to maintain accurate accounting records and to file promptly all federal, state and local income tax returns on behalf of the Partnership;
 - (xviii) to distribute Partnership cash or other Partnership assets in accordance with this Agreement;
- (xix) to form or acquire an interest in, and contribute property to, any further limited or general partnerships, joint ventures or other relationships that it deems desirable (including, without limitation, the acquisition of interests in, and the contributions of property to, its Subsidiaries and any other Person in which it has an equity interest from time to time);

- (xx) to establish Partnership reserves for working capital, capital expenditures, contingent liabilities, or any other valid Partnership purpose;
- (xxi) to do any and all acts and things necessary or prudent to ensure that the Partnership will not be classified as a "publicly traded partnership" for purposes of Section 7704 of the Code; and
- (xxii) to take such other action, execute, acknowledge, swear to or deliver such other documents and instruments, and perform any and all other acts that the General Partner deems necessary or appropriate for the formation, continuation and conduct of the business and affairs of the Partnership (including, without limitation, all actions consistent with allowing the General Partner at all times to qualify as a REIT unless the General Partner voluntarily terminates its REIT status) and to possess and enjoy all of the rights and powers of a general partner as provided by the Act.
- (b) Intentionally Omitted.
- (c) Except as otherwise provided herein, to the extent the duties of the General Partner require expenditures of funds to be paid to third parties, the General Partner shall not have any obligations hereunder except to the extent that partnership funds are reasonably available to it for the performance of such duties, and nothing herein contained shall be deemed to authorize or require the General Partner, in its capacity as such, to expend its individual funds for payment to third parties or to undertake any individual liability or obligation on behalf of the Partnership.
- (d) Notwithstanding anything contained in this Agreement to the contrary, the Special OP Unitholder shall be entitled from time to time to initiate a Crystallization Event for purposes of the Special OP Unitholder receiving a distribution of OP Units in accordance with Section 5.2(c); provided that the first Crystallization Event shall not be initiated prior to the second anniversary of the Effective Date of the Original Agreement, and any subsequent Crystallization Event shall not be initiated prior to the 24-month anniversary of the prior Crystallization Event. Any OP Units so received shall provide their holder with the right to redeem them pursuant to all of the terms and conditions of Section 8.4 hereof.

- (e) In the event the General Partner, the Partnership or any Subsidiary thereof is unable to secure financing from third-party lenders on satisfactory terms, it may be necessary for an Affiliate of the Special OP Unitholder to guarantee indebtedness in order to secure financing on satisfactory terms (such guaranteeing Person, a "Guarantor"). The Guarantor shall be entitled to a guaranty fee equal to one percent (1%) of the principal amount of any indebtedness incurred by the General Partner, the Partnership or any of their Subsidiaries with a third-party lender that is guaranteed by such Guarantor.
- (f) Notwithstanding anything contained in this Agreement to the contrary, the General Partner shall not, and shall not cause the Partnership to, (i) transfer the Property, whether by sale, disposition, merger, consolidation, contribution or by any other means, to any Affiliate controlled by the Special OP Unitholder or the Advisor, other than a wholly owned, direct or indirect, subsidiary of the Partnership or (ii) acquire any additional Properties, except as follows:
 - (i) During the period commencing on the twelve-month anniversary of the Effective Date and ending on the eighteen-month anniversary of the Effective Date, (A) the General Partner shall be permitted to transfer, or cause the Partnership to transfer, the Property by any means to any Affiliate controlled by the Special OP Unitholder or the Advisor; provided that the Partnership receives the greater of (I) the fair market value of the Property, as determined by an independent third party selected by the General Partner, or (II) a price such that the remaining OP Unitholders would receive a 1.25X multiple of their aggregate Capital Contributions with respect to their outstanding OP Units, or (B) in lieu of a transfer of the Property the General Partner shall have the right to offer to redeem any outstanding OP Units at a price per OP Unit that would be equal to the greater of (I) a price per OP Unit that would be obtained if the Partnership sold the Property (subject to its liabilities) at its fair market value, as determined by an independent third party selected by the General Partner, and distributed the proceeds in liquidation (after satisfaction of any remaining liabilities of the Partnership in full in cash according to their terms) or (II) a price such that the remaining OP Unitholders would receive a 1.25X multiple of their aggregate Capital Contributions with respect to their outstanding OP Units; provided the OP

Unitholders shall not be required to redeem their OP Units in connection with an offer to redeem OP Units pursuant to this Section 6.1(f)(i)(B);

(ii) Commencing on the day after the eighteen-month anniversary of the Effective Date and ending on the twenty four-month anniversary of the Effective Date, (A) the General Partner shall be permitted to transfer, or cause the Partnership to transfer, the Property by any means to any Affiliate controlled by the Special OP Unitholder or the Advisor; provided that Partnership receives the greater of (I) the fair market value of the Property, as determined by an independent third party selected by the General Partner, or (II) a price such that the remaining OP Unitholders would receive a 1.35X multiple of their aggregate Capital Contributions with respect to their outstanding OP Units, or (B) in lieu of a transfer of the Property the General Partner shall have the right to offer to redeem any outstanding OP Units at a price per OP Unit that would be equal to the greater of (I) a price per OP Unit that would be obtained if the Partnership sold the Property (subject to its liabilities) at its fair market value, as determined by an independent third party selected by the General Partner, and distributed the proceeds in liquidation (after satisfaction of any remaining liabilities of the Partnership in full in cash according to their terms) or (II) a price such that the remaining OP Unitholders would receive a 1.35X multiple of their aggregate Capital Contributions with respect to their outstanding OP Units; provided the OP Unitholders shall not be required to redeem their OP Units in connection with an offer to redeem OP Units pursuant to this Section 6.1(f)(ii)(B);

30

(iii) Commencing on the day after the twenty four-month anniversary of the Effective Date and at any time thereafter, the General Partner shall be permitted to (A) transfer, or cause the Partnership to transfer, the Property by any means to any Affiliate controlled by the Special OP Unitholder or the Advisor; provided that Partnership receives the fair market value of the Property, as determined by an independent third party selected by the General Partner, and (B) acquire, or cause the Partnership to acquire, additional Properties; provided that in either case the General Partner shall have offered to redeem any outstanding OP Units at a price per OP Unit that would be equal to a price per OP Unit that would be obtained if the Partnership sold the Property (subject to its liabilities) at its fair market value, as determined by an independent third party selected by the General Partner, and distributed the proceeds in liquidation (after satisfaction of any remaining liabilities of the Partnership in full in cash according to their terms); provided further that the OP Unitholders shall not be required to redeem their OP Units in connection with an offer to redeem OP Units pursuant to this Section 6.1(f)(iii); and

(iv) If the General Partner exercises its right to offer to redeem OP Units set forth above in Sections 6.1(f)(i)(B) and 6.1(f)(ii)(B), or offers to redeem OP Units as set forth above in Sections 6.1(f)(iii), upon the completion of any such redemption, regardless of whether any OP Unitholders participate in such redemption, the General Partner shall be permitted, and shall be permitted to cause the Partnership, to acquire additional Properties and sell any Partnership Properties.

For the avoidance of doubt, this <u>Section 6.1(f)</u> shall (i) not restrict the ability of the Partnership, or the General Partner to cause the Partnership, (A) to transfer the Property at any time to any Person or Persons that is not an Affiliate controlled by the Special OP Unitholder or the Advisor or (B) to acquire additional Properties subsequent to a transfer of the Property or the completion of an offer to redeem OP Units pursuant to this <u>Section 6.1(f)</u>, and (ii) apply only to OP Units other than OP Units held by the General Partner and the Special OP Unitholder (and, accordingly, does not apply to REIT Shares).

6.2 <u>Delegation of Responsibility</u>. The General Partner may delegate any or all of its powers, rights and obligations hereunder, and may appoint, employ, contract or otherwise deal with any Person for the transaction of the business of the Partnership, which Person may, under supervision of the General Partner, perform any acts or services for the Partnership as the General Partner may approve, including, for the avoidance of doubt, pursuant to the Advisory Agreement.

6.3 Indemnification and Exculpation of Indemnitees.

- (a) The Partnership shall indemnify an Indemnitee from and against any and all losses, claims, damages, liabilities, joint or several, expenses (including reasonable legal fees and expenses), judgments, fines, settlements, and other amounts arising from any and all claims, demands, actions, suits or proceedings, civil, criminal, administrative or investigative, that relate to the operations of the Partnership as set forth in this Agreement in which any Indemnitee may be involved, or is threatened to be involved, as a party or otherwise, unless it is established that: (i) the act or omission of the Indemnitee was material to the matter giving rise to the proceeding and either was committed in bad faith or was the result of active and deliberate dishonesty; (ii) the Indemnitee actually received an improper personal benefit in money, property or services; or (iii) in the case of any criminal proceeding, the Indemnitee had reasonable cause to believe that the act or omission was unlawful. Any indemnification pursuant to this Section 6.3 shall be made only out of the assets of the Partnership.
- (b) The Partnership shall reimburse an Indemnitee for reasonable expenses incurred by an Indemnitee who is a party to a proceeding in advance of the final disposition of the proceeding upon receipt by the Partnership of (i) a written affirmation by the Indemnitee's good faith belief that the standard of conduct necessary for indemnification by the Partnership as authorized in this Section 6.3 has been met, and (ii) a written undertaking by or on behalf of the Indemnitee to repay the amount if it shall ultimately be determined that the standard of conduct has not been met.
- (c) The indemnification provided by this <u>Section 6.3</u> shall be in addition to any other rights to which an Indemnitee or any other Person may be entitled under any agreement, pursuant to any vote of the Partners, as a matter of law or otherwise, and shall continue as to an Indemnitee who has ceased to serve in such capacity.
- (d) The Partnership may purchase and maintain insurance, on behalf of the Indemnitees and such other Persons as the General Partner shall determine, against any liability that may be asserted against or expenses that may be incurred by such Person in connection with the Partnership's activities, regardless of whether the Partnership would have the power to indemnify such Person against such liability under the provisions of this Agreement.
- (e) For purposes of this <u>Section 6.3</u>, the Partnership shall be deemed to have requested an Indemnitee to serve as fiduciary of an employee benefit plan whenever the performance by it of its duties to the Partnership also imposes duties on, or otherwise involves services by, it to the plan or participants or beneficiaries of the plan; excise taxes assessed on an Indemnitee with respect to an employee benefit plan pursuant to applicable law shall constitute fines within the meaning of this <u>Section 6.3</u>; and actions taken or omitted by the Indemnitee with respect to an employee benefit plan in the performance of its duties for a purpose reasonably believed by it to be in the interest of the participants and beneficiaries of the plan shall be deemed to be for a purpose which is not opposed to the best interests of the Partnership.

- (f) In no event may an Indemnitee subject the Limited Partners to personal liability by reason of the indemnification provisions set forth in this Agreement.
- (g) An Indemnitee shall not be denied indemnification in whole or in part under this <u>Section 6.3</u> because the Indemnitee had an interest in the transaction with respect to which the indemnification applies if the transaction was otherwise permitted by the terms of this Agreement.
- (h) The provisions of this <u>Section 6.3</u> are for the benefit of the Indemnitees, their heirs, successors, assigns and administrators and shall not be deemed to create any rights for the benefit of any other Persons.
- (i) Notwithstanding the foregoing, the Partnership may not indemnify or hold harmless an Indemnitee for any liability or loss unless all of the following conditions are met: (i) the Indemnitee has determined, in good faith, that the course of conduct that caused the loss or liability was in the best interests of the Partnership; (ii) the Indemnitee was acting on behalf of or performing services for the Partnership; (iii) the liability or loss was not the result of (A) negligence or misconduct, in the case that the Indemnitee is a director of the General Partner (other than an Independent Director), the Advisor or an Affiliate of the Advisor or (B) gross negligence or willful misconduct, in the case that the Indemnitee is an Independent Director; and (iv) the indemnification or agreement to hold harmless is recoverable only out of net assets of the Partnership. In addition, the Partnership shall not provide indemnification for any loss, liability or expense arising from or out of an alleged violation

of federal or state securities laws by such party unless one or more of the following conditions are met: (i) there has been a successful adjudication on the merits of each count involving alleged material securities law violations as to the Indemnitee; (ii) such claims have been dismissed with prejudice on the merits by a court of competent jurisdiction as to the Indemnitee; or (iii) a court of competent jurisdiction approves a settlement of the claims against the Indemnitee and finds that indemnification of the settlement and the related costs should be made, and the court considering the request for indemnification has been advised of the position of the Securities and Exchange Commission and of the published position of any state securities regulatory authority in which Securities were offered or sold as to indemnification for violations of securities laws.

6.4 Liability of the General Partner.

(a) Notwithstanding anything to the contrary set forth in this Agreement, the General Partner shall not be liable for monetary damages to the Partnership or any Partners for losses sustained or liabilities incurred as a result of errors in judgment or of any act or omission if the General Partner acted in good faith. The General Partner shall not be in breach of any duty that the General Partner may owe to the Limited Partners or the Partnership or any other Persons under this Agreement or of any duty stated or implied by law or equity provided the General Partner, acting in good faith, abides by the terms of this Agreement.

- (b) Each Limited Partner expressly acknowledges and agrees that whenever in this Agreement the General Partner is permitted to take any action, make any decision or determination or otherwise vote on or give its consent to any action, the General Partner shall be entitled to exercise its sole and absolute discretion in connection therewith after considering only such interests and factors as it desires and, without limiting the generality of the foregoing, it is specifically agreed and acknowledged that the General Partner in taking any action or declining to take any action hereunder may consider exclusively its own interests or the interests of its shareholders and shall have no duty or obligation to consider the separate interests of or factors affecting the Partnership or any other Partner (including, without limitation, the tax consequences to Limited Partners or the tax consequences of some, but not all, of the Limited Partners). The General Partner shall not be liable for monetary damages for losses sustained, liabilities incurred, or benefits not derived by Limited Partners in connection with such decisions, provided that the General Partner has acted in good faith.
- (c) Subject to its obligations and duties as General Partner set forth in <u>Section 6.1</u> hereof, the General Partner may exercise any of the powers granted to it under this Agreement and perform any of the duties imposed upon it hereunder either directly or by or through its agents. The General Partners shall not be responsible for any misconduct or negligence on the part of any such agent appointed by it in good faith.
- (d) Notwithstanding any other provisions of this Agreement or the Act, any action of the General Partner on behalf of the Partnership or any decision of the General Partner to refrain from acting on behalf of the Partnership, undertaken in the good faith belief that such action or omission is necessary or advisable in order (i) to protect the ability of the General Partner to continue to qualify as a REIT or (ii) to prevent the General Partner from incurring any taxes under Section 857, Section 4981, or any other provision of the Code, is expressly authorized under this Agreement and is deemed approved by all of the Limited Partners.
- (e) Any amendment, modification or repeal of this <u>Section 6.4</u> or any provision hereof shall be prospective only and shall not in any way affect the limitations on the General Partner's liability to the Partnership and the Limited Partners under this <u>Section 6.4</u> as in effect immediately prior to such amendment, modification or repeal with respect to matters occurring, in whole or in part, prior to such amendment, modification or repeal, regardless of when claims relating to such matters may arise or be asserted.
- (f) In accordance with Section 17-1101(d) of the Act, the Partners hereby acknowledge and agree that the provisions of this Agreement, including the provisions of this Article 6, to the extent they restrict or eliminate the duties (including fiduciary duties) and liabilities relating thereto otherwise existing at law or in equity, replace completely and absolutely such other duties (including fiduciary duties) and liabilities relating thereto and further acknowledge and agree that the provisions of this subsection (f) and the other provisions of this Article 6 are fundamental elements to the agreement of the Partners to enter into this Agreement.

6.5 Reimbursement of General Partner.

- (a) Except as provided in this <u>Section 6.5</u> and elsewhere in this Agreement (including the provisions of <u>Article 5</u> and <u>Article 6</u> regarding distributions, payments, and allocations to which it may be entitled), the General Partner shall not be compensated for its services as general partner of the Partnership.
- (b) The General Partner shall be reimbursed on a monthly basis, or such other basis as the General Partner may determine in its sole and absolute discretion, for all Administrative Expenses incurred by the General Partner.
- 6.6 Outside Activities. Subject to (a) Section 6.8 hereof, (b) the Charter and (c) any agreements entered into by the General Partner or its Affiliates with the Partnership, a Subsidiary or any officer, director, employee, agent, trustee, Affiliate or shareholder of the General Partner, the General Partner shall be entitled to and may have business interests and engage in business activities in addition to those relating to the Partnership, including business interests and activities substantially similar or identical to those of the Partnership. Neither the Partnership nor any of the Limited Partners shall have any rights by virtue of this Agreement in any such business ventures, interests or activities. None of the Limited Partners nor any other Person shall have any rights by virtue of this Agreement or the partnership relationship established hereby in any such business ventures, interests or activities, and the General Partner shall have no obligation pursuant to this Agreement to offer any interest in any such business ventures, interests and activities to the Partnership or any Limited Partner, even if such opportunity is of a character which, if presented to the Partnership or any Limited Partner, could be taken by such Person.

6.7 Employment or Retention of Affiliates.

- (a) Any Affiliate of the General Partner may be employed or retained by the Partnership and may otherwise deal with the Partnership (whether as a buyer, lessor, lessee, manager, furnisher of goods or services, broker, agent, lender or otherwise) and may receive from the Partnership any compensation, price, or other payment therefor which the General Partner determines to be fair and reasonable.
- (b) The Partnership may lend or contribute to its Subsidiaries or other Persons in which it has an equity investment, and such Persons may borrow funds from the Partnership, on terms and conditions established in the sole and absolute discretion of the General Partner. The foregoing authority shall not create any right or benefit in favor of any Subsidiary or any other Person.
- (c) The Partnership may transfer assets to joint ventures, other partnerships, corporations or other business entities in which it is or thereby becomes a participant upon such terms and subject to such conditions as the General Partner deems are consistent with this Agreement, applicable law and the REIT status of the General Partner.

- (d) Except as expressly permitted by this Agreement, neither the General Partner nor any of its Affiliates shall sell, transfer or convey any property to, or purchase any property from, the Partnership, directly or indirectly, except pursuant to transactions that are, in the General Partner's sole discretion, on terms that are market based, arm's length commercially reasonable.
- **6.8** <u>General Partner Participation</u>. The General Partner agrees that all business activities of the General Partner, including activities pertaining to the acquisition, development or ownership of any Asset, shall be conducted through the Partnership or one or more Subsidiary Partnerships.
- **6.9** <u>Title to Partnership Assets</u>. Title to Partnership assets, whether real, personal or mixed and whether tangible or intangible, shall be deemed to be owned by the Partnership as an entity, and no Partner, individually or collectively, shall have any ownership interest in such Partnership assets or any portion thereof. Title to any or all of the Partnership assets shall be held in the name of the Partnership or any of its Subsidiaries. All Partnership assets shall be recorded as the property of the Partnership in its books and records.

6.10 Redemption of REIT Shares. In the event the General Partner redeems any REIT Shares, then the General Partner shall cause the Partnership to purchase from the General Partner a number of Partnership Units as determined based on the application of the Conversion Factor on the same terms that the General Partner redeemed such REIT Shares. Moreover, if the General Partner makes a cash tender offer or other offer to acquire REIT Shares, then the General Partner shall cause the Partnership to make a corresponding offer to the General Partner to acquire an equal number of Partnership Units held by the General Partner that have the same class or series designation as the REIT Shares that are subject to the offer. In the event any REIT Shares are redeemed by the General Partner pursuant to such offer, the Partnership shall redeem an equivalent number of the General Partner's Partnership Units having the same class or series designation as the redeemed REIT Shares for an equivalent purchase price based on the application of the Conversion Factor for that class or series of Partnership Units.

6.11 No Duplication of Fees or Expenses. The Partnership may not incur or be responsible for any fee or expense (in connection with the Offering or otherwise) that would be duplicative of fees and expenses paid by the General Partner.

36

ARTICLE 7

CHANGES IN GENERAL PARTNER

7.1 <u>Transfer of the General Partner's Partnership Interest</u>.

- (a) The General Partner shall not transfer all or any portion of its General Partnership Interest or withdraw as General Partner except as provided in, or in connection with a transaction contemplated by, Sections 7.1(c), 7.1(d) or 7.1(e).
 - (b) The General Partner agrees that its Percentage Interest will at all times be in the aggregate, at least 0.1%.
- (c) Except as otherwise provided in Section 6.4(b) or Section 7.1(d) or 7.1(e) hereof, the General Partner shall not engage in any merger, consolidation or other combination with or into another Person or sale of all or substantially all of its assets (other than in connection with a change in the General Partner's state of incorporation or organizational form) in each case which results in a change of control of the General Partner (a "Transaction"), unless:
 - (i) the consent of Limited Partners holding more than 50% of the Percentage Interests and the consent of the Special OP Unitholder is obtained;
 - (ii) as a result of such Transaction: (A) all Limited Partners will receive for each Partnership Unit an amount of cash, securities, or other property equal to the product of the Conversion Factor and the greatest amount of cash, securities or other property paid in the Transaction to a holder of one REIT Share in consideration of such REIT Share, provided that if, in connection with the Transaction, a purchase, tender or exchange offer ("Offer") shall have been made to and accepted by the holders of more than 50% of the outstanding REIT Shares, each holder of Partnership Units shall be given the option to exchange its Partnership Units for the greatest amount of cash, securities, or other property which a Limited Partner holding Partnership Units would have received had it (1) exercised its OP Unit Redemption Right and (2) sold, tendered or exchanged pursuant to the Offer the REIT Shares received upon exercise of the OP Unit Redemption Right immediately prior to the expiration of the Offer and (B) the Special OP Unitholder will receive for the Special OP Units an amount of cash, securities or other property (as applicable based upon the type of consideration and the proportions thereof paid to holders of REIT Shares in the Transaction) determined as set forth pursuant to Section 5.2(b) or Section 8.5 hereof, as applicable; or
 - (iii) the General Partner is the surviving entity in the Transaction and either (A) the holders of REIT Shares do not receive cash, securities, or other property in the Transaction or (B) all Limited Partners (other than the General Partner or any Subsidiary) receive (1) in exchange for their Partnership Units, an amount of cash, securities, or other property (expressed as an amount per REIT Share) that is no less than the product of the Conversion Factor and the greatest amount of cash, securities, or other property (expressed as an amount per REIT Share) received in the Transaction by any holder of REIT Shares, and (2) the Special OP Unitholder receives in exchange for the Special OP Units, an amount of cash, securities or other property (as applicable based upon the type of consideration and the

(d) Notwithstanding Section 7.1(c), the General Partner may merge with or into or consolidate with another entity if immediately after such merger or consolidation (i) substantially all of the assets of the successor or surviving entity (the "Survivor"), other than Partnership Units held by the General Partner, are contributed, directly or indirectly, to the Partnership as a Capital Contribution in exchange for Partnership Units with a fair market value equal to the value of the assets so contributed as determined by the Survivor in good faith and (ii) the Survivor expressly agrees to assume all obligations of the General Partner, as appropriate, hereunder. Upon such contribution and assumption, the Survivor shall have the right and duty to amend this Agreement as set forth in this Section 7.1(d). The Survivor shall in good faith arrive at a new method for the calculation of the Cash Amount, the REIT Shares Amount and Conversion Factor for a Partnership Unit after any such merger or consolidation so as to approximate the existing method for such calculation as closely as reasonably possible. Such calculation shall take into account, among other things, the kind and amount of securities, cash and other property that was receivable upon such merger or consolidation by a holder of REIT Shares or options, warrants or other rights relating thereto, and which a holder of Partnership Units could have acquired had such Partnership Units been exchanged immediately prior to such merger or consolidation. Such amendment to this Agreement shall provide for adjustment to such method of calculation, which shall be as nearly equivalent as may be practicable to the adjustments provided for with respect to the Conversion Factor. The Survivor also shall in good faith modify the definition of REIT Shares and make such amendments to Sections 8.4 and 8.5 hereof so as to approximate the existing rights and obligations set forth in Sections 8.4 and 8.5 as closely as reasonably possible. The above provisions of this Section 7.1(d) shall similarly apply to successive mergers or consolidations permitted hereunder.

(e) Notwithstanding Section 7.1(c),

- (i) the General Partner may transfer all or any portion of its General Partnership Interest to (A) a wholly-owned Subsidiary of the General Partner or (B) the owner of all of the ownership interests of the General Partner, and following a transfer of all of its General Partnership Interest, may withdraw as General Partner; and
- (ii) the General Partner may engage in any transaction that is not required to be submitted to the vote of the holders of the REIT Shares by (A) law or (B) the rules of any national securities exchange on which the REIT Shares are traded.

- **7.2** Admission of a Substitute or Additional General Partner. A Person shall be admitted as a substitute or additional General Partner of the Partnership only if the following terms and conditions are satisfied:
 - (a) the Person to be admitted as a substitute or additional General Partner shall have accepted and agreed to be bound by all the terms and provisions of this Agreement (as then in effect), by executing a counterpart thereof and such other documents or instruments as may be required or appropriate in order to effect the admission of such Person as a General Partner, and a certificate evidencing the admission of such Person as a General Partner shall have been filed for recordation and all other actions required by Section 2.5 hereof in connection with such admission shall have been performed;
 - (b) if the Person to be admitted as a substitute or additional General Partner is a corporation or a partnership it shall have provided the Partnership with evidence satisfactory to counsel for the Partnership of such Person's authority to become a General Partner and to be bound by the terms and provisions of this Agreement; and
 - (c) counsel for the Partnership shall have rendered an opinion (relying on such opinions from other counsel and the state or any other jurisdiction as may be necessary) that (x) the admission of the person to be admitted as a substitute or additional General Partner is in conformity with the Act and (y) none of the actions taken in connection with the admission of such Person

as a substitute or additional General Partner will cause (i) the Partnership to be classified other than as a partnership for federal tax purposes, or (ii) the loss of any Limited Partner's limited liability.

7.3 Effect of Bankruptcy, Withdrawal, Death or Dissolution of a General Partner.

- (a) Upon the occurrence of an Event of Bankruptcy as to the sole remaining General Partner (and its removal pursuant to Section 7.4(a) hereof) or the death, withdrawal, deemed removal or dissolution of the sole remaining General Partner (except that, if the sole remaining General Partner is on the date of such occurrence a partnership, the withdrawal, death, dissolution, Event of Bankruptcy as to, or removal of a partner in, such partnership shall be deemed not to be a dissolution of such General Partner if the business of such General Partner is continued by the remaining partner or partners), the Partnership shall be dissolved and terminated unless the Partnership is continued pursuant to Section 7.3(b) hereof. The merger of the General Partner with or into any entity that is admitted as a substitute or successor General Partner pursuant to Section 7.2 hereof shall not be deemed to be the withdrawal, dissolution or removal of the General Partner.
- (b) Following the occurrence of an Event of Bankruptcy as to the sole remaining General Partner (and its removal pursuant to Section 7.4(a) hereof) or the death, withdrawal, removal or dissolution of the sole remaining General Partner (except that, if the sole remaining General Partner is, on the date of such occurrence, a partnership, the withdrawal of, death, dissolution, Event of Bankruptcy as to, or removal of a partner in, such partnership shall be deemed not to be a dissolution of such General Partner if the business of such General Partner is continued by the remaining partner or partners), the Limited Partners, within ninety (90) days after such occurrence, may elect to continue the business of the Partnership for the balance of the term specified in Section 2.4 hereof by selecting, subject to Section 7.2 hereof and any other provisions of this Agreement, a substitute General Partner by consent of a majority in interest of the Limited Partners. If the Limited Partners elect to continue the business of the Partnership and admit a substitute General Partner, the relationship with the Partners and of any Person who has acquired an interest of a Partner in the Partnership shall be governed by this Agreement.

39

7.4 Removal of a General Partner.

- (a) Upon the occurrence of an Event of Bankruptcy as to, or the dissolution of, a General Partner, such General Partner shall be deemed to be removed automatically; provided, however, that if a General Partner is on the date of such occurrence a partnership, the withdrawal, death or dissolution of, Event of Bankruptcy as to, or removal of, a partner in, such partnership shall be deemed not to be a dissolution of such General Partner if the business of such General Partner is continued by the remaining partner or partners. The Limited Partners may not remove the General Partner, with or without cause.
- (b) If a General Partner has been removed pursuant to this Section 7.4 and the Partnership is continued pursuant to Section 7.3 hereof, such General Partner shall promptly transfer and assign its General Partnership Interest in the Partnership to the substitute General Partner approved by a majority in interest of the Limited Partners in accordance with Section 7.3(b) hereof and otherwise admitted to the Partnership in accordance with Section 7.2 hereof. At the time of assignment, the removed General Partner shall be entitled to receive from the substitute General Partner the fair market value of the General Partnership Interest of such removed General Partner as reduced by any damages caused to the Partnership by such General Partner. Such fair market value shall be determined by an appraiser mutually agreed upon by such General Partner and a majority in interest of the Limited Partners within ten (10) days following the removal of the General Partner. In the event that the parties are unable to agree upon an appraiser, the removed General Partner and a majority in interest of the Limited Partners each shall select an appraiser. Each such appraiser shall complete an appraisal of the fair market value of the removed General Partner's General Partnership Interest within thirty (30) days of the General Partner's removal, and the fair market value of the removed General Partner's General Partnership Interest shall be the average of the two appraisals; provided, however, that if the higher appraisal exceeds the lower appraisal by more than 10% of the amount of the lower appraisal, the two appraisers, no later than forty (40) days after the removal of the General Partner, shall select a third appraiser who shall complete an appraisal of the fair market value of the removed General Partner's General Partnership Interest no later than sixty (60) days after the removal of the General Partner. In such case, the fair market value of the removed General Partner's General Partnership Interest shall be the average of the two appraisals closest in value.
- (c) The General Partnership Interest of a removed General Partner, until transfer under Section 7.4(b), shall be converted to that of a special Limited Partner; provided, however, such removed General Partner shall not have any rights to

participate in the management and affairs of the Partnership, and shall not be entitled to any portion of the income, expense, profit, gain or loss allocations or cash distributions allocable or payable, as the case may be, to the Limited Partners. Instead, such removed General Partner shall receive and be entitled only to retain distributions or allocations of such items that it would have been entitled to receive in its capacity as General Partner, until the transfer is effective pursuant to Section 7.4(b).

(d) All Partners shall have given and hereby do give such consents, shall take such actions and shall execute such documents as shall be legally necessary, desirable and sufficient to effect all the foregoing provisions of this Section.

40

ARTICLE 8

RIGHTS AND OBLIGATIONS OF THE LIMITED PARTNERS

- **8.1** Management of the Partnership. The Limited Partners shall not participate in the management or control of Partnership business nor shall they transact any business for the Partnership, nor shall they have the power to sign for or bind the Partnership, such powers being vested solely and exclusively in the General Partner.
- **8.2** <u>Power of Attorney</u>. Each Limited Partner hereby irrevocably appoints the General Partner its true and lawful attorney-infact, who may act for each Limited Partner and in its name, place and stead, and for its use and benefit, to sign, acknowledge, swear to, deliver, file or record, at the appropriate public offices, any and all documents, certificates, and instruments as may be deemed necessary or desirable by the General Partner to carry out fully the provisions of this Agreement and the Act in accordance with their terms, which power of attorney is coupled with an interest and shall survive the death, dissolution or legal incapacity of the Limited Partner, or the transfer by the Limited Partner of any part or all of its Partnership Interest.
- **8.3** <u>Limitation on Liability of Limited Partners</u>. No Limited Partner shall be liable for any debts, liabilities, contracts or obligations of the Partnership. A Limited Partner shall be liable to the Partnership only to make payments of its Capital Contribution, if any, as and when due hereunder. After its Capital Contribution is fully paid, no Limited Partner shall, except as otherwise required by the Act, be required to make any further Capital Contributions or other payments or lend any funds to the Partnership.

8.4 OP Unit Redemption Right.

(a) Subject to Sections 8.4(b), 8.4(c), 8.4(d), 8.4(e), 8.4(f) and 8.5 hereof, the provisions of any agreements between the Partnership and one or more Limited Partners with respect to OP Units held by them, each Limited Partner, other than the General Partner, shall have the right (subject to the terms and conditions set forth herein), but shall not be obligated, to require the Partnership to redeem (a "Redemption") all or a portion of the OP Units, held by such Limited Partner (such OP Units, the "Tendered Units"), in exchange (an "OP Unit Redemption Right") alternatively, for either REIT Shares or the Cash Amount, as determined by the General Partner in its sole discretion. The consideration payable in respect of Tendered Units shall be issued or paid, as the case may be, on the Specified Redemption Date. Any OP Unit Redemption Right shall be exercised pursuant to a Notice of Redemption delivered to the Partnership (with a copy to the General Partner) by the Limited Partner exercising the OP Unit Redemption Right (the "Tendering Party"). A Limited Partner may not exercise the OP Unit Redemption Right for fewer than 1,000 OP Units or, if such Limited Partner holds fewer than 1,000 OP Units, all of the OP Units held by such Partner. The Tendering Party shall have no right, with respect to any OP Units so redeemed, to receive any distribution paid with respect to OP Units if the record date for such distribution is on or after the Specified Redemption Date.

41

(b) If the General Partner elects to cause the Tendered Units to be exchanged for REIT Shares rather than the Cash Amount, then the Partnership shall direct the General Partner to issue and deliver such REIT Shares to the Tendering Party pursuant to the terms set forth in this Section 8.4(b), in which case, (i) the General Partner, acting as a distinct legal entity, shall assume directly the Partnership's redemption obligation with respect thereto and shall satisfy the Tendering Party's exercise

of its OP Unit Redemption Right, and (ii) such transaction shall be treated, for federal income tax purposes, as a transfer by the Tendering Party of such Tendered Units to the General Partner in exchange for REIT Shares. The percentage of the Tendered Units which are to be so exchanged for REIT Shares (rather than the Cash Amount) is referred to as the "Applicable Percentage." In making such election to exchange Tendered Units for the Cash Amount or REIT Shares, the General Partner shall act in a fair, equitable and reasonable manner that neither prefers one group or class of Limited Partners over another nor discriminates against a group or class of Limited Partners. If the General Partner determines to redeem any Tendered Units for REIT Shares, rather than the Cash Amount, on the Specified Redemption Date, the Tendering Party shall sell such number of the Tendered Units to the General Partner in exchange for a number of REIT Shares equal to the product of the REIT Shares Amount and the Applicable Percentage. Such amount of REIT Shares shall be delivered by the General Partner as duly authorized, validly issued, fully paid and nonassessable REIT Shares, free of any pledge, lien, encumbrance or restriction, other than the Ownership Limit (as calculated in accordance with the Charter) and other restrictions provided in the Articles of Incorporation, the bylaws of the General Partner, the Securities Act and relevant state securities or "blue sky" laws. Notwithstanding the provisions of Section 8.4(a) and this Section 8.4(b), the Tendering Parties shall have no rights under this Agreement that would otherwise be prohibited under the Charter.

- (c) In connection with an exercise of OP Unit Redemption Rights pursuant to this <u>Section 8.4</u>, the Tendering Party shall submit the following to the General Partner, in addition to the Notice of Redemption:
 - (i) A written affidavit, dated the same date as the Notice of Redemption, (a) disclosing the actual and constructive ownership, as determined for purposes of Code Sections 856(a)(6) and 856(h), of REIT Shares by (i) such Tendering Party and (ii) any Related Party and (b) representing that, after giving effect to the Redemption, neither the Tendering Party nor any Related Party will own REIT Shares in excess of the Ownership Limit (or, if applicable the Excepted Holder Limit);
 - (ii) A written representation that neither the Tendering Party nor any Related Party has any intention to acquire any additional REIT Shares prior to the closing of the Redemption on the Specified Redemption Date; and
 - (iii) An undertaking to certify, at and as a condition to the closing of the Redemption on the Specified Redemption Date, that either (a) the actual and constructive ownership of REIT Shares by the Tendering Party and any Related Party remain unchanged from that disclosed in the affidavit required by Section 8.4(c)(i) or (b) after giving effect to the Redemption, neither the Tendering Party nor any Related Party shall own REIT Shares in violation of the Ownership Limit (or, if applicable, the Excepted Holder Limit).

- (iv) Any other documents as the General Partner may reasonably require in connection with the issuance of REIT Shares upon the exercise of the OP Unit Redemption Right.
- (d) Any Cash Amount to be paid to a Tendering Party pursuant to this Section 8.4 shall be paid on the Specified Redemption Date; provided, however, that the General Partner may elect to cause the Specified Redemption Date to be delayed for up to an additional 180 days to the extent required for the General Partner to provide financing to be used to make such payment of the Cash Amount, by causing the issuance of additional REIT Shares or otherwise. Notwithstanding the foregoing, the General Partner agrees to use its commercially reasonable efforts to cause the closing of the acquisition of Tendered Units hereunder to occur as quickly as reasonably possible.
- (e) Notwithstanding any other provision of this Agreement, the General Partner shall place appropriate restrictions on the ability of the Limited Partners to exercise their OP Unit Redemption Rights to prevent, among other things, (i) any person from owning shares in excess of the Ownership Limit and the Excepted Holder Limit, (ii) the General Partner's common stock from being owned by fewer than 100 persons, (iii) the General Partner from being "closely held" within the meaning of Section 856(h) of the Code, (iv) violations or what would be likely to constitute a violation of any applicable federal or state securities law, (v) violations of any provision of the General Partner's Charter or Bylaws and (vi) as and if deemed necessary to ensure that the Partnership does not constitute a "publicly traded partnership" under Section 7704 of the Code. If and when the General Partner determines that imposing such restrictions is necessary, the General Partner shall give prompt written notice thereof (a "Restriction Notice") to each of the Limited Partners holding Partnership Units, which notice shall be accompanied by a copy

of an opinion of counsel to the Partnership which states that, in the opinion of such counsel, restrictions are necessary in order to avoid having the Partnership be treated as a "publicly traded partnership" under Section 7704 of the Code.

(f) A redemption fee may be charged in connection with an exercise of OP Unit Redemption Rights pursuant to this Section 8.4.

8.5 Redemption, Exchange or Retention of the Special OP Units in a Terminal Event.

(a) <u>Terminal Events</u>. In connection with: (i) a merger, consolidation or sale of substantially all of the Partnership's assets, a purchase, tender or exchange offer accepted by the holders of more than 50% of the outstanding REIT Shares, or any similar transaction, (ii) any transaction pursuant to which a majority of the Directors then in office are replaced or removed which is not otherwise described in (i) above, other than the removal of a majority of the Directors for cause in accordance with Section 5.8 of the Charter, (iii) a refinancing of the Partnership's assets, or (iv) the termination or nonrenewal of the Advisory Agreement for any reason other than by the Advisor or a termination pursuant to Section 14.02(i) of the Advisory Agreement, (the events described in (i) through (iv) are hereinafter referred to individually as a "**Terminal Event**" and collectively as the "**Terminal Events**"), then at the election of the Special OP Unitholder, and as further provided in <u>Section 8.5(b)</u> below, the Special OP Unitholder may (1) exchange its Special OP Units for OP Units, (2) exchange its Special OP Units for OP Units and immediately thereafter redeem such OP Units received in such exchange pursuant to <u>Section 8.5(b)</u> hereof, or (3) retain its Special OP Units.

43

(b) Redemption, Exchange or Retention of the Special OP Units.

- (i) If the Special OP Unitholder elects to exchange its Special OP Units for OP Units in connection with a Terminal Event, then the Special OP Units shall be exchanged for a number of OP Units equal in value to (the "Special OP Unit Value") the aggregate amount of distributions that would have been made with respect to the Special OP Units under Section 5.2(b)(ii) if all assets (subject to their liabilities) of the Partnership were sold for their fair market value, as determined in good faith by the General Partner, immediately prior to the Terminal Event and in the manner set forth below, any remaining liabilities of the Partnership were satisfied in full in cash according to their terms and assuming for these purposes that all such remaining liabilities had matured, and Net Sales Proceeds (after satisfaction of such liabilities) were distributed in full pursuant to Section 5.2(b). The Special OP Unit Value shall be determined (1) in connection with a Terminal Event described in Section 8.5(a)(i) above, by reference to the value of the consideration received or to be received by the holders of REIT Shares and the implied value of the assets and liabilities of the General Partner and the Partnership as a result thereof, (2) in connection with a Terminal Event described in Sections 8.5(a)(iii) above, by reference to the fair market value of the assets and liabilities of the General Partner and the Partnership as determined in connection with such refinancing, and (3) in connection with a Terminal Event described in Sections 8.5(a)(ii) or (iv) above, by an independent third party selected by the General Partner. The valuation mechanisms referred to in the immediately preceding clauses (1) through (3) are hereinafter referred to as the "Valuation Mechanisms". If multiple Terminal Events are triggered in connection with a series of related events, the General Partner, in its sole discretion, may determine which Valuation Mechanism should be used to value the OP Units and determine the Special OP Unit Value. For the avoidance of doubt, the Special OP Unit Value shall be based on the incremental value of the assets of the Partnership in excess of the highest value of the assets of the Partnership utilized in any preceding Crystallization Events or Terminal Event.
- (ii) If the Special OP Unitholder elects to receive OP Units in exchange for its Special OP Units in connection with the Terminal Event but not to have such OP Units redeemed, then such OP Units shall thereafter be subject to all of the applicable provisions of this Agreement, including providing their holder with the right to redeem such OP Units pursuant to all the terms and conditions of Section 8.4 hereof and including that such OP Unit may be subject to claw back in accordance with Section 5.2(f). If the Special OP Unitholder elects to immediately redeem its OP Units, then the Special OP Unitholder shall receive, at its option, upon redemption of such OP Units, cash, a non-interest bearing promissory note, or REIT Shares (subject to applicable restrictions), with an aggregate value equal to the Special OP Unit Value as determined under subsection (i) of this Section 8.5(b). If the Special OP Unitholder elects payment in the form of a non-interest bearing promissory note, such non-interest bearing promissory note shall be payable in 12 equal quarterly installments, provided, however, that no payment will be made in any quarter in which such payment

- (iii) In connection with a Terminal Event, the Special OP Unitholder may elect not to have the Special OP Units exchanged for OP Units. In such event, the Special OP Unitholder shall receive a cash distribution or a non-interest bearing promissory note equal to the aggregate amount that the Special OP Unitholder would have been entitled to receive under subparagraph (ii) of this Section 8.5(b) above as if the Special OP Unitholder had elected to exchange the Special OP Units into OP Units which were thereafter immediately redeemed (the "Special OP Unit Terminal Distribution"); provided, however that the Special OP Units shall not, under such circumstances, be exchanged for Partnership Units and shall instead remain outstanding and subject to all of the applicable provisions of this Agreement.
- (iv) If REIT Shares are to be issued in connection with the redemption of Partnership Units pursuant to this Section 8.5(b), then the General Partner shall issue such REIT Shares in accordance herewith and the exchange of Partnership Units shall be treated in accordance with Section 8.4(b) as if the Partnership Units were Tendered Units. All cash payments required to be made pursuant to this Section 8.5 shall be made by wire transfer of immediately available funds to an account designated by the recipient of such payment.
- (c) <u>Limitation on Exchange and Redemption</u>. Notwithstanding anything herein to the contrary, no exchange or redemption pursuant to <u>Section 8.5(b)</u> shall be permitted unless and until Unitholders have received (or are deemed to have received pursuant to the deemed valuations set forth in such sections) aggregate, cumulative distributions from the Partnership to Unitholders for all years from operating income, sales proceeds and other sources in an amount equal to (i) the sum of the aggregate Capital Contributions to the Partnership by the Unitholders for all years plus (ii) a Priority Return on the amount described in the immediately preceding subclause (i).
- 8.6 Redemption and Exchange of Partnership Units by the Special OP Unitholder. The Special OP Unitholder shall have the right to redeem the Partnership Units it holds pursuant to all the terms and conditions of Section 8.4 hereof. Notwithstanding the foregoing, the Special OP Unitholder shall not be entitled to exercise such right to redeem the portion of the OP Units it holds that are attributable to the rollover of its indirect interest in the Property in connection with the formation of the Partnership (the "Restricted OP Units") prior to August 1, 2022; provided that as of that date 50% of the Restricted OP Units shall no longer be treated as Restricted OP Units and shall be exchangeable; provided further, that at such time as OP Unitholders have received (or are deemed to have received in a hypothetical liquidation of the Partnership) aggregate, cumulative distributions from the Partnership to Unitholders for all years from operating income, sales proceeds and other sources in an amount equal to (i) the sum of the aggregate Capital Contributions to the Partnership by the OP Unitholders for all years plus (ii) a Priority Return on the amount described in the immediately preceding subclause (i), an additional 40% of the original Restricted OP Units shall not longer be treated as Restricted OP Units. For so long as the Special OP Unitholder remains a partner of the Partnership, it shall retain at least 10% of the original Restricted OP Units.

45

ARTICLE 9

TRANSFERS OF LIMITED PARTNERSHIP INTERESTS

9.1 Purchase for Investment.

(a) Each Limited Partner hereby represents and warrants to the General Partner and to the Partnership that the acquisition of his Partnership Interest is made as a principal for his account for investment purposes only and not with a view to the resale or distribution of such Partnership Interest.

(b) Each Limited Partner agrees that he will not sell, assign or otherwise transfer his Partnership Interest or any fraction thereof, whether voluntarily or by operation of law or at judicial sale or otherwise, to any Person who does not make the representations and warranties to the General Partner set forth in <u>Section 9.1(a)</u> above and similarly agree not to sell, assign or transfer such Partnership Interest or fraction thereof to any Person who does not similarly represent, warrant and agree.

9.2 <u>Restrictions on Transfer of Limited Partnership Interests</u>.

- (a) Subject to the provisions of Section 9.2(b) and 9.2(c), no Limited Partner may offer, sell, assign, hypothecate, pledge or otherwise transfer all or any portion of his Limited Partnership Interest, or any of such Limited Partner's economic rights as a Limited Partner, whether voluntarily or by operation of law or at judicial sale or otherwise (collectively, a "Transfer") without the prior consent of the General Partner, which consent may be granted or withheld in its sole and absolute discretion. Any such purported transfer undertaken without such consent shall be considered to be null and void ab initio and shall not be given effect. The General Partner may require, as a condition of any Transfer to which it consents, that the transferor assume all costs incurred by the Partnership in connection therewith.
- (b) No Limited Partner may withdraw from the Partnership other than: (i) as a result of a permitted Transfer (i.e., a Transfer consented to as contemplated by clause (a) above or clause (c) below or a Transfer pursuant to Section 9.5 below) of all of its Partnership Interest pursuant to this Article 9, (ii) pursuant to a redemption of all of its Partnership Units pursuant to Section 8.4 or (iii) pursuant to the redemption of the Limited Partner's Special OP Units pursuant to Section 8.5 and Partnership Units pursuant to Section 8.4. Upon the permitted Transfer or redemption of all of a Limited Partner's Partnership Units and Special OP Units, if any, such Limited Partner shall cease to be a Limited Partner.
- (c) Notwithstanding Section 9.2(a) and subject to Sections 9.2(d), 9.2(e) and 9.2(f) below, a Limited Partner may not Transfer, without the prior consent of the General Partner, which consent will not be unreasonably withheld, all or a portion of its Partnership Interest to (i) a parent or parent's spouse, natural or adopted descendant or descendants, spouse of such descendant, or brother or sister, or a trust created by such Limited Partner for the benefit of such Limited Partner and/or any such person(s), of which trust such Limited Partner or any such person(s) is a trustee, (ii) a corporation controlled by a Person or Persons named in (i) above, or (iii) if the Limited Partner is an entity, its beneficial owners.

- (d) No Limited Partner may effect a Transfer of its Limited Partnership Interest, in whole or in part, if, in the opinion of legal counsel for the Partnership, such proposed Transfer would require the registration of the Limited Partnership Interest under the Securities Act or would otherwise violate any applicable federal or state securities or blue sky law (including investment suitability standards).
- (e) No Transfer by a Limited Partner of its Partnership Interest, in whole or in part, may be made to any Person if (i) in the opinion of legal counsel for the Partnership, the transfer would result in the Partnership's being treated as an association taxable as a corporation (other than a qualified REIT subsidiary within the meaning of Section 856(i) of the Code), (ii) in the opinion of legal counsel for the Partnership, it would adversely affect the ability of the General Partner to continue to qualify as a REIT or subject the General Partner to any additional taxes under Section 857 or Section 4981 of the Code, or (iii) such transfer is effectuated through an "established securities market" or a "secondary market (or the substantial equivalent thereof)" within the meaning of Section 7704 of the Code.
- (f) No Transfer by a Limited Partner of any Partnership Interest may be made to a lender to the Partnership or any Person who is related (within the meaning of Regulations Section 1.752-4(b)) to any lender to the Partnership whose loan constitutes a nonrecourse liability (within the meaning of Regulations Section 1.752-1(a)(2)), without the consent of the General Partner, which may be withheld in its sole and absolute discretion, provided that as a condition to such consent the lender will be required to enter into an arrangement with the Partnership and the General Partner to exchange or redeem for the Cash Amount any Partnership Units in which a security interest is held simultaneously with the time at which such lender would be deemed to be a Partner in the Partnership for purposes of allocating liabilities to such lender under Section 752 of the Code.
- (g) Any Transfer in contravention of any of the provisions of this <u>Article 9</u> shall be void and ineffectual and shall not be binding upon, or recognized by, the Partnership.

(h) Prior to the consummation of any Transfer under this <u>Article 9</u>, the transferor and/or the transferee shall deliver to the General Partner such opinions, certificates and other documents as the General Partner shall request in connection with such Transfer.

9.3 Admission of Substitute Limited Partner.

- (a) Subject to the other provisions of this <u>Article 9</u>, an assignee of the Limited Partnership Interest of a Limited Partner (which shall be understood to include any purchaser, transferee, donee, or other recipient of any disposition of such Limited Partnership Interest) shall be deemed admitted as a Limited Partner of the Partnership only with the consent of the General Partner, which consent may be granted or withheld in its sole and absolute discretion, and upon the satisfactory completion of the following:
 - (i) The assignee shall have accepted and agreed to be bound by the terms and provisions of this Agreement by executing a counterpart or an amendment thereof, including a revised Exhibit A, and such other documents or instruments as the General Partner may require in order to effect the admission of such Person as a Limited Partner.

- (ii) To the extent required, an amended Certificate evidencing the admission of such Person as a Limited Partner shall have been signed, acknowledged and filed for record in accordance with the Act.
- (iii) The assignee shall have delivered a letter containing the representation set forth in <u>Section 9.1(a)</u> hereof and the agreement set forth in Section 9.1(b) hereof.
- (iv) If the assignee is a corporation, partnership or trust, the assignee shall have provided the General Partner with evidence satisfactory to counsel for the Partnership of the assignee's authority to become a Limited Partner under the terms and provisions of this Agreement.
- (v) The assignee shall have executed a power of attorney containing the terms and provisions set forth in Section 8.2 hereof.
- (vi) The assignee shall have paid all legal fees and other expenses of the Partnership and the General Partner and filing and publication costs in connection with its substitution as a Limited Partner.
- (vii) The assignee has obtained the prior written consent of the General Partner to its admission as a Substitute Limited Partner, which consent may be given or denied in the exercise of the General Partner's sole and absolute discretion.
- (b) For the purpose of allocating income and losses and distributing cash received by the Partnership, a Substitute Limited Partner shall be treated as having become, and appearing in the records of the Partnership as, a Partner upon the filing of the Certificate described in Section 9.3(a)(ii) hereof or, if no such filing is required, the later of the date specified in the transfer documents or the date on which the General Partner has received all necessary instruments of transfer and substitution.
- (c) The General Partner shall cooperate with the Person seeking to become a Substitute Limited Partner by preparing the documentation required by this Section and making all official filings and publications. The Partnership shall take all such action as promptly as practicable after the satisfaction of the conditions in this <u>Article 9</u> to the admission of such Person as a Limited Partner of the Partnership.

- (a) Subject to the provisions of <u>Sections 9.1</u> and <u>9.2</u> hereof, except as required by operation of law, the Partnership shall not be obligated for any purposes whatsoever to recognize the assignment by any Limited Partner of its Partnership Interest until the Partnership has received notice thereof.
- (b) Any Person who is the assignee of all or any portion of a Limited Partner's Limited Partnership Interest, but does not become a Substitute Limited Partner and desires to make a further assignment of such Limited Partnership Interest, shall be subject to all the provisions of this <u>Article 9</u> to the same extent and in the same manner as any Limited Partner desiring to make an assignment of its Limited Partnership Interest.
- 9.5 Effect of Bankruptcy, Death, Incompetence or Termination of a Limited Partner. The occurrence of an Event of Bankruptcy as to a Limited Partner, the death of a Limited Partner or a final adjudication that a Limited Partner is incompetent (which term shall include, but not be limited to, insanity) shall not cause the termination or dissolution of the Partnership, and the business of the Partnership shall continue if an order for relief in a bankruptcy proceeding is entered against a Limited Partner, the trustee or receiver of his estate or, if he dies, his executor, administrator or trustee, or, if he is finally adjudicated incompetent, his committee, guardian or conservator, shall have the rights of such Limited Partner for the purpose of settling or managing his estate property and such power as the bankrupt, deceased or incompetent Limited Partner possessed to assign all or any part of his Partnership Interest and to join with the assignee in satisfying conditions precedent to the admission of the assignee as a Substitute Limited Partner.
- 9.6 Joint Ownership of Interests. A Partnership Interest may be acquired by two individuals as joint tenants with right of survivorship, provided that such individuals either are married or are related and share the same home as tenants in common. The written consent or vote of both owners of any such jointly held Partnership Interest shall be required to constitute the action of the owners of such Partnership Interest; provided, however, that the written consent of only one joint owner will be required if the Partnership has been provided with evidence satisfactory to the counsel for the Partnership that the actions of a single joint owner can bind both owners under the applicable laws of the state of residence of such joint owners. Upon the death of one owner of a Partnership Interest held in a joint tenancy with a right of survivorship, the Partnership Interest shall become owned solely by the survivor as a Limited Partner and not as an assignee. The Partnership need not recognize the death of one of the owners of a jointly- held Partnership Interest until it shall have received notice of such death. Upon notice to the General Partner from either owner, the General Partner shall cause the Partnership Interest to be divided into two equal Partnership Interests, which shall thereafter be owned separately by each of the former owners.

ARTICLE 10

BOOKS AND RECORDS; ACCOUNTING; TAX MATTERS

10.1 Books and Records. At all times during the continuance of the Partnership, the Partners shall keep or cause to be kept at the Partnership's specified office true and complete books of account in accordance with generally accepted accounting principles, including: (a) a current list of the full name and last known business address of each Partner, (b) a copy of the Certificate of Limited Partnership and all Certificates of amendment thereto, (c) copies of the Partnership's federal, state and local income tax returns and reports, (d) copies of this Agreement and amendments thereto and any financial statements of the Partnership for the three most recent years and (e) all documents and information required under the Act. Any Partner or its duly authorized representative, upon paying the costs of collection, duplication and mailing, shall be entitled to inspect or copy such records during ordinary business hours.

10.2 Custody of Partnership Funds; Bank Accounts.

- (a) All funds of the Partnership not otherwise invested shall be deposited in one or more accounts maintained in such banking or brokerage institutions as the General Partner shall determine, and withdrawals shall be made only on such signature or signatures as the General Partner may, from time to time, determine.
- (b) All deposits and other funds not needed in the operation of the business of the Partnership may be invested by the General Partner in investment grade instruments (or investment companies whose portfolio consists primarily thereof), government obligations, certificates of deposit, bankers' acceptances and municipal notes and bonds. The funds of the Partnership shall not be commingled with the funds of any other Person except for such commingling as may necessarily result from an investment in those investment companies permitted by this Section 10.2(b).

- 10.3 Fiscal and Taxable Year. The fiscal and taxable year of the Partnership shall be the calendar year.
- **10.4** Annual Tax Information and Report. Within seventy-five (75) days after the end of each fiscal year of the Partnership, the General Partner shall furnish to each person who was a Limited Partner at any time during such year the tax information necessary to file such Limited Partner's individual tax returns as shall be reasonably required by law.

10.5 Partnership Representative; Tax Elections; Special Basis Adjustments.

(a) Partnership Representative.

(i) The General Partner shall be designated as the partnership representative of the Partnership (the "Partnership Representative") within the meaning of Section 6223 of the Code in accordance with Regulations Section 301.6223-1 and any other applicable guidance by the Service. The General Partner shall also appoint an individual (the "Designated Individual") through whom the Partnership Representative acts in accordance with Regulations Section 301.6223-1 and any other applicable guidance by the Service. The General Partner is authorized to revoke and replace from time to time the Partnership Representative or the Designated Individual in accordance with Regulations Section 301.6223-1 and any other applicable guidance by the Service. The General Partner shall make all designations and appointments under similar or analogous state, local or non-U.S. laws. The Partnership Representative shall have the right and obligation to take all actions authorized and required, respectively, by the Code and Regulations for the Partnership Representative. The taking of any action and the incurring of any expense by the Partnership Representative in connection with any applicable proceeding, except to the extent required by law, is a matter in the sole and absolute discretion of the Partnership Representative, and the provisions relating to indemnification of the Indemnitees set forth in Section 6.3 hereof shall be fully applicable to the Partnership Representative and the Designated Individual, if any, acting as such.

- (ii) Each Partner agrees that such Partner shall not treat any Partnership-related item inconsistently on such Partner's federal, state, local or non-U.S. tax return with the treatment of the item on the Partnership's return. Any deficiency for taxes imposed on any Partner with respect to such Partner's interest in the Partnership (including penalties, additions to tax or interest imposed with respect to such taxes and any tax deficiency imposed pursuant to Section 6226 of the Code) will be paid by such Partner. Without duplication of Section 5.2(c) hereof, if the Partnership is required to pay (and actually pays) an imputed underpayment (including penalties, additions to tax or interest imposed with respect to such taxes, pursuant to Section 6225 of the Code) with respect to a reviewed year, or bears the economic burden of imputed underpayments made by entities in which it is a partner, such amounts paid will be recoverable from the reviewed-year Partners. To the extent that the Partnership or the Partnership Representative, as applicable, does not make an election under Sections 6221(b) (if available) or 6226 of the Code, the Partnership shall use commercially reasonable efforts to (i) make any modifications available under Section 6225(c) of the Code, and (ii) if requested by a Partner, provide to such Partner information allowing such Partner to file an amended federal income tax return, as described in Section 6225(c)(2) of the Code, to the extent such amended return and payment of any related federal income taxes would reduce any taxes payable by Partnership. Each Limited Partner shall, including any time after such Limited Partner withdraws from or otherwise ceases to be a Limited Partner, take all actions requested by the General Partner, including timely provision of requested information and consents in connection with implementing any elections or decisions made by the Partnership or the Partnership Representative (or Person acting in a similar capacity under similar or analogous state, local or non-U.S. laws) related to any tax audit or examination of the Partnership (including to implement any modifications to any imputed underpayment or similar amount under Section 6225(c) of the Code, any elections under Sections 6221 or 6226 of the Code and any administrative adjustment request under Section 6227 of the Code).
- (iii) Notwithstanding anything to the contrary in this Agreement, any information, representations, certificates, forms, or documentation provided pursuant to this Section 10.5(a) may be disclosed to any applicable taxing authority. Each Partner agrees to be bound by the provisions of this Section 10.5(a) at all times, including any time after such Partner ceases to be a Partner solely with respect to matters directly related to such Partner's interest in the Partnership, and the provisions of Section 10.5(a) shall survive the winding up, liquidation and dissolution of the Partnership.

- (b) All elections required or permitted to be made by the Partnership under the Code or any applicable state or local tax law shall be made by the General Partner in its sole and absolute discretion.
- (c) In the event of a transfer of all or any part of the Partnership Interest of any Partner, the Partnership, at the option of the General Partner may elect pursuant to Section 754 of the Code to adjust the basis of the Partnership's assets. Notwithstanding anything contained in <u>Article 5</u> of this Agreement, any adjustments made pursuant to Section 754 of the Code shall affect only the successor in interest to the transferring Partner and in no event shall be taken into account in establishing, maintaining or computing Capital Accounts for the other Partners for any purpose under this Agreement. Each Partner will furnish the Partnership with all information necessary to give effect to such election.

10.6 Reports to Limited Partners.

- (a) As soon as practicable after the close of each fiscal quarter (other than the last quarter of the fiscal year), the General Partner shall cause to be mailed to each Limited Partner a quarterly report containing financial statements of the Partnership, or of the General Partner if such statements are prepared solely on a consolidated basis with the General Partner, for such fiscal quarter, presented in accordance with generally accepted accounting principles. As soon as practicable after the close of each fiscal year, the General Partner shall cause to be mailed to each Limited Partner an annual report containing financial statements of the Partnership, or of the General Partner if such statements are prepared solely on a consolidated basis with the General Partner, for such fiscal year, presented in accordance with generally accepted accounting principles. The annual financial statements shall be audited by accountants selected by the General Partner.
- (b) Any Partner shall further have the right to a private audit of the books and records of the Partnership at the expense of such Partner, provided such audit is made for Partnership purposes and is made during normal business hours.
- 10.7 <u>Safe Harbor Election</u>. The Partners agree that, in the event the Safe Harbor Regulation is finalized, the Partnership shall be authorized and directed to make the Safe Harbor Election and the Partnership and each Partner (including any person to whom an interest in the Partnership is transferred in connection with the performance of services) agrees to comply with all requirements of the Safe Harbor with respect to all interests in the Partnership transferred in connection with the performance of services while the Safe Harbor Election remains effective. The Tax Matters Partner shall be authorized to (and shall) prepare, execute, and file the Safe Harbor Election.

52

ARTICLE 11

AMENDMENT OF AGREEMENT

The General Partner's consent shall be required for any amendment to this Agreement. The General Partner, without the consent of the Limited Partners, may amend this Agreement in any respect or merge or consolidate the Partnership with or into any other partnership or business entity (as defined in Section 17-211 of the Act) in a transaction pursuant to Section 7.1(c)(ii) or 7.1(c)(iii), 7.1(d) or 7.1(e) hereof; provided, however, that the following amendments and any other merger or consolidation of the Partnership shall require the consent of Limited Partners holding more than 67% of the Percentage Interests of the Limited Partners:

- (a) any amendment affecting the operation of the Conversion Factor or the OP Unit Redemption Right (except as provided in Section 7.4(d) or 7.1(d) hereof) in a manner adverse to the Limited Partners;
- (b) any amendment that would adversely affect the rights of the Limited Partners to receive the distributions payable to them hereunder, other than with respect to the issuance of additional Partnership Units pursuant to Section 4.2 hereof; or

(c) any amendment that would alter the Partnership's allocations of income and loss to the Limited Partners, other than with respect to the issuance of additional Partnership Units pursuant to Section 4.2 hereof; and any amendment that would impose on any Limited Partner any obligation to make additional Capital Contributions to the Partnership or otherwise alter such Limited Partner's right to receive distributions of cash or other property or allocations of items of income, gain, deduction loss or credit shall require the written consent of both the General Partner and any such Limited Partner. In addition, any amendment to Sections 5.1(b), 5.2, or 8.5 shall require the consent of the Special OP Unitholder, and any amendment to this Article 11 shall require the written consent of all Partners.

53

ARTICLE 12

GENERAL PROVISIONS

- 12.1 Notices. All communications required or permitted under this Agreement shall be in writing and shall be deemed to have been given when delivered personally or upon deposit in the United States mail, registered, postage prepaid return receipt requested, to the Partners at the addresses set forth in Exhibit A attached hereto; provided, however, that any Partner may specify a different address by notifying the General Partner in writing of such different address. Notices to the Partnership shall be delivered at or mailed to its specified office.
- **12.2** <u>Survival of Rights</u>. Subject to the provisions hereof limiting transfers, this Agreement shall be binding upon and inure to the benefit of the Partners and the Partnership and their respective legal representatives, successors, transferees and assigns.
- **12.3** <u>Additional Documents</u>. Each Partner agrees to perform all further acts and execute, swear to, acknowledge and deliver all further documents which may be reasonable, necessary, appropriate or desirable to carry out the provisions of this Agreement or the Act.
- 12.4 <u>Severability</u>. If any provision of this Agreement shall be declared illegal, invalid, or unenforceable in any jurisdiction, then such provision shall be deemed to be severable from this Agreement (to the extent permitted by law) and in any event such illegality, invalidity or unenforceability shall not affect the remainder hereof.
- 12.5 <u>Entire Agreement</u>. This Agreement and exhibits attached hereto constitute the entire Agreement of the Partners and supersede all prior written agreements and prior and contemporaneous oral agreements, understandings and negotiations with respect to the subject matter hereof.
- 12.6 <u>Pronouns and Plurals</u>. When the context in which words are used in the Agreement indicates that such is the intent, words in the singular number shall include the plural and the masculine gender shall include the neuter or female gender as the context may require.
- **12.7** <u>Headings</u>. The Article headings or sections in this Agreement are for convenience only and shall not be used in construing the scope of this Agreement or any particular Article.
- 12.8 <u>Counterparts</u>. This Agreement may be executed in several counterparts, each of which shall be deemed to be an original copy and all of which together shall constitute one and the same instrument binding on all parties hereto, notwithstanding that all parties shall not have signed the same counterpart.
- **12.9** <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware; provided, however, that any cause of action for violation of federal or state securities laws shall not be governed by this <u>Section 12.9</u>.

IN WITNESS WHEREOF, the parties hereto have hereunder affixed their signatures to this Agreement, all as of the 30th day of September, 2022.

GENERAL PARTNER:

NRI REAL TOKEN INC., a Maryland corporation

By: /s/ Brent M. Reynolds

Name: Brent M. Reynolds Title: Chief Executive Officer

SPECIAL OP UNITHOLDER:

NRI REAL TOKEN THESIS LLC, a Delaware limited liability company

NRI Real Token Advisors LLC, a Delaware By: limited liability company and Manager of Special OP Unitholder

By: Nolan Reynolds International, LLC, a Delaware limited liability company and the Sole Member of NRI Real Token Advisors, LLC

By: /s/ Brent M. Reynolds

Brent M. Reynolds Its Chief Executive Officer

Annex A - 55

EXECUTION COPY

AMENDED AND RESTATED ADVISORY AGREEMENT

Among

NRI REAL TOKEN ADVISORS LLC,

NRI REAL TOKEN LP,

and

NRI REAL ESTATE INVESTMENT & TECHNOLOGY, INC.

September 30, 2022

TABLE OF CONTENTS

		<u>Page</u>
Article 1 DEFINITIONS		1
Article 2 APPOINTMENT		4
Article 3 DUTIES OF THE ADVISOR		4
3.01	Offering Services	4
3.02	Acquisition Services	5
3.03	Asset Management Services	6
3.04	Accounting and Other Administrative Services:	6
3.05	Stockholder Services	7
3.06	Financing Services	8
	Disposition Services	8
Article 4 AUTHORITY OF ADVISOR		8
4.01	General	8
	Powers of the Advisor	8
4.03	Approval by Directors	9
Article 5 BANK ACCOUNTS		9
Article 6 RECORDS AND FINANCIAL STATEMENTS		9
Article 7 LIMITATION ON ACTIVITIES		9
Article 8 RELATIONSHIP WITH DIRECTORS AND OFFICERS		10
Article 9 FEES		10
9.01	Acquisition Fees	10
9.02	Asset Management, Accounting and Other Administrative Services Fees	10
9.03	Debt Financing Fees	10
	Disposition Fees	10
Article 10 EXPENSES		11
	General	11
10.02	Reimbursement to Advisor	12

Article 11 OTHER SERVICES		
Article 12 RELATIONSHIP OF ADVISO	OR AND COMPANY; OTHER ACTIVITIES OF THE ADVISOR	12
12.01 Relationship		12
12.02 Time Commitment		12
12.03 Investment Opportunities	s and Allocation	13
Article 13 THE NRI NAME		13
Article 14 TERM AND TERMINATION OF THE AGREEMENT		13
14.01 Term		13
14.02 Termination by the Comp	pany	13
14.03 Termination by the Advi		14
14.04 Payments on Termination	n and Survival of Certain Rights and Obligations	14
14.05 Repurchase of Units		14
Article 15 ASSIGNMENT		14
Article 16 INDEMNIFICATION AND L	IMITATION OF LIABILITY	15
16.01 Indemnification by the C		15
16.02 Indemnification by the A	dvisor	15
16.03 Advisor's Liability		16
Article 17 MISCELLANEOUS		16
17.01 Notices		16
17.02 Modification		17
17.02 Severability		17
17.04 Construction		17
17.05 Entire Agreement		17
17.06 Waiver		17
17.07 Gender		17
17.08 Titles Not to Affect Inter	pretation	17
17.09 Counterparts		17

ii

AMENDED AND RESTATED ADVISORY AGREEMENT

This Amended and Restated Advisory Agreement, dated as of September 30, 2022, is among NRI Real Token Advisors LLC, a Delaware limited liability company, NRI Real Token LP, a Delaware limited partnership, and NRI Real Estate Investment & Technology, Inc. (f/k/a NRI Real Token Inc.), a Maryland corporation (the "Agreement").

WITNESSETH

WHEREAS, the parties hereto and NRI Real Token Thesis LLC entered into the Advisory Agreement on October 29, 2021 (the "Original Agreement");

WHEREAS, the General Partner and NRI Real Token Thesis LLC have agreed to convert NRI Real Token Thesis LLC's general partnership interest in the Company into a limited partnership interest as reflected in the Limited Partnership Agreement;

WHEREAS, due to NRI Real Token Thesis LLC ceasing to be a general partner of the Company, NRI Real Token Thesis LLC shall be removed as a party to this Agreement effective as of the date hereof;

WHEREAS, Section 17.02 of the Original Agreement requires the written agreement of all parties to the Original Agreement in order to change or modify the Original Agreement, and NRI Real Token Thesis LLC's signature hereto shall serve solely as its acknowledgement that it has agreed to the changes to the Original Agreement reflected by this Agreement;

WHEREAS, the Company (as hereinafter defined) desires to continue to avail itself of the knowledge, experience, sources of information, advice, assistance and certain facilities available to the Advisor (hereinafter defined) and to have the Advisor undertake the duties and responsibilities hereinafter set forth on the terms set forth in this Agreement; and

WHEREAS, the Advisor is willing to undertake to render such services on the terms and conditions hereinafter set forth.

NOW, THEREFORE, in consideration of the foregoing and of the mutual covenants and agreements contained herein, the parties hereto agree as follows:

ARTICLE 1 DEFINITIONS

The following defined terms used in this Amended and Restated Advisory Agreement shall have the meanings specified below:

"Acquisition Expenses" shall have the meaning given such term in Section 10.01(i) of this Agreement.

"Advisor" means (i) NRI Real Token Advisors LLC, a Florida limited liability company, or (ii) any successor advisor to the Company.

"Affiliate" means, with respect to any Person, (i) any Person directly or indirectly owning, controlling or holding, with the power to vote, ten percent or more of the outstanding voting securities of such other Person; (ii) any Person ten percent or more of whose outstanding voting securities are directly or indirectly owned, controlled or held, with the power to vote, by such other Person; (iii) any Person directly or indirectly controlled by or under common control with such other Person; (iv) any executive officer, director, trustee or general partner of such other Person and (v) any legal entity for which such Person acts as an executive officer, director, trustee or general partner. For the purposes of this Agreement, the Advisor shall not be deemed to be an Affiliate of the Company, and vice versa.

"Average Invested Assets" means, for a specified period, the average of the aggregate book value of the assets of the Company invested, directly or indirectly, in Properties, loans and other permitted investments secured by real estate before reserves for depreciation or bad debts or other similar non-cash reserves, computed by taking the average of such values at the end of each month during such period.

"Board of Directors" means the Board Directors of the General Partner.

"Bylaws" means the bylaws of the General Partner, as amended from time to time.

"Charter" means the Articles of Incorporation of the General Partner, as amended, supplemented, corrected and restated from time to time.

"Code" means the Internal Revenue Code of 1986, as amended from time to time, or any successor statute thereto. Reference to any provision of the Code shall mean such provision as in effect from time to time, as the same may be amended, and any successor provision thereto, as interpreted by any applicable regulations as in effect from time to time.

"Company" means NRI Real Token LP, a Delaware limited partnership. Within the context of discussions of the operations, business and administration of the Company, the term "Company" shall mean, collectively, NRI Real Token LP and the General Partner for the purposes of this Agreement.

"Director" means a member of the Board of Directors of the General Partner.

"General Partner" means NRI Real Estate Investment & Technology, Inc. (f/k/a NRI Real Token Inc.), a Maryland corporation, as described in the Limited Partnership Agreement.

"Gross Revenues" means the gross revenues generated by the Company for an applicable period, as reasonably determined by the General Partner.

"Independent Director" means an individual other than (i) an officer or employee of the General Partner, (ii) an officer or employee of the Advisor or any of its Affiliates or (iii) any other individual having a relationship which, in the opinion of the Board of Directors, would interfere with the exercise of independent judgment in carrying out the responsibilities of a Director.

2

"Initial Asset Value" means (i) in the case of a real estate investment other than a loan which the Company originates, the gross purchase price of real estate investments acquired directly by the Company, including any debt attributable to such investments, or the Company's pro rata share of the gross asset value of real estate investments held by entities in which the Company invests, and (ii) in the case of a loan which the Company originates, the total principal amount committed under the loan.

"<u>Limited Partnership Agreement</u>" means the Amended and Restated Limited Partnership Agreement of NRI Real Token LP, dated as of even date herewith, as the same may be amended and restated from time to time.

"Limited Partnership Interests" has the meaning set forth in the Limited Partnership Agreement.

"NRI" means Nolan Reynolds International, LLC and its Affiliates.

"Offering" means the offering of the OP Units in the Company, which may be convertible at the election of the holder into the Shares of the General Partner issued in the form of Security Tokens (or, in certain circumstances at the election of the General Partner, into cash), as described in the Offering Memorandum.

"Offering Memorandum" means that certain Private Offering Memorandum of OP Units in the Company, as may be amended or supplemented from time to time.

"OP Units" has the meaning set forth in the Limited Partnership Agreement.

"Original Agreement" has the meaning set forth in the recitals.

"Person" means an individual, corporation, partnership, estate, trust, a portion of a trust permanently set aside for or to be used exclusively for the purposes described in Section 642(c) of the Code, association, private foundation within the meaning of Section 509(a) of the Code, joint stock company or other entity.

"Properties" has the meaning set forth in the Limited Partnership Agreement.

"Property Management Agreement" means any Property Management Agreement between the Company and a Property Manager.

"Property Manager" means, collectively, NR Operations, LLC, a Florida limited partnership, Alpareno Restaurant Group, LLC, Gables Residential Services, Inc., Legacy Parking Company, LLC, Hersha Hospitality Management L.P., or in each case, an Affiliate thereof, when serving as the property manager for any property owned by the Company pursuant to a Property Management Agreement, together with any future property manager of all or any portion of the Properties.

"REIT" means a "real estate investment trust" under Sections 856 through 860 of the Code.

3

"Securities" means any class or series of units or shares of the Company or the General Partner, including shares of common stock, shares of preferred stock, units, special units or shares and any other evidences of equity or beneficial or other interests, voting trust

certificates, bonds, debentures, notes or other evidences of indebtedness, secured or unsecured, convertible, subordinated or otherwise, or in general any instruments commonly known as "securities" or any certificates of interest, shares or participations in, temporary or interim certificates for, receipts for, guarantees of, or warrants, options or rights to subscribe to, purchase or acquire, any of the foregoing.

"Security Tokens" or "Tokens" means the representation of the Shares in digital form using blockchain technology, as may be modified or replaced from time to time.

- "Shares" means shares of common stock of the General Partner; par value \$.01 per share.
- "Special OP Unitholder" has the meaning set forth in the Limited Partnership Agreement.
- "Special OP Units" has the meaning set forth in the Limited Partnership Agreement.
- "Stockholders" means the registered holders of outstanding Shares.
- "Termination Date" means the date of termination of this Agreement.

"Termination Fee" means a fee payable to the Advisor upon termination of this Agreement, other than in accordance with Section 14.02, in the amount equal to three (3) times the sum of (i) the average annual acquisition fees in accordance with Section 9.01, (ii) the average annual asset management fees in accordance with Section 9.02, (iii) the average annual debt financing fees in accordance with Section 9.03; and (iv) the average annual asset disposition fees in accordance with Section 9.04, in each case earned by the Advisor during the twelve (12) month period immediately preceding the most recently completed calendar quarter prior to the termination of this Agreement.

ARTICLE 2 APPOINTMENT

The Company hereby appoints the Advisor to serve as its advisor on the terms and conditions set forth in this Agreement, and the Advisor hereby accepts such appointment.

ARTICLE 3 DUTIES OF THE ADVISOR

The Advisor is responsible for managing, operating, directing and supervising the operations and administration of the Company and its real estate investments to the fullest extent allowed by law. The Advisor shall, either directly or by engaging an Affiliate or third party, perform the following duties:

- 3.01 Offering Services. The Advisor shall manage and supervise:
- (i) Development of the product offering, including the determination of the specific terms of the Securities to be offered by the General Partner and/or the Company, preparation of all offering and related documents, and obtaining all required regulatory approvals of such documents;

- (ii) Approval of the participating broker-dealers and negotiation of the related selling agreements;
- (iii) Coordination of the due diligence process relating to participating broker-dealers, placement agents and other third-party websites, and their review of the Offering Memorandum, any prospectus and other Offering and Company documents;
- (iv) Preparation and approval of all marketing materials contemplated to be used by the Placement Agents (as defined in the Offering Memorandum) or others in the Offering of the Company's and General Partner's Securities;
- (v) Negotiation and coordination with the transfer agent for the receipt, collection, processing and acceptance of subscription agreements, commissions, and other administrative support functions;

- (vi) Negotiations and coordination with one or more Alternative Trading Systems (each an "ATS" or collectively, "ATSs") and licensed custodians with respect to the possible trading of the Security Tokens, in whole or fractional interests;
- (vii) Creation and implementation of various technology and electronic communications related to the Offering of the Company's and the General Partner's Securities; and
- (viii) All other services related to organization of the Company, the General Partner or the Offering, whether performed and incurred by the Advisor or its Affiliates.

3.02 Acquisition Services.

- (i) Serve as the Company's investment advisor and obtain certain market research and economic and statistical data in connection with the Company's real estate investments and investment objectives and policies;
- (ii) Subject to Section 4 hereof and the investment objectives and policies of the Company: (a) locate, analyze and select potential investments; (b) structure and negotiate the terms and conditions of transactions pursuant to which real estate investments will be made; and (c) acquire real estate investments on behalf of the Company;
 - (iii) Oversee the due diligence process;
- (iv) Prepare reports regarding prospective investments which include recommendations and supporting documentation necessary for the Directors to evaluate the proposed investments;

5

- (v) Obtain reports (which may be prepared by the Advisor or its Affiliates), where appropriate, concerning the value of contemplated investments of the Company; and
 - (vi) Negotiate and execute approved investments and other transactions.

3.03 <u>Asset Management Services.</u>

- (i) Investigate, select, and, on behalf of the Company, engage and conduct business with such Persons as the Advisor deems necessary to the proper performance of its obligations hereunder, including, but not limited to, consultants, accountants, lenders, technical advisors, attorneys, brokers, underwriters, corporate fiduciaries, escrow agents, depositaries, custodians, agents for collection, insurers, insurance agents, developers, construction companies and any and all Persons acting in any other capacity deemed by the Advisor necessary or desirable for the performance of any of the foregoing services;
- (ii) Monitor applicable markets and obtain reports (which may be prepared by the Advisor or its Affiliates) where appropriate, concerning the value of investments of the Company;
 - (iii) Monitor and evaluate the performance of investments of the Company;
- (iv) Provide daily management services to the Company and perform and supervise the various management and operational functions related to the Company's investments;
 - (v) Coordinate with any Property Manager; and
 - (vi) Coordinate and manage relationships between the Company and any joint venture partners.

3.04 Accounting and Other Administrative Services:

(i) Manage and perform the various administrative functions necessary for the management of the day-to-day operations of the Company;

- (ii) From time-to-time, or at any time reasonably requested by the Board of Directors, make reports to the Board of Directors on the Advisor's performance of services to the Company under this Agreement;
- (iii) Coordinate with the Company's independent accountants and auditors to prepare and deliver to the General Partner's audit committee an annual report covering the Advisor's compliance with certain material aspects of this Advisory Agreement;
- (iv) Provide or arrange for administrative services and items, legal and other services, office space, office furnishings, personnel and other overhead items necessary and incidental to the Company's business and operations;

- (v) Provide financial and operational planning services and portfolio management functions;
- (vi) Maintain accounting data and any other information concerning the activities of the Company as shall be needed to prepare and file all periodic financial and other reports and returns required to be filed by the General Partner with the Securities and Exchange Commission and any other regulatory agency, including annual and quarterly financial statements, in each case as applicable;
 - (vii) Maintain all appropriate books and records of the Company;
- (viii) Oversee tax and compliance services and risk management services and coordinate with appropriate third parties, including independent accountants and other consultants, on related tax matters;
- (ix) Supervise the performance of such ministerial and administrative functions as may be necessary in connection with the daily operations of the Company;
 - (x) Provide the Company with all necessary cash management services;
- (xi) Manage and coordinate with the transfer agent the distribution process and payments to stockholders of the General Partner and partners of the Company;
- (xii) Consult with the directors and officers of the General Partner and assist in evaluating and obtaining adequate insurance coverage based upon risk management determinations;
- (xiii) Provide the directors and officers of the General Partner with timely updates related to the overall regulatory environment affecting the Company, as well as managing compliance with such matters;
- (xiv) Consult with the directors and officers of the General Partner and the Board of Directors relating to the corporate governance structure and appropriate policies and procedures related thereto;
- (xv) Oversee all reporting, record keeping, internal controls and similar matters in a manner to allow the General Partner to comply with applicable law; and
- (xvi) Assist the General Partner in exercising and executing its rights and obligations in accordance with agreements between the General Partner and the ATSs and custodians with respect to the trading of the Security Tokens on the secondary market.

3.05 Stockholder Services.

(i) Manage communications with stockholders, including preparing and sending written and electronic reports and other communications; and

(ii) Establish technology infrastructure to assist in providing stockholder support and service.

3.06 Financing Services.

- (i) Identify and evaluate potential financing and refinancing sources, engaging a third-party broker, if necessary;
- (ii) Negotiate terms, arrange and execute financing agreements;
- (iii) Manage relationships between the Company and its lenders; and
- (iv) Monitor and oversee the service of the Company's debt facilities and other financings.

3.07 <u>Disposition Services.</u>

- (i) Consult with the Board of Directors and provide assistance with the evaluation and approval of potential asset dispositions, sales or other liquidity events; and
- (ii) Structure and negotiate the terms and conditions of transactions pursuant to which real estate investments may be sold.

ARTICLE 4 AUTHORITY OF ADVISOR

- 4.01 General. All rights and powers to manage and control the day-to-day business and affairs of the Company shall be vested in the Advisor to the fullest extent allowed by law. The Advisor shall have the power to delegate all or any part of its rights and powers to manage and control the business and affairs of the Company to such officers, employees, Affiliates, agents and representatives of the Advisor or the Company as it may from time to time deem appropriate. Any authority delegated by the Advisor to any other Person shall be subject to applicable law and the limitations on the rights and powers of the Advisor specifically set forth in this Agreement and the Charter and Bylaws.
- 4.02 Powers of the Advisor. Subject to the investment parameters established by the Board of Directors, with the aid of a business plan prepared by Advisor and taking into account requirements of any applicable mortgage or other loan agreement, and the delegation of authority to the Advisor to make investments on behalf of the Company, to the fullest extent allowed by law be vested in the Advisor, Advisor shall have the power by itself and shall be authorized and empowered on behalf and in the name of the Company to carry out any and all of the objectives and purposes of the Company and to perform all acts and enter into and perform all contracts and other undertakings that it may in its sole discretion deem necessary, advisable or incidental thereto to perform its obligations under this Agreement, including without limitation the acquisition of additional investment properties.

8

4.03 Approval by Directors.

- (i) The Advisor will deliver to the Board of Directors all documents required by it to properly evaluate any proposed investment.
- (ii) If the Charter, the Bylaws or applicable law requires, or the Board of Directors deems it advisable, that a transaction be approved by the Independent Directors, the Advisor will deliver to the Independent Directors all documents required by them to properly evaluate the proposed real estate investment. The prior approval of a majority of the Independent Directors will be required for each transaction between the Company, on the one hand, and the Advisor or its Affiliates, on the other hand, unless the Board of Directors determines to submit the matter to a stockholder vote or determines that the transaction is otherwise fair and reasonable to the Company.

ARTICLE 5 BANK ACCOUNTS

The Advisor will maintain one or more bank accounts in the name of the Company and will collect and deposit into any such account or accounts, and disburse from any such account or accounts, any money on behalf of the Company. Notwithstanding the foregoing, no funds shall be commingled with the funds of the Advisor.

ARTICLE 6 RECORDS AND FINANCIAL STATEMENTS

The Advisor, in the conduct of its responsibilities to the Company, shall maintain adequate and separate books and records for the Company's operations in accordance with United States generally accepted accounting principles ("GAAP"), which shall be supported by sufficient documentation to ascertain that such books and records are properly and accurately recorded. Such books and records shall be the property of the Company. Such books and records shall include all information necessary to calculate and audit the fees or reimbursements paid under this Agreement. Advisor shall utilize procedures to attempt to ensure such control over accounting and financial transactions as is reasonably required to protect the Company's assets from theft, error or fraudulent activity. All financial statements the Advisor delivers to the Company shall be prepared on an accrual basis in accordance with GAAP, except for special financial reports which by their nature require a deviation from GAAP. The Advisor shall maintain necessary liaison with the Company's independent accountants and shall provide such accountants with such reports and other information as the Company shall request.

ARTICLE 7 LIMITATION ON ACTIVITIES

Notwithstanding any provision in this Agreement to the contrary, the Advisor shall not take any action which, in its sole judgment made in good faith, would (i) adversely affect the ability of the General Partner to qualify or continue to qualify to be taxed as a REIT, (ii) subject the Company or the General Partner to regulation under the Investment Company Act of 1940, as amended, (iii) violate any law, rule or regulation of any governmental body or agency having jurisdiction over the Company, the General Partner or their Securities, or (iv) violate the Charter or Bylaws. In the event an action that would violate (i) through (iv) of the preceding sentence but such action has been ordered by the Board of Directors acting on behalf of the General Partner, the Advisor shall notify the Board of Directors of the Advisor's judgment of the potential impact of such action and shall refrain from taking such action until it receives further clarification or instructions from the Board of Directors. In such event, the Advisor shall, to the fullest extent allowed by law, have no liability for acting in accordance with the specific instructions of the Board of Directors so given. Notwithstanding the foregoing, none of the Advisor, its Affiliates and none of their managers, directors, officers, employees and equityholders, shall be liable to the Company, the General Partner, the Board of Directors or the Stockholders for any act or omission by such Persons or individuals, except as provided in this Agreement. THE PARTIES HERETO INTEND THAT THE LIMITATION OF LIABILITY SET FORTH IN THIS SECTION BE CONSTRUED AND APPLIED AS WRITTEN NOTWITHSTANDING ANY RULE OF CONSTRUCTION TO THE CONTRARY. WITHOUT LIMITING THE FOREGOING, THE LIMITATION OF LIABILITY SHALL, TO THE FULLEST EXTENT ALLOWED BY LAW, APPLY NOTWITHSTANDING ANY STATE'S "EXPRESS NEGLIGENCE RULE" OR SIMILAR RULE THAT WOULD DENY COVERAGE BASED ON A PERSON'S SOLE, CONCURRENT OR CONTRIBUTORY ACTIVE OR PASSIVE NEGLIGENCE, GROSS NEGLIGENCE OR STRICT LIABILITY. IT IS THE INTENT OF THE PARTIES THAT, TO THE EXTENT PROVIDED IN THIS SECTION, THE LIMITATION OF LIABILITY SET FORTH HEREIN SHALL, TO THE FULLEST EXTENT ALLOWED BY LAW, APPLY TO A PERSON'S SOLE, CONCURRENT OR CONTRIBUTORY ACTIVE OR PASSIVE NEGLIGENCE, GROSS NEGLIGENCE OR STRICT LIABILITY. THE PARTIES AGREE THAT THIS PROVISION IS "CONSPICUOUS" FOR PURPOSES OF ALL STATE LAWS.

9

ARTICLE 8 RELATIONSHIP WITH DIRECTORS AND OFFICERS

Managers, Directors, officers and employees of the Advisor or any direct or indirect Affiliate of the Advisor may serve as Directors, and as officers of the General Partner, except that no manager, director, officer or employee of the Advisor or any of its Affiliates who also is a Director or officer of the General Partner shall receive any compensation from the Company or General Partner for serving as a Director or officer other than reasonable reimbursement for travel and related expenses incurred in attending meetings of the Board of Directors.

ARTICLE 9 FEES

- 9.01 <u>Acquisition Fees</u>. The Company will pay the Advisor in cash or OP Units, or a combination of both, the form of payment to be determined in the sole discretion of the Advisor, as compensation for services including those described in Section 3.02, an acquisition fee of 1.0% of the Initial Asset Value of each real estate investment acquired by the Company, as well as reimburse the Advisor for all expenses incurred by the Advisor in connection with such services as required by Article 10. The Advisor shall submit an invoice to the Company following the closing or closings of each acquisition, accompanied by a computation of the fee. The fee shall be payable within ten business days after receipt of the invoice by the Company.
- 9.02 Asset Management, Accounting and Other Administrative Services Fees. The Company will pay the Advisor in cash or OP Units, or a combination of both, the form of payment to be determined in the sole discretion of the Advisor, as compensation for services including those described in Section 3.03, an asset management fee in accordance with this Section 9.02, as well as reimburse the Advisor for all expenses incurred by the Advisor in connection with such services as required by Article 10. This asset management fee shall be earned monthly and the amount of this asset management fee payable monthly, in arrears, by the Company to the Advisor equal 0.083% per month (equivalent to 1.00% per annum) of the Company's Gross Revenues for such month, determined as of the last day of each month. Notwithstanding anything herein to the contrary, under no circumstances shall the asset management fee be less than zero.
- 9.03 <u>Debt Financing Fees</u>. The Company will pay the Advisor in cash or OP Units, or a combination of both, the form of payment to be determined in the sole discretion of the Advisor, as compensation for services including those described in Section 3.06, debt financing fees in accordance with this Section 9.03, as well as to reimburse the Advisor for all expenses incurred by the Advisor in connection with such services as required by Article 10. The debt financing fees shall equal (i) \$ ONE MILLION AND 00/100 (\$1,000,000.00) in consideration of the mortgage and mezzanines loans closed on May 10, 2021 with Starwood Property Mortgage Sub-12-A, L.L.C., a Delaware limited liability company, and (ii) 1.0% of the amount of any further debt financing obtained, assumed or entered into by, or on behalf of, the Company or the General Partner, or the pro rata share of any debt financing obtained, assumed or entered into by, or on behalf of, a joint venture in which the Company and/or the General Partner has an interest. The debt financing fees payable to Advisor hereunder are in addition to any loan guaranty fee payable to the Special OP Unitholder or any of its Affiliates, as specified in the Limited Partnership Agreement.
- 9.04 <u>Disposition Fees</u>. The Company will pay the Advisor in cash or OP Units, or a combination of both, the form of payment to be determined in the sole discretion of the Advisor, as compensation for providing a substantial amount of services in an effort to sell real estate investments, including the services described in Section 3.07, disposition fees in accordance with this Section 9.04, as well as to reimburse the Advisor for all expenses incurred by the Advisor in connection with such services as required by Article 10. The disposition fees shall equal 1.0% of (i) the sales price of any real estate investment sold, that is owned directly by the Company, and (ii) when the Company owns the real estate investment indirectly through another entity, the Company's pro rata share of the sales price of any real estate investment sold by that entity.

10

ARTICLE 10 EXPENSES

- 10.01 <u>General</u>. In addition to the compensation paid to the Advisor pursuant to Article 9 hereof, the Company shall pay directly or reimburse the Advisor for all of the expenses paid or incurred by the Advisor or its Affiliates in connection with the services provided to the Company pursuant to this Agreement, including, but not limited to:
- (i) acquisition expenses incurred in connection with the selection and acquisition of real estate investments, including such direct expenses incurred related to real estate investments pursued or considered but not ultimately acquired by the Company;
- (ii) the actual out-of-pocket cost of goods and services used by the Company or the General Partner and obtained from entities not Affiliated with the Advisor, including brokerage fees paid in connection with the purchase and sale of real estate investments;
- (iii) taxes and assessments on income or assets and taxes as an expense of doing business and any other taxes otherwise imposed on the Company and its business or income;

- (iv) out-of-pocket costs associated with insurance required in connection with the business of the Company or by its officers and Directors;
- (v) all out-of-pocket expenses in connection with payments to the Board of Directors and meetings of the Board of Directors and Stockholders;
- (vi) personnel and related employment direct costs incurred by the Advisor or its Affiliates (a) in performing the services described in Section 3.05 and in providing professional services for the Company and the General Partner in-house, including legal services, tax services, internal audit services, technology-related services and services in connection with compliance with applicable laws and regulations, or (b) as otherwise approved by Independent Directors, including, but not limited to, salary, benefits, burdens and overhead of all employees directly involved in the performance of such services, plus all out-of-pocket costs incurred;
- (vii) out-of-pocket expenses of maintaining communications with Stockholders, including the cost of preparation, printing, and mailing annual reports and other Stockholder reports, proxy statements and other reports required by governmental entities;
- (viii) audit, accounting and legal fees, and other fees for professional services relating to the operations of the Company and all such fees incurred at the request, or on behalf of, the Independent Directors or any committee of the Board of Directors;
 - (ix) out-of-pocket costs for the Company to comply with all applicable laws, regulation and ordinances;
 - (x) all other out-of-pocket costs incurred by the Advisor in performing its duties hereunder; and
 - (xi) all other out-of-pocket costs necessary for the operation of the Company and its real estate investments.

Except as specifically provided for above in (vi), or as contemplated by Article 11, the expenses and payments subject to reimbursement by the Company in this Section 10.01 do not include personnel and related direct employment or overhead costs of the investment personnel of the Advisor or its Affiliates. The Company shall also reimburse the Advisor or Affiliates of the Advisor for all expenses incurred on behalf of the Company or the General Partner prior to the execution of this Agreement.

10.02 <u>Reimbursement to Advisor</u>. Expenses incurred by the Advisor on behalf of the Company and payable pursuant to this Section 10 shall be reimbursed to the Advisor within ten (10) business days after the Advisor provides the Company with an invoice and/ or supporting documentation relating to such reimbursement.

ARTICLE 11 OTHER SERVICES

Should (i) the General Partner request that the Advisor or any manager, officer or employee thereof render services for the Company other than as set forth in this Agreement or (ii) there are changes to the regulatory environment in which the Advisor or Company operates that would increase significantly the level of services performed such that the costs and expenses borne by the Advisor for which the Advisor is not entitled to separate reimbursement for personnel and related employment direct costs and overhead under Article 10 of this Agreement would increase significantly, such services shall be separately compensated at such rates and in such amounts as are agreed by the Advisor and the Independent Directors, and shall not be deemed to be services pursuant to the terms of this Agreement.

ARTICLE 12 RELATIONSHIP OF ADVISOR AND COMPANY; OTHER ACTIVITIES OF THE ADVISOR

12.01 <u>Relationship</u>. To the fullest extent allowed by law, the Company and the Advisor are not partners or joint venturers with each other, and nothing in this Agreement shall be construed to make them such partners or joint venturers. Nothing herein contained shall prevent the Advisor from engaging in other activities, including, without limitation, the rendering of advice to other Persons and the management of other programs advised, sponsored or organized by the Advisor or its Affiliates. Nor shall this Agreement limit or restrict the right of any manager, director, officer, employee, or equity holder of the Advisor or its Affiliates to engage in any other business or to render services of any kind to any other Person. The Advisor may, with respect to any investment in which the Company is a participant,

also render advice and service to each and every other participant therein. The Advisor shall promptly disclose to the Board of Directors the existence of any condition or circumstance, existing or anticipated, of which it has knowledge, which creates or could create a conflict of interest between the Advisor's obligations to the Company and its obligations to or its interest in any other Person.

12.02 <u>Time Commitment</u>. The Advisor shall, and shall cause its Affiliates and their respective employees, officers and agents to, devote to the Company such time as shall be reasonably necessary to conduct the business and affairs of the Company in an appropriate manner consistent with the terms of this Agreement. The Company acknowledges that the Advisor and other Affiliates of NRI and their respective employees, officers and agents may also engage in activities unrelated to the Company and may provide services to Persons other than the Company or any of its Affiliates.

12

12.03 <u>Investment Opportunities and Allocation</u>. The Advisor shall be required to use commercially reasonable efforts to present a continuing and suitable investment program to the Company which is consistent with the investment policies and objectives of the Company, but neither the Advisor nor any Affiliate of the Advisor shall be obligated generally to present any particular investment opportunity to the Company even if the opportunity is of character which, if presented to the Company, could be taken by the Company.

ARTICLE 13 THE NRI NAME

The Advisor, NRI and their Affiliates have a proprietary interest in the acronym "NRI". The Advisor hereby grants to the Company a non-transferable, non-assignable, non-exclusive royalty-free right and license to use the acronym "NRI" during the term of this Agreement. Accordingly, and in recognition of this right, if at any time the Company ceases to retain NRI or an Affiliate thereof to perform the services of Advisor, the Company (including the General Partner) will, promptly after receipt of written request from NRI, cease to conduct business under or use the name Nolan Reynolds International or the acronym "NRI" or any derivative thereof and the Company and the General Partner shall change the name of the Company and the General Partner to a name that does not contain the acronym "NRI" or any other letters, word or words that might, in the reasonable discretion of the Advisor, be susceptible of indication of some form of relationship between the Company and the Advisor or any Affiliate thereof. At such time, the Company will also make any changes to any trademarks, servicemarks or other marks necessary to remove any references to the acronym "NRI". Consistent with the foregoing, it is specifically recognized that the Advisor or one or more of its Affiliates has in the past and may in the future organize, sponsor or otherwise permit to exist other investment vehicles (including vehicles for investment in real estate) and financial and service organizations having "NRI" as a part of their name, all without the need for any consent (and without the right to object thereto) by the Company or the General Partner.

ARTICLE 14 TERM AND TERMINATION OF THE AGREEMENT

14.01 <u>Term</u>. This Agreement shall have an initial term of two years from the date of the Original Agreement. This Agreement may be renewed for an unlimited number of successive one-year terms upon mutual consent of the parties. Any such renewal must be approved by a majority of the Independent Directors. The General Partner (through the Independent Directors) will evaluate the performance of the Advisor annually before renewing the Agreement, and each such renewal shall be for a term of no more than one year.

14.02 <u>Termination by the Company</u>. This Agreement may be terminated immediately by the Company upon (i) any fraudulent conduct, criminal conduct, willful misconduct or the negligent breach of fiduciary duty of or by the Advisor, (ii) a material breach of this Agreement by the Advisor not cured within 10 business days after the Advisor receives written notice of such breach, or (iii) an event of the bankruptcy of the Advisor or commencement of any bankruptcy or similar insolvency proceedings of the Advisor.

13

14.03 <u>Termination by the Advisor</u>. This Agreement may be terminated immediately by the Advisor in the event of (i) the bankruptcy of the Company or commencement of any bankruptcy or similar insolvency proceedings of the Company, or (ii) any material breach of this Agreement by the Company not cured by the Company within ten 10 business days after written notice thereof.

- 14.04 <u>Payments on Termination and Survival of Certain Rights and Obligations</u>. After the Termination Date, the Advisor shall not be entitled to compensation for further services hereunder except it shall be entitled to receive from the Company within 30 days after the effective date of such termination (1) the Termination Fee, and (2) all unpaid reimbursements of expenses and all earned but unpaid fees payable to the Advisor prior to termination of this Agreement. The Advisor shall promptly upon termination:
- (a) pay over to the Company all money collected pursuant to this Agreement, if any, after deducting any accrued compensation and reimbursement for its expenses to which it is then entitled;
- (b) deliver to the Directors a full accounting, including a statement showing all payments collected by it and a statement of all money held by it, covering the period following the date of the last accounting furnished to the Directors;
 - (c) deliver to the Directors all assets and documents of the Company then in the custody of the Advisor; and
 - (d) cooperate with the Company to provide an orderly transition of advisory functions.

Upon the expiration or termination of this Agreement, neither party shall have any further rights or obligations under this Agreement, except that Articles 13, 14, 16 and 17 shall survive the termination or expiration of this Agreement.

14.05 Repurchase of Units.

The Company shall repurchase or redeem the Limited Partnership Interests held by the Advisor or any of the Advisor's affiliates as required by the Limited Partnership Agreement.

ARTICLE 15 ASSIGNMENT

This Agreement may be assigned by the Advisor to an Affiliate with the consent of the General Partner by approval of a majority of the Independent Directors. The Advisor may assign any rights to receive fees or other payments under this Agreement without obtaining the approval of the Board of Directors. This Agreement shall not be assigned by the Company without the consent of the Advisor.

14

ARTICLE 16 INDEMNIFICATION AND LIMITATION OF LIABILITY

16.01 Indemnification by the Company. The Company shall indemnify and hold harmless the Advisor and its Affiliates, including their respective managers, officers, directors, partners and employees, from all liability, claims, damages or losses arising in the performance of their duties hereunder, and related expenses, including reasonable attorneys' fees, to the extent such liability, claims, damages or losses and related expenses are not fully reimbursed by insurance, subject to any limitations imposed by the laws of the State of Florida, the Charter, Bylaws or Agreement of Limited Partnership of the Company, provided that: (i) the Advisor and its Affiliates have determined that the course of conduct which caused the loss or liability was in the best interests of the Company, (ii) the Advisor and its Affiliates were acting on behalf of or performing services for the Company, (iii) the indemnified claim was not the result of gross negligence, misconduct, or fraud of the indemnified person or resulted from a breach of the agreement by the Advisor, and (iv) in the event the loss, liability or expense arises from or out of an alleged violation of federal or state securities laws by the Advisor or its Affiliates, the conditions set forth in Section 6.3 of the Limited Partnership Agreement must be satisfied (deeming, for purposes of this Agreement, that the Advisor or its Affiliates are each an "Indemnitee" as such term is used in such clauses) for the Company to provide such indemnification. Any indemnification of the Advisor may be made only out of the net assets of the Company and not from the Stockholders.

16.02 <u>Indemnification by the Advisor</u>. The Advisor shall indemnify and hold harmless the Company from contract or other liability, claims, damages, taxes or losses and related expenses, including attorneys' fees, to the extent that such liability, claims, damages, taxes or losses and related expenses are not fully reimbursed by insurance and are incurred by reason of the Advisor's bad faith, fraud, willful misconduct or reckless disregard of its duties, but the Advisor shall not be held responsible for any action

of the Board of Directors in following or declining to follow any of the Advisor's advice or recommendation. THE PARTIES HERETO INTEND THAT THE INDEMNITIES SET FORTH IN THIS AGREEMENT BE CONSTRUED AND APPLIED AS WRITTEN NOTWITHSTANDING ANY RULE OF CONSTRUCTION TO THE CONTRARY. WITHOUT LIMITING THE FOREGOING, THE INDEMNITIES SHALL, TO THE FULLEST EXTENT ALLOWED BY LAW, AND TO THE EXTENT PROVIDED IN THIS AGREEMENT, APPLY NOTWITHSTANDING ANY STATE'S "EXPRESS NEGLIGENCE RULE" OR SIMILAR RULE THAT WOULD DENY COVERAGE BASED ON AN INDEMNIFIED PERSON'S SOLE, CONCURRENT OR CONTRIBUTORY ACTIVE OR PASSIVE NEGLIGENCE OR STRICT LIABILITY OR GROSS NEGLIGENCE. IT IS THE INTENT OF THE PARTIES THAT, TO THE EXTENT PROVIDED IN THIS AGREEMENT, THE INDEMNITIES SET FORTH HEREIN SHALL, TO THE FULLEST EXTENT ALLOWED BY LAW, APPLY TO AN INDEMNIFIED PERSON'S SOLE, CONCURRENT OR CONTRIBUTORY ACTIVE OR PASSIVE NEGLIGENCE OR STRICT LIABILITY OR GROSS NEGLIGENCE. THE PARTIES AGREE THAT THIS PROVISION IS "CONSPICUOUS" FOR PURPOSES OF ALL STATE LAWS.

15

16.03 Advisor's Liability

- (i) Notwithstanding any other provisions of this Agreement, in no event shall the Company make any claim against Advisor, or its Affiliates, on account of any good faith interpretation by Advisor of the provisions of this Agreement (even if such interpretation is later determined to be a breach of this Agreement) or any alleged errors in judgment made in good faith and in accordance with this Agreement in connection with the operation of the operations of the Company hereunder by Advisor or the performance of any advisory or technical services provided by or arranged by the Advisor. The provisions of this Section 16.3(i) shall not be deemed to release Advisor from liability for its gross negligence.
- (ii) The Company shall not object to any expenditures made by the Advisor in good faith in the course of its performance of its obligations under this Agreement or in settlement of any claim arising out of the operation of the Company unless such expenditure is specifically prohibited by this Agreement. The provisions of this Section 16.03(ii) shall not be deemed to release Advisor from liability for its gross negligence.
- (iii) IN NO EVENT WILL EITHER PARTY BE LIABLE FOR DAMAGES BASED ON LOSS OF INCOME, PROFIT OR SAVINGS OR INDIRECT, INCIDENTAL, CONSEQUENTIAL, EXEMPLARY, PUNITIVE OR SPECIAL DAMAGES OF THE OTHER PARTY OR PERSON, INCLUDING THIRD PARTIES, EVEN IF SUCH PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES IN ADVANCE, AND ALL SUCH DAMAGES ARE EXPRESSLY DISCLAIMED. IN NO EVENT WILL ADVISOR'S AGGREGATE LIABILITY UNDER THIS AGREEMENT EVER EXCEED THE TOTAL AMOUNT OF FEES IT ACTUALLY RECEIVES FROM THE COMPANY PURSUANT TO ARTICLE 9.
- (iv) THE PARTIES HERETO INTEND THAT THE RELEASE FROM LIABILITY SET FORTH IN SECTION 16.03 BE CONSTRUED AND APPLIED AS WRITTEN NOTWITHSTANDING ANY RULE OF CONSTRUCTION TO THE CONTRARY. WITHOUT LIMITING THE FOREGOING, THE RELEASE FROM LIABILITY SHALL, TO THE FULLEST EXTENT ALLOWED BY LAW, APPLY NOTWITHSTANDING ANY STATE'S "EXPRESS NEGLIGENCE RULE" OR SIMILAR RULE THAT WOULD DENY COVERAGE BASED ON A PERSON'S SOLE, CONCURRENT OR CONTRIBUTORY ACTIVE OR PASSIVE NEGLIGENCE OR STRICT LIABILITY. IT IS THE INTENT OF THE PARTIES THAT, TO THE EXTENT PROVIDED IN SECTION 16.03, THE RELEASE FROM LIABILITY SET FORTH HEREIN SHALL, TO THE FULLEST EXTENT ALLOWED BY LAW, APPLY TO A RELEASED PERSON'S SOLE, CONCURRENT OR CONTRIBUTORY ACTIVE OR PASSIVE NEGLIGENCE OR STRICT LIABILITY. THE PARTIES AGREE THAT THIS PROVISION IS "CONSPICUOUS" FOR PURPOSES OF ALL STATE LAWS.

ARTICLE 17 MISCELLANEOUS

17.01 Notices. Any notice, report or other communication required or permitted to be given hereunder shall be in writing unless some other method of giving such notice, report or other communication is required by the Charter, the Bylaws, or accepted by the party to whom it is given, and shall be given by being delivered by hand or by overnight mail or other overnight delivery service to the addresses set forth herein:

To the Company, the General Partner or the Directors:

NRI Real Token LP c/o NRI Real Estate Investment & Technology, Inc. 1340 S. Dixie Highway Coral Gables, Fl 33134

To the Advisor:

NRI Real Token Advisors LLC 1340 S. Dixie Highway Coral Gables, FL 33134

16

Either party may at any time give notice in writing to the other party of a change in its address for the purposes of this Section 17.01.

- 17.02 <u>Modification</u>. This Agreement shall not be changed, modified, terminated, or discharged, in whole or in part, except by an instrument in writing signed by all parties hereto, or their respective successors or assignees.
- 17.03 <u>Severability</u>. The provisions of this Agreement are independent of and severable from each other, and no provision shall be affected or rendered invalid or unenforceable by virtue of the fact that for any reason any other or others of them may be invalid or unenforceable in whole or in part.
- 17.04 <u>Construction</u>. The provisions of this Agreement shall be construed and interpreted in accordance with the laws of the State of Florida.
- 17.05 Entire Agreement. This Agreement contains the entire agreement and understanding among the parties hereto with respect to the subject matter hereof, and supersedes all prior and contemporaneous agreements, understandings, inducements and conditions, express or implied, oral or written, of any nature whatsoever with respect to the subject matter hereof. The express terms hereof control and supersede any course of performance and/or usage of the trade inconsistent with any of the terms hereof. This Agreement may not be modified or amended other than by an agreement in writing.
- 17.06 <u>Waiver</u>. Neither the failure nor any delay on the part of a party to exercise any right, remedy, power or privilege under this Agreement shall operate as a waiver thereof, nor shall any single or partial exercise of any right, remedy, power or privilege preclude any other or further exercise of the same or of any other right, remedy, power or privilege, nor shall any waiver of any right, remedy, power or privilege with respect to any occurrence be construed as a waiver of such right, remedy, power or privilege with respect to any other occurrence. No waiver shall be effective unless it is in writing and is signed by the party asserted to have granted such waiver.
- 17.07 <u>Gender</u>. Words used herein regardless of the number and gender specifically used, shall be deemed and construed to include any other number, singular or plural, and any other gender, masculine, feminine or neuter, as the context requires.
- 17.08 <u>Titles Not to Affect Interpretation</u>. The titles of Articles and Sections contained in this Agreement are for convenience only, and they neither form a part of this Agreement nor are they to be used in the construction or interpretation hereof.
- 17.09 Counterparts. This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original as against any party whose signature appears thereon, and all of which shall together constitute one and the same instrument. This Agreement shall become binding when one or more counterparts hereof, individually or taken together, shall bear the signatures of all of the parties reflected hereon as the signatories.

[The remainder of this page is intentionally left blank. Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date and year first above written.

NRI Real Token LP

By: NRI Real Estate Investment & Technology Inc.

Its: General Partner

By: /s/ Brent Reynolds

Name: Brent Reynolds

Title: President and Chief Executive

NRI Real Token Advisors LLC

By: Nolan Reynolds International, LLC

Its: Sole Member

By: /s/ Brent Reynolds

Name: Brent Reynolds
Title: Manager

NRI Real Estate Investment & Technology Inc.

By: /s/ Brent Reynolds

Name: Brent Reynolds

Title: President and Chief Executive Officer

By signing below, NRI Real Token Thesis LLC acknowledges its agreement to the changes to the Original Agreement as reflected by this Agreement as of the date and year first above written.

NRI Real Token Thesis LLC

By: /s/ Brent Reynolds

Name: Brent Reynolds

Title: Authorized Representative

[Signature Page for Amended and Restated Advisory Agreement]

Cover Sep. 30, 2022

Cover [Abstract]

Document Type 8-K
Amendment Flag false

<u>Document Period End Date</u> Sep. 30, 2022

Current Fiscal Year End Date --12-31
Entity File Number 000-56395

Entity Registrant Name NRI Real Estate Investment and Technology, Inc.

Entity Central Index Key 0001877561 Entity Tax Identification Number 87-1031361

Entity Incorporation, State or Country Code MD

Entity Address, Address Line One 1340 South Dixie Highway

<u>Entity Address, Address Line Two</u>

<u>Entity Address, City or Town</u>

Suite 612

Coral Gables

Entity Address, State or Province
Entity Address, Postal Zip Code
City Area Code
Local Phone Number
Written Communications
Soliciting Material
FL
33146
(305)
259-9928
FL
59-9928
FL
6305)

Pre-commencement Tender OfferfalsePre-commencement Issuer Tender OfferfalseEntity Emerging Growth CompanytrueElected Not To Use the Extended Transition Period false

```
|
|-
| "labellish": (
| "less!": [
| "net-20220930_lab.eml"
                                     "presentationLink": [
"local": [
"noi-20220931 pre-mal"]
                        Framework (1997) and the control of 
"National Action Communication (1982) (Action Communication (1982) (Action Communication Communicati
                        "Section of the control of the contr
                                                 "Sand"
Panel Translation of the Control of the Cont
            "Tanks off; |
"Tank" |
"Tank" |
"Tank" |
"Tank" |
"Tank" |
"Tank" |
"Tanks 
                                     The state of the s
                                     "Seminary", "Seminary lengther",
"Seminary", "Seminary lengther",
"Seminary Land Seminary Land Semin
                                                 | |
| "Londines": "CorrectEstatThesEndints";
| "Annal": "Things / Annal Lear, queries / 2022";
| "Annal": "Things / Annal Lear, queries / 2022";
| "Annal Lear": "Annal Correct"
| "Annal Lear": "Bloom Annal Lear": "Annal Lear":
                                                             "pd9"

[r
"lang", |
"mermath |
"m
                                                             "Tear";
Tearg";
Tearges;
Tearg
                                     Variable Control of the Control of t
                                                 "Seal Same", "Showmen' Faired Date Fromt",
"Seal Same", "Showmen' Faired Date Fromt",
"Shape John Same",
"Sh
```

), "localname": "DocumentPeriodEndDate", "nessi": "bothps//akel.eer.gov/dei/2022",

```
**Section of the section of the sect
          | The constitute of the consti
                    The state of the s
          "*:33"
[**Ing": [
**Ing": [
**Ing: [
**In
          And the second s
Takasianan', "Takasanah lawayara milyini farana Takikianin',
"Takasianan', "Takasanin lawayara milyini farana Takikianin',
"Takasi Takikianin lawayara milyini farana fara
                                                   merum": [
*pain": [
*pain": |
*dicumentation": "Address Line 1 much as Atto, Excilding Hase, Except Hase"

*Label": "Entity Address, Address Line Cue"
          | Translater", "Enhistphisterschlers vilet",
"massis", "behard/saks.com.gov/des/2022",
"massis", "behard/saks.com.gov/des/2022",
"behard/des/and/saks.com.gov/des/2022",
"behard/des/and/saks.com.gov/des/2022",
"behard/des/and/saks.com.gov/des/2022",
"des/and/saks.com.gov/des/and/saks.com.gov/des/2022",
"des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov/des/and/saks.com.gov
                                                   terma": [
"mole": [
"documentation": "Address Line 2 such as Street or Suite number",
"label": "Entity Address, Address Line Two"
.
Juniorania - Statisphiemshiamshire? 

- Small 
                    terus": |
"sole": |
"documentation": "Hame of the City or Trun",
"label": "Entity Johnson, City or Trun",
          Section of the sectio
"Seminary" "Schriphianschronistigsfeder,
"Seminary" "Schriphianschronistigsfeder,
"Seminary" "Seminary Schriphianschronistigsfeder,
"Seminary" "Seminary Schriphianschronistigs"
Schriphianschronistigsfeder,
"Seminary" "Seminary Schriphianschronistigs"
"Seminary Schriphianschronistig
"Manipulation of Manipulational State Office of Manipulation State Office 
          |-
| "Records |
| "Records |
| "Association": "The registrants involved in landscaping per
| "Association": "Entity Backenpley Proceedings, Separating Correct"
| Laboration of the Control of the Control of Correct"
                    Figure 1

Street 1

Street 7

Street 7

Street 8

Street 8

Street 8

Street 9

Street
                    "Foundamen", "MankipContenlinesding",
"Foundamen", "DahkipContenlinesding",
"Foundamen", "Dahkip(/Adal.ton.gov/don/2022",
"Pankip(/MZ/cale)Contenlinesding",
"Analyze", "enthealIndesdingSchoolSype"
des [DakkyContenlinesdingSchoolSype"
des [DakkyContenlinesdingSchoolSype

          "invaluame": "EntityCurrentEsportingDistur",
"neuri": "http://akel.esr.gov/dei/2023",
```

```
"Tankinge": "SchilgergingTreckhingsory",
"SchilgergingTreckhingsory",
"SchilgergingTreckhingsory",
"SchilgergingTreckhingsory",
"SchilgergingTreckhingTreckhingsory",
"SchilgergingTreckhingTreckhingsory"
"SchilgergingTreckhingTreckhing":
"SchilgergingTreckhingTreckhing":
"SchilgergingTreckhingTreckhing":
"SchilgergingTreckhingTreckhing":
"SchilgergingTreckhingTreckhing":
"SchilgergingTreckhingTreckhing":
"SchilgergingTreckhingTreckhingTreckhing":
"SchilgergingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhingTreckhin
                                             "sig" |
"India" |
"Indiana if as emerging greath suspany has elected set to use the eat
"India" |
"India" 
                      """
"Inninam", "Entipfildbakes",
"mark", "http://chal.com.que/de/2022",
"http://chal.com.que/de/2022",
"http://de/2022",
"de/2022", "ElefuniesItenType"
"de/2022", "de/2022",
"de/2022", "elefuniesItenType"
"de/2022",
"sch.que", "elefuniesItenType", [
"sch.que", "elefuniesItenType", [
"sch.que", "elefuniesItenType", [
"sch.que", [
                      Construction of the constr
                                  "suff "Temp": [
Temp": [
Temp": [
Temp": Temp": Temp: Research Substitution (SE) Number for the Entry.,
Tempt: Tempty Persons SE Substitution (SE) Number for the Entry.,
                      ""
"Invaluam", "EnklyPrimaryEinhoshom",
"Invaluam", "EnklyPrimaryEinhoshom",
"EnklyPrimaryEinhos
                                                                               ang": {
'colo": {
'decumentation': "The approprie n
'label": "Entry Public Float"
                                             "Warming": {
    "Same": {
    "Same": {
        "Same": {
        "Same " }
        "Annex of the same of the "Label": "Denishy Registerant Name"

                                  ""
"Incaloses", "MonkingSegion acedimes",
"Monkey / Aded Leno, quer/dee/ 2022",
"Makey / (Mail / eno, quer/dee/ 2022",
"Makey / (Mail / eno eno)
"Makey / (Mail / eno eno)
"Makey / (Mail / eno eno)
"Makey / makey / enough incalibration inglices 2 year
"Makey Makey Enough incalibration in the 
                                  "Tensiame", "MonkipShellCompany",
"Tensiame", "MonkipShellCompany",
"Tensiame", "MonkipShellCompany",
"Tensiame", "Tensiame",
"Tensiame", "Tensiame",
"Tensiame", "Tensiame",
"Tensiame", "Tensiame",
"Tensiame", "Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",
"Tensiame",

                                             le
"Tange"; {
"Serve"; {
"Serve"; {
"Self"; {
"Advantation"; "Indicates that the company is a Beslier Reporting Company (EEC).",
"Index"; "Exhibity Beall Resines"
                                  Target |
Security |
Se
                      Variations (Notice plantament than contained to the conta
                                             The control of the co
                                                                               ang's |

nervol's |

"sole"s |

"decommodation"s "Extension number for local phone number.",

"label"s "Extension"
                      James Learning, "Management," Physical Learning and Control of State 1, 2022 and Learning, "Management," Physical Learning and Learning
                                  The second of th
                                  **Size**

**Size
and Repairing Resident (1975 and 1975 ages)

| The control of the
```

```
"excus": [
"exit": [
"exit": [
"discountiation": "Horizon High that in two when the From B-E filling is inhunded to natisfy
"label": "Free-commonment Insure Tender Office"
                                                                                    "Parallemen", "Parilementement lance Tender Differ",
"Standause", "Parilementement lance Tender Differ",
"Parameter lance",
"Standause lance",
"Standause lance Deres"
"Standause", "Standause lance lance
"Standause", "Standause lance lance
"Standause", "Standause lance lance
"Standause", "Standause lance l
                                                                                                  "sg" | Same" | Same | Sa
                                                                      Seminary Comment (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987) (1987)
                                                                      Variational ** Amount plantation |
**The control of the control of
                                                        Transform (
"Thing," (
                                                                      "Tonal name", "DemonstryDemonstryDelignation",
"Invalid name", "DemonstryDemonstryDelignation",
"presentations",
"Balays / Mall raise/DemonstragEndignationStreeDype"
"Associated "", "securityDemonstragEndignationStreeDype"
"", "securityDemonstragEndignationStre
                                                                                                  Commission of Mathematical Conference of Con
                                                        Construction of Performance of Perfo
"The second seco
                            "Subsections", "$"

**GOT* | "Subsections of the "Subsections", "Subsections
                            "Debanchism": "a"
];
"c13": [
"Stam": Them 21-TT,
"Stamber": T328",
"Stables": T328",
"Stables
                            "Submention": "4"
];
"ed3": [
"Stand": "Flores 10-K, 10-Q, 20-F",
"Sumbset": "540",
"Sublisher": "580",
"Sublisher": "18",
"Submention": "18",
"Submention": "18",
              "Subsension", "a-1"
"side", "supplement a-1",
"side", "supplement a-1",
"Subsens", "Side",
"Subsension", "
                            ),
"418", (
"Esse": "Essentition Joh",
"Subject": "230",
"Publisher": "ESC",
"Section": "423"
              "Manchine", "All's",
"sel's" ( "Manchine Joh",
"Souls", "Manchine Joh",
"Souls", "Souls", "War,
"Souls in", "War,
"Souls in", "War,
"Souls in", "War,
"Souls in", "Souls in ",
"Souls in ", "Souls in ",
"Souls in "
Common State of Common State o
```